CARDIOVASCULAR AND CEREBROVASCULAR FUNCTION IN SYNCOPAL & NON-SYNCOPAL PATIENTS

Doctor of Medicine Thesis

University of Leicester

2001

Brian John Carey

MB, BCh, BAO, MRCPI

UMI Number: U601217

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI U601217 Published by ProQuest LLC 2013. Copyright in the Dissertation held by the Author. Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346 To my wife, Emma, and daughters, Elisha and Anna.

Table of Contents

Title page	i
Dedication	ii
Table of Contents	iii
List of Figures	xi
List of Tables	xv
List of Abbreviations	xvii
Study & Ethical Declarations	xx
Acknowledgements	ххі

1.	Introduction	1
1.1.	Vasovagal syncope	2
1.1.1.	Historical aspects	2
1.1.2.	Epidemiology	3
1.1.3.	Pathophysiology	4
1.1.3.1.	Normal physiological responses to adoption of upright posture	4
1.1.3.2.	Pathophysiological cardiovascualar changes during pre-syncope	5
1.1.3.3.	Venous pooling & reduced intravascular volume	6
1.1.3.4.	Ventricular volume and contraction	7
1.1.3.5.	Central processing & neurotransmitters	8
1.1.3.6.	Autonomic function	10
1.1.3.7.	Baroreceptor sensitivity	12
1.1.3.8.	Plasma catecholamines	13
1.1.3.9.	Cerebral autoregulation	14
1.1.3.10.	Respiration	15

----- iii

1.1.3.11.	Endothelial factors	16
1.1.4.	Clinical characteristics	17
1.1.5.	Investigations	19
1.1.5.1.	Normal cardiovascular, respiratory & cerebrovascular	20
	responses to head-up tilt	
1.1.5.2.	Indications, contraindications and risks of head-up tilt testing	21
1.1.5.3.	Equipment	22
1.1.5.4.	Prior instructions and environment	22
1.1.5.5.	Tilt angle and duration	23
1.1.5.6.	Provocation during head-up tilt testing	24
1.1.5.6.1.	Isoproterenol	24
1.1.5.6.2.	Sublingual nitroglycerin	25
1.1.5.6.3.	Other pharmaceutical agents	26
1.1.5.6.4.	Lower body negative pressure	26
1.1.5.7.	Positivity criteria	26
1.1.5.8.	Sensitivity, specificity and reproducibility	27
1.1.5.9.	Testing in older patients	29
1.1.6.	Treatment	30
1.2.	Cerebral blood flow	31
1.3.	Cerebral autoregulation	33
1.3.1.	Historical perspective	33
1.3.2.	Mechanisms of cerebral autoregulation	36
1.3.2.1.	Tissue perfusion theory	36
1.3.2.2.	Metabolic theory	37
1.3.2.3.	Myogenic theory	38

1.3.2.4.	Neurogenic theory	39
1.3.2.5.	Endothelial derived factor theory	41
1.3.3.	Factors that may affect cerebral autoregulation	43
1.3.3.1.	Carbon dioxide partial pressure	43
1.3.3.2.	Age	44
1.3.3.3.	Gender	46
1.3.3.4.	Blood pressure	46
1.3.3.5.	Drug therapy	48
1.3.3.6.	Posture	49
1.3.3.7.	Miscellaneous factors	50
2.	Methods	57
2.1.	Measurement of cerebral blood flow	58
2.1.1.	Nitrous oxide method	58
2.1.2.	Xenon inhalation / injection method	59
2.1.3.	Xenon enhanced computer tomography	60
2.1.4.	Other radiological techniques	60
2.1.5.	Transcranial Doppler ultrasonography	61
2.1.5.1.	Physical and technical aspects	61
2.1.5.2.	Vessel identification and reproducibility	63
2.1.5.3.	Validation as a method of measuring cerebral blood flow	64
2.1.5.4.	Calculation of indices of cerebrovascular resistance & critical	66
	closing pressure using transcranial Doppler ultrasound	
	measured cerebral blood flow velocity	
2.2.	Non-invasive continuous blood pressure monitoring	68

_____ V

2.3.	Measurement of carbon dioxide partial pressure	71
2.3.1.	End-tidal carbon dioxide measurement	72
2.3.2.	Transcutaneous carbon dioxide measurement	73
2.4.	Assessment of cerebral autoregulation	75
2.4.1.	Assessment of static cerebral autoregulation	76
2.4.2.	Assessment of dynamic cerebral autoregulation	77
2.4.2.1.	Aaslid's method	77
2.4.2.2.	Frequency domain analysis	80
2.5.	Methods used to induce static and dynamic blood pressure	84
	changes	
2.5.1.	Thigh cuff inflation and release	84
2.5.2.	Valsalva manoeuvre	85
2.5.3.	Lower body negative pressure	86
2.6.	Study protocol	88
2.7.	Data analysis	89
3.	Effect of ageing on cerebral autoregulation	112
3.1.	Summary	113
3.2.	Background	113
3.3.	Objectives	115
3.4.	Methods	115
3.4.1.	Subjects	115
3.4.2.	Study protocol	116
3.4.2.1.	Static pressor stimulus	116
3.4.2.2.	Static depressor stimulus	116

----- vi

3.4.2.3.	Dynamic pressor stimuli	117
3.4.2.4.	Dynamic depressor stimuli	117
3.4.3.	Data analysis	118
3.4.4.	Determination of cerebral autoregulation	118
3.4.4.1.	Determination of static cerebral autoregulation	118
3.4.4.2.	Determination of dynamic cerebral autoregulation using	118
	Aaslid's model	
3.4.4.3.	Determination of dynamic cerebral autoregulation using	119
	step response analysis	
3.4.5.	Determination of cardiac baroreceptor sensitivity	119
3.4.6.	Statistical analysis	120
3.5.	Results	121
3.6.	Discussion	122
3.6.1.	Pressor and depressor changes	123
3.6.2.	Autoregulatory indices	123
3.6.3.	Magnitude of blood pressure changes	124
3.6.4.	Step response analysis	124
3.6.5.	Age and cerebrovascular function	125
3.6.6.	Limitations	126
3.7.	Conclusions	127
4.	Baseline Cerebral Autoregulation in Patients With	133
	Recurrent Vasovagal Syncope	
4.1.	Summary	134
4.2.	Background	134

_____ vii

		Contents
4.3.	Objectives	136
4.4.	Methods	136
4.4.1.	Subjects	136
4.4.2.	Study protocol	137
4.4.3.	Data analysis	138
4.4.4.	Determination of dynamic cerebral autoregulation using	138
	Aaslid's method	
4.4.5.	Determination of static cerebral autoregulation	139
4.4.6.	Frequency domain analysis	139
4.4.7.	Statistical analysis	140
4.5.	Results	140
4.6.	Discussion	141
4.6.1.	Cerebral autoregulation during pre-syncope	142
4.6.2.	Autoregulatory indices	142
4.6.3.	Static & dynamic blood pressure stimuli	143
4.6.4.	Frequency domain analysis	143
4.6.5.	Limiting factors	144
4.7.	Conclusions	145
5.	Cerebral Autoregulatory Responses To Head-Up Tilt in	155
	Normal Subjects & Patients With Recurrent Vasovagal	
	Syncope	
5.1.	Summary	156
5.2.	Background	157
5.3.	Objectives	158

viii viii

. .

<u></u>		Contents
5.4.	Methods	158
5.4.1.	Subjects	158
5.4.2.	Study protocol	159
5.4.3.	Data analysis	160
5.4.4.	Statistical analysis	161
5.5.	Results	161
5.6.	Discussion	163
5.6.1.	Dynamic cerebral autoregulation during orthostatic stress	164
5.6.2.	Dynamic cerebral autoregulation and carbon dioxide changes	165
	after head-up tilt	
5.6.3.	Dynamic cerebral autoregulation during pre-syncope	165
5.6.4.	Model validity and accuracy	166
5.6.5.	Limiting factors	166
5.7.	Conclusions	167
6.	Carbon Dioxide, Critical Closing Pressure & Cerebral	178
	Haemodynamics During Pre-Syncope	
6.1.	Summary	179
6.2.	Background	180
6.3.	Objectives	181
6.4.	Methods	181
6.4.1.	Subjects	181
6.4.2.	Study protocol	182
6.4.3.	Data analysis	182
6.4.4.	Statistical analysis	183

_____ ix

6.5.	Results	183
6.6.	Discussion	185
6.6.1.	Carbon dioxide changes during pre-syncope	186
6.6.2.	Pre-syncopal changes in cerebrovascular resistance	186
6.6.3.	Pre-syncopal changes in critical closing pressure	187
6.6.4.	Carbon dioxide and critical closing pressure during pre-syncope	188
6.6.5.	Similar changes in patients and syncopal controls	188
6.6.6.	Limiting factors	189
6.7.	Conclusions	190
7.	Conclusions of the thesis	203
7.1.	Summary of results	204
7.2.	Study limitations	209
7.3.	Prospects for further studies	211
7.4.	Conclusion	213
Appendix		216
Publications a	Publications arising from this thesis	
References		219

List of Figures

1.1	Cerebral autoregulatory curves	52
2.1	Circle of Willis	91
2.2	Transtemporal acoustic windows	92
2.3	A subject undergoing transcranial Doppler ultrasound	93
2.4	Continuous middle cerebral artery blood flow velocity profiles as	94
	displayed using the transcranial Doppler ultrasound apparatus	
2.5	Method of calculation of critical closing pressure and resistance	95
	area product	
2.6	The Ohmeda 2300 Fianpres device with cuff applied	96
2.7	End-tidal carbon dioxide monitoring using capnography	97
2.8	Transcutaneous carbon dioxide monitoring using the TINA device	98
2.9	Schematic representation of normal and impaired static cerebral	99
	autoregulation	
2.10	Aaslid's model of mathematical curves of dynamic cerebral	100
	autoregulation in response to a step change in blood pressure	
2.11	Coherence (a), phase (b), amplitude (c), impulse response (d) and step	101
	response (e) plots of subjects with normal and impaired dynamic	
	cerebral autoregulation derived using frequency domain analysis	
2.12	A volunteer undergoing thigh cuff inflation and release	106
2.13	A volunteer performing a Valsalva manoeuvre	107
2.14	Application of lower body negative pressure	108

_____ xi

3.1	Mean step responses and standard errors of normal young and older	129
	subjects	
4.1	Mean arterial blood pressure and cerebral blood flow velocity	147
	coherence responses of 17 patients with recurrent vasovagal syncope	
	and 17 matched control subjects derived using supine rest recordings	
	and frequency domain analysis	
4.2	Mean phase frequency responses of 17 patients with recurrent	148
	vasovagal syncope and 17 matched control subjects derived using	
	supine rest recordings and frequency domain analysis	
4.3	Mean amplitude frequency responses of 17 patients with recurrent	149
	vasovagal syncope and 17 matched control subjects derived using	
	supine rest recordings and frequency domain analysis	
4.4	Mean impulse responses of 17 patients with recurrent vasovagal	150
	syncope and 17 matched control subjects derived using supine rest	
	recordings and frequency domain analysis	
4.5	Mean step responses of 17 patients with recurrent vasovagal syncope	151
	and 17 matched control subjects derived using supine rest recordings	
	and frequency domain analysis	
5.1	Actual data record of a non-syncopal control around the point of	169
	head-up tilt	
5.2	Actual data record of a patient around the point of syncope	170
5.3	Mean middle cerebral artery mean pressure and cerebral blood flow	171
	velocity changes of 17 patients with recurrent vasovagal syncope and	

— xii

17 pair matched controls after head-up tilt

- 5.4 Mean transcutaneous and end-tidal carbon dioxide changes of 17
 patients with recurrent vasovagal syncope and 17 pair matched controls
 after head-up tilt
- 6.1 Mean middle cerebral artery systolic, mean and diastolic blood
 192
 pressure changes of 16 patients with recurrent vasovagal syncope and
 14 syncopal control subjects before and after syncope
- 6.2 Mean pulse interval changes of 16 patients with recurrent vasovagal
 193 syncope and 14 syncopal control subjects before and after syncope
- 6.3 Mean middle cerebral artery systolic, mean and diastolic blood flow
 194 velocity changes of 16 patients with recurrent vasovagal syncope and
 14 syncopal control subjects before and after syncope
- 6.4 Mean resistance area product and classical cerebrovascular resistance
 195 changes of 16 patients with recurrent vasovagal syncope and
 14 syncopal control subjects before and after syncope
- 6.5 Mean Gosling's pulsatility index changes of 16 patients with recurrent 196
 vasovagal syncope and 14 syncopal control subjects before and after
 syncope
- 6.6 Mean transcutaneous carbon dioxide changes of 16 patients with 197 recurrent vasovagal syncope and 14 syncopal control subjects and mean end-tidal carbon dioxide changes of 5 patients and 2 syncopal controls combined before and after syncope
- 6.7 Mean middle cerebral artery diastolic blood pressure and critical 198 closing pressure changes of 16 patients with recurrent vasovagal

syncope and 14 syncopal control subjects before and after syncope

6.8 Instantaneous arterial blood pressure-cerebral blood flow velocity 199
 relationships of a representative patient 180 and 20 seconds prior to syncope

List of Tables

1.1.	Causes of syncope	53
1.2.	Investigations for syncope	54
1.3.	Classification of vasovagal syncope	55
2.1.	Identification of basal cerebral arteries using transcranial Doppler	109
	ultrasound	
2.2.	Static and dynamic pressor and depressor stimuli	110
3.1.	Demographic and baseline characteristics of young and older groups	130
3.2.	Induced and spontaneous mean arterial pressor and depressor changes	131
	and autoregulatory indices of young and older groups	
4.1.	Demographic and baseline characteristics of patient and control groups	152
4.2.	Static and dynamic autoregulatory indices of patient and control groups	153
5.1.	Demographic and baseline characteristics of the patient group,	173
	non-syncopal control group and syncopal control groups	
5.2.	Middle cerebral artery mean blood flow velocity changes of patient	174
	and non-syncopal control groups before, during and after head-up tilt	
5.3.	Mean arterial pressure and middle cerebral artery mean pressure	174
	changes of patient and non-syncopal control groups before, during	
	and after head-up tilt	
5.4.	Pulse interval changes of patient and non-syncopal control groups	175

– XV

before, during and after head-up tilt

5.5.	Transcutaneous and end-tidal carbon dioxide changes of patient and	175
	non-syncopal control groups before, during and after head-up tilt	
5.6.	Dynamic cerebral autoregulatory indices and correlation coefficients	176
	assessing how closely measured cerebral blood flow velocity fitted	
	model predicted velocity in patient and non-syncopal control subjects	
	for one-minute periods before during and after head-up tilt	
6.1.	Demographic and baseline characteristics of syncopal patients,	200
	syncopal control and non-syncopal control subjects	
6.2.	Cardiovascular and cerebrovascular parameters 180 seconds prior	201
	to, and at the point of, being returned supine in 16 syncopal patients	
	and 14 syncopal control subjects	

----- xvi

List of Abbreviations

ANOVA	analysis of variance
ARI	autoregulatory index
AVM	arteriovenous malformation
BP	blood pressure
BRS	cardiac baroreceptor sensitivity
CA	cerebral autoregulation
CBF(V)	cerebral blood flow (velocity)
CSM	carotid sinus massage
CO ₂	carbon dioxide
CrCP	critical closing pressure
CCVR	classical cerebrovascular resistance
СТ	computerised tomography
CVR	cerebrovascular resistance
DBP	diastolic blood pressure
ECG	electrocardiogram
EDCF	endothelial derived constricting factor
EDRF	endothelial derived relaxing factor
EEG	electroencephalogram
ETCO ₂	end tidal carbon dioxide
FFT	fast Fourier transform
GPI	Gosling's pulsatility index
HR	heart rate
HUT	head-up tilt

- ICP intracranial pressure
- LBNP lower body negative pressure
- MAP mean arterial blood pressure
- MCA middle cerebral artery
- MCADP middle cerebral artery diastolic blood pressure
- MCADV middle cerebral artery diastolic blood flow velocity
- MCAMP middle cerebral artery mean blood pressure
- MCAMV middle cerebral artery mean blood flow velocity
- MCASP middle cerebral artery systolic blood pressure
- MCASV middle cerebral artery systolic blood flow velocity
- MCAV middle cerebral artery blood flow velocity
- MRI magnetic resonance imaging
- NO(S) nitrous oxide (synthetase)
- OH orthostatic hypotension
- PI pulse interval
- PP perfusion pressure
- RAP resistance-area product
- RSD spontaneous depressor blood pressure changes at rest
- RSP spontaneous pressor blood pressure changes at rest
- SBP systolic blood pressure
- SD standard deviation
- SSNSA systemic sympathetic nervous system activity
- TCD transcranial Doppler ultrasound
- TCO₂ transcutaneous carbon dioxide
- THC thigh cuff

- TPR total peripheral resistance
- VVS vasovagal syncope

Study Declaration

As required by the University of Leicester, this declaration outlines the persons responsible for this thesis and the extent to which the work is my own.

The design, organisation and administration of the study was performed by myself with the help and advice of Professor John F. Potter, Professor of Medicine for the Elderly, University of Leicester, and Professor Ronney B. Panerai, Professor of Physiological Measurements, Department of Medical Physics, University of Leicester.

Patient recruitment, data handling and statistical analysis in this study were performed by myself. I am very grateful, however, for the considerable help of Dr. Penelope Eames and Dr. Melanie Blake with the recruitment and data analysis of young and older subjects reported in Chapter 3. In addition, I am very grateful to Mr. Bradley Manktelow and Miss Clare Gillies, Medical Statisticians, Department of Epidemiology, University of Leicester, who assisted with statistical analysis of data reported in Chapters 3, 5 and 6.

Professor Panerai developed the software for the recording, calibration and editing of raw data and the software used for frequency domain analysis and calculation of static and dynamic autoregulatory indices and cardiac baroreceptor sensitivity.

Ethical Declaration

In accordance with ethical requirements for medical research, all subjects gave written informed consent. All techniques and studies were approved by the Leicestershire Hospital Ethics Committee. I am indebted to my supervisor, Professor John Potter, for his advice, support and patience throughout this study.

I would also like to thank Professor Ronney Panerai, Mr. Bradley Manktelow, Miss Clare Gillies and Mrs. Sue Lewin for their help, tolerance and enthusiasm during the completion of this thesis.

A special word of thanks to Dr. Penelope Eames, my 'partner in crime', who played a large part in this thesis seeing the light of day.

CHAPTER ONE

Introduction

— 1

1.1. Vasovagal syncope

1.1.1. Historical aspects

Syncope is a transient loss of consciousness that is accompanied by loss of postural tone. The word 'syncope' is derived from the Greek word for 'cut short' and this concept of sudden cutting short of circulation and conscious awareness has been recognised as a clinical entity throughout the ages, with Maimonides (1135-1204 AD) stating:

"Only if one knows the causes of syncope will he be able to recognise its onset and combat the cause."

Crucifixion is the only method of execution in history which depended on the induction of recurrent syncope to cause death.

Gowers first coined the term "vaso-vagal" (hyphenated) in the early years of the last century [Gowers 1907]. This work addressed various clinical forms of epilepsy and the label was used to describe a symptom complex now more conventionally called anxiety/panic/hyperventilation attack syndrome. Some 25 years later, with due respect and acknowledgement, Sir Thomas Lewis redefined and appropriated Gowers' terminology [Lewis 1932]. In his analysis, Lewis provided crucial insights into the pathophysiology of 'vasovagal syncope' (no hyphen), particularly by pointing out the apparent dissociation between the vasodepressor and cardioinhibitory components (see section 1.1.3.):

"The proof that slowing is vagal is given by atropine...... Slowing....to 50 or 40, exceptionally to 30 beats per minute, is sufficient to induce unconsciousness..... Undoubtedly the main cause of the fall of blood pressure in these attacks, and the enfeeblement or loss of pulse, is independent of the vagus, and lies in the blood vessels. Atropine, while raising the pulse rate up to and beyond normal levels during

the attack, leaves the blood pressure below normal and the patient still pale and not fully conscious."

Lewis' paper proposed the carotid sinus baroreceptors as a possible site of origin of the depressor reflex and recognised that VVS could "occur also in people in apparently robust health". Sharpey-Schafer's classic review [1956] contained most of what is clinically accepted today and proposed the mainstay of current management consisting of advice "in avoiding the circumstances which cause the attack".

1.1.2. Epidemiology

Syncope is common, comprising up to 3% of Accident & Emergency attendances [Day 1982], 1% to 6% of hospital admissions [Kapoor 1991], with up to 30% of the generally population experiencing a syncopal episode during their lifetime [Sutton 1996]. In addition, syncope can be dangerous [Manolis 1990], disabling [Linzer 1991, Silverstein 1982] and difficult to diagnose [Linzer 1997, Kapoor 1983]. Recurrent episodes of syncope, therefore, can have huge physical, psychological and social consequences similar to those resulting from chronic disabling illnesses such as rheumatoid arthritis [Linzer 1991]. The list of causes of syncope is extensive (table 1.1), encompassing neurogenic, cardiac, neurologic, iatrogenic and psychiatric aetiologies. A substantial proportion of cases, however, defy diagnosis [Kapoor 1983, Linzer 1997], though the advent of head-up tilt testing has greatly improved the diagnostic yield [Kenny 1986, Grubb 1991b, Strasburg 1989, Raviele 1989, Abi-Samra 1987].

Vasovagal syncope (VVS) is the commonest form of syncope presenting to medical practitioners [Sutton 1996] and is the focus of attention in this thesis. VVS occurs frequently in young adults but affects all age groups [Simon 2000] and is a significant

cause of morbidity in the elderly [Manolis 1990, Silverstein 1982, Grubb 1996a]. It comprises approximately 58% of all syncopal episodes presenting to emergency rooms [Sutton 1996] with an equal sex incidence [Simon 2000]. There is frequently a family history of the disorder (~37%), especially if the onset is below the age of 20 years [Mathias 1998]. The syndrome can be induced in everyone if a big enough stressor is applied [el-Bedawi 1994a], implying that lasting pathology need not necessarily be present for VVS to occur.

1.1.3. Pathophysiology

1.1.3.1. Normal physiological responses to adoption of upright posture

Adoption of the upright posture in humans causes a gravity-mediated increase of ~500-700ml of venous pooling in the lower limbs and results in a decrease in cardiac filling pressure, stroke volume and arterial BP [Hellebrandt 1943]. This is followed by a compensatory reflex increase in heart rate and total peripheral resistance (TPR) through sympathetic activation, withdrawal of parasympathetic tone and release of renin and vasopressin [Hellebrandt 1943, Abboud 1993]. The afferent limb of the reflex consists of arterial baroreceptors and cardiac mechanoreceptors embedded in the aortic arch, carotid sinus and cardiac chambers which are sensitive to mechanical deformation and chemical, paracrine and endocrine factors [Coleridge 1980, Abboud 1986]. A decrease in the stimulation of these sensory receptors leads to an decrease in the frequency of C fibre action potentials transmitted to the brainstem via the glossopharyngeal and vagal nerves. These signals are integrated in the nucleus tractus solitarius and rostral ventro-medial and ventro-lateral areas of the medulla resulting in altered efferent activity to vagal preganglionic nuclei in the medulla, sympathetic

nervous system centres. A subsequent increase in sympathetic nervous system activity and decrease in vagal activity results in reflex tachycardia and peripheral vasoconstriction which reverse the decline in blood pressure and maintain cerebral perfusion [Lagerhof 1951]. The heart has dual sensory and effector roles in this reflex. As a sensory organ, it relays a decrease in afferent activity to the brainstem and, as an effector organ, it uses an increased level of efferent sympathetic activity to increase heart rate and cardiac output. A simultaneous increase in sympathetic and neurohormonal stimulation of the peripheral vasculature also helps to maintain systemic BP through vasoconstriction. The normal physiological response to standing, therefore, consists of an increase in heart rate and diastolic pressure with an unchanged or slightly decreased systolic blood pressure.

1.1.3.2. Pathophysiological cardiovascular changes during pre-syncope

During pre-syncope, an initial gradual decrease in systemic BP and TPR is followed by a precipitous fall in BP, TPR with or without a fall in heart rate (HR) [Hainsworth 1991]. VVS has been classified, according to the BP and HR responses immediately prior to syncope, into three different types - cardioinhibitory, pure vasodepressor and mixed [Sutton 1992, Grubb 1996a]. A descriptive classifiation of any pathophysiology at such an early stage in its understanding is always fraught with difficulties. Nevertheless, the mixed form of VVS has been recognised as the commonest form and tailored treatment regimes have been proposed, with no subsequent validation, on the basis of this classification [Sutton 1992]. As hypotension and/or bradycardia become progressively more profound, CBF is compromised and, irrespective of the precipitant or type of cardiovascular changes,

syncope is ultimately thought to result from neuronal dysfunction due to cerebral hypoperfusion [Van Lieshout 1991, Sung 2000, Grubb 1991a,1998].

The pathophysiology and biochemistry underlying the above changes are a matter of some debate. Anomalies have been demonstrated in many different research areas which contribute to our understanding of the sequence of events but the precise sequence of events leading to VVS has not been fully elucidated. There is, however, general acceptance about events in the terminal stages of the pre-syncopal period. Excessive venous pooling leads to a gradual and then sudden fall in venous return to the heart [Streeten 1988]. The depleted ventricular volume causes excessively vigourous ventricular contraction with stimulation of a large number of mechanoreceptors (or C fibres) and a consequent surge in afferent neural traffic to nucleus tractus solitarius in the medulla oblongata [Samoil 1992, Mark 1983, Van Lieshout 1991]. Increased afferent neural output to the brainstem simulates hypertensive conditions and stimulates an anti-hypertensive homeostatic reflex resulting in a fall in peripheral vascular resistance and/or bradycardia - the so-called Bezold-Jarish reflex mechanism [Mark 1983, Van Lieshout 1991].

1.1.3.3. Venous Pooling & reduced intravascular volume

The physiological tendency to pooling of blood in the compliant veins of the lower limbs during orthostasis, with a consequent reduction in central circulatory volume, may be pathophysiologically increased in patients with recurrent VVS [Streeten 1988, Yamanouchi 1996, Hainsworth 1992], possibly due to a lack of variation in venous tone [Hargreaves 1992]. A highly significant correlation has been demonstrated between orthostatic tolerance and both plasma and blood volume [El-Sayed 1995] and salt loading has been shown to increase both plasma volume and tolerance to

orthostatic stress in certain subjects [El-Sayed 1996]. Similarly, a moderate exercise program has been shown to significantly increase plasma and blood volumes as well as improving symptoms and orthostatic tolerance [Mtinangi 1998,1999].

1.1.3.4. Ventricular volume and contraction

Increased venous pooling is hypothesised to result in a progressively smaller left ventricular volume, increased cardiac inotropy and consequent excitation of ventricular C fibre mechanoreceptors. In support of this hypothesis, the reduction of stroke index, ejection fraction, and the rate of reduction of end-diastolic volume indices during graded head-up tilt is larger in patients with VVS than normal subjects [Yamanouchi 1996]. In addition, a reduction in left ventricular volume assessed by echocardiography occurs at the onset of VVS [Fitzpatrick 1992, Mizumaki 1995, Shalev 1991]. Further support comes from reports that negative inotropic agents may attenuate the hypotension and bradycardia of VVS induced by head-up tilt [Rea 1989].

The hypothesis is challenged, however, by the fact that end-systolic left ventricular volume during central hypovolaemia does not surpass ~25%, suggesting that there is no progression to excessive cardiac emptying before the onset of syncope [Shalev 1991]. In addition, cardiac contractility as measured by fractional ventricular shortening does not increase during pre-syncope [Novak 1996]. Force imposed on the left ventricular wall and mechanoreceptors may be more accurately assessed, however, by left ventricular wall stress than by systolic fibre shortening. End systolic stress has recently been shown to be reduced during HUT in patients susceptible to VVS, suggesting that, if paradoxical ventricular mechanoreceptor activation occurs

during pre-syncope, it is not due to left ventricular hypercontractility or increased systolic wall stress [Liu 2000].

1.1.3.5. Central processing & neurotransmitters

Afferent cardiovascular baroreceptor and chemoreceptor C fibre nerve traffic arrives at the medulla via the vagus and glossopharyngeal nerves and is integrated in the nucleus tractus solitarius which also receives afferent impulses from other cranial nerves, hypothalamus, spinal cord and brainstem [Pelletier 1973, Donald 1978]. Signals from the nucleus tractus solitarius address the vagal preganglionic nuclei in the medulla, sympathetic preganglionic nuclei in the in the spinal cord, other brainstem nuclei and higher central nervous system centres. Two extreme haemodynamic reactions in response to afferent impulses to the nucleus tractus solitarius are recognised. The 'defence' or 'fight or flight' reaction is characterised by somatomotor and sympathetic neural activation while the alternative response, possibly mediated by activation of the cingulate gyrus by the nucleus tractus solitarius and/or somatic and visceral afferent signals, results in vagally mediated bradycardia, sympathetic withdrawal and diminished muscular tone [Benditt 1997]. The outcome may be of sufficient severity to elicit cerebral hypoperfusion and syncope and may be comparable to the 'playing dead' strategy of some animals. Electrical stimulation of the "defence area" in the hypothalamus of anaesthetised cats results in increases in heart rate and BP and vasodilatation in skeletal muscle [Eliasson 1951]. Emotional stress in humans can also result in tachycardia, hypertension and skeletal muscle vasodilatation [Barcroft 1960]. In view of these findings, and the fact that VVS may be precipitated by anxiety, physical pain, mental anguish, emotional distress and even the anticipation of physical pain or trauma, has led to the belief that the

pathophysiology of VVS may be, at least in part, centrally controlled [Hainsworth 1992].

Certain neurotransmitters may play a role in eliciting or facilitating the hypotension and bradycardia of VVS. Serotonin (5-hydroxytryptamine) may be involved in a sympathoinhibitory pathway in the central nervous system that results in hypotension and bradycardia during acute central hypovolaemia [Morgan 1988]. Depletion of serotonin stores attenuates the BP and HR decreases in response to acute blood loss in conscious mammals, a central hypovolaemic state similar to that prior to VVS [Elam 1985]. In addition, the serotonin receptor blocker, methylsergide, induces a marked pressor response during acute haemorrhage [Elam 1985]. Central intracerebroventricular serotonin has been shown to induce systemic hypotension, excitation of adrenal sympathetic nerve activity and inhibition of renal sympathetic nerve activity in rats [Morgan 1988, Abboud 1993]. The increase in adrenal sympathetic nerve activity can be abolished by the administration of parachlorophenylalanine, an inhibitor of serotonin synthesis [Morgan 1988]. Activation of cerebral serotonin receptors has been shown to inhibit sympathetic nervous system activity and to facilitate a vasodepressor response [Kosinski 1994, Gonzalez-Heydrich 1990]. The hypothesis that serotonin plays a role in the pathophysiology of VVS is strengthened by the demonstration that the serotonin reuptake inhibitor, fluoxetine hydrochloride, may be effective in preventing both spontaneous and HUT-induced VVS [Grubb 1993a].

The discovery of an increase in plasma β endorphin levels in humans during syncope [Perna 1990] has raised questions about the role of endogenous opiates in VVS. Endogenous opoids have a tonic inhibitory effect on sympathetic tone [van Lieshout 1997]. Naloxone, an opiate receptor antagonist, enhances the cardiopulmonary

baroreflex excitation of sympathetic activity [Schobel 1992] while opiate receptor agonists inhibit renal sympathetic nerve activity during hypotensive haemorrhage in conscious rabbits [Morita 1988]. In addition, in an experimental haemorrhagic model in rabbits, intracisternal administration of naloxone was more effective than intravenous administration of the drug in preventing hypotension [Evans 1989]. However, parenteral naloxone, even in very high doses, has failed to prevent induction of VVS during lower body negative pressure studies in humans [Smith 1993].

Increased levels of a number of other centrally released vasoactive agents have been associated with VVS. Pancreatic polypeptide is closely associated with, and is often considered a marker of, parasympathetic neural activity and has been found to be elevated during pre-syncope [Sander-Jensen 1985, 1986]. Vasopressin levels are also increased in the pre-syncopal period, a finding attributable to neural connections from the nucleus tractus solitarius [Fitzpatrick 1990]. The role of nitric oxide (NO) is discussed in section 1.1.3.11.

1.1.3.6. Autonomic function

Bradycardia during VVS is thought to be due to an increase in parasympathetic nerve activity as heart rate can be maintained by the administration of atropine [Lewis 1932]. In addition, failure to decrease parasympathetic tone during upright tilt may be a feature of patients with recurrent VVS [Lippman 1995]. The presumption of an increase in parasympathetic tone during the pre-syncopal period [Abboud 1993] has been challenged, however, with a number of investigators failing to find an increase in high frequency power that would support this view [Kouakam 1999, Morillo 1994, Sneddon 1993, Prinz-Zaiss 1995].

Vasodilatation and the subsequent decline in peripheral resistance and systemic BP are thought to be primarily the result of diminished sympathetic vasoconstrictor tone. Microneurographic recordings have illustrated sudden sympathetic withdrawal from skeletal muscle prior to VVS [Wallin 1982, Van Lieshout 1991, Chosy 1965, Bie 1986], a finding supported by the disappearance of muscle sympathetic nerve activity (MSNA) assessed using spectral analysis [Mosqueda-Garcia 1997, Morillo 1997]. Patients with recurrent VVS display blunted increases in MSNA at low levels of headup tilt [Mosqueda-Garcia 1997] and a subtle though significant decline MSNA has been demonstrated even before the development of pre-syncopal symptoms [Morillo 1997]. Studies of forearm muscle blood flow in fainting humans has demonstrated vasodilatation to be confined to muscle [Roddie 1977, Barcroft 1944,1945]. Spectral analysis of heart rate variability during HUT in normal subjects has demonstrated an increase in low frequency power that has been interpreted as mainly indicating an increase in sympathetic tone [Pagani 1986, Vybiral 1989, Hayono 1991, Montano 1994]. Similar testing in patients prone to VVS has yielded conflicting results with both normal increases and abnormal decreases in LF power being reported [Theodorakis 1992, Pruvot 1994, Morillo 1994, Lippman 1995, Prinz-Zaiss 1995, Boulos 1996, Kochiadakis 1997]. Most of these studies have examined responses during late pre-syncope, but similar methods have also demonstrated inadequate sympathovagal balance immediately after head-up tilt in subjects who subsequently develop VVS [Morillo 1994, Kochiadakis 1997, Kouakam 1999]. A decrease in pulse interval low frequency to high frequency ratio (LF:HF) after five minutes of head-up tilt has a 92% positive and 86% negative predictive value for the occurrence of VVS [Kouakam1999]. A recent review, however, suggested that the

use of pulse interval LF:HF ratio as a measure of sympathovagal balance may have limitations [Eckberg 1997].

Evidence for passive vasodilatation has also been found in disproportionately low plasma noradrenaline levels in pre-syncopal humans with severe hypotension [Eckberg 1986]. Little evidence exists for an active vasodilator mechanism during pre-syncope in humans [Van Lieshout 1997].

More recently, Furlan et al [1998] used spectral analysis of heart rate variability to propose two different sympathetic patterns prior to HUT-induced VVS in normal subjects. One syncopal group demonstrated continued sympathetic activation immediately prior to syncope while a second group exhibited a slow inversion of the cardiac sympathovagal balance with progressive sympathetic inhibition. BP variability was not assessed in Furlan's study, however, and the findings may not be similar in patients with recurrent VVS.

Release of acetylcholine by the parasympathetic nervous system is known to exert a feedback inhibitory effect on noradrenaline release from nearby sympathetic nerve endings [Muscholl 1980, Levy 1984, Benditt 1997]. Conversely, release of noradrenaline and neuropeptide-Y from sympathetic nerve endings inhibits synaptic acetylcholine release [Potter 1985, 1987]. The temporal sequence, therefore, as well as the magnitude of sympathetic and parasympathetic activation may have important consequences in terms of the severity of vasodilation or the degree of bradycardia associated with pre-syncope.

1.1.3.7. Baroreceptor sensitivity

As early as 1932, Lewis proposed abnormalities in cardiopulmonary baroreceptor sensitivity (BRS) as a possible explanation for recurrent VVS. Cardiac BRS

decreases in normal subjects during head-up tilt [Cooke 1999], but a number of investigators have found significantly lower cardiac BRS during head-up tilt and after pharmacological testing in patients with recurrent VVS [Mosqueda- Garcia 1997, Haruhiko 1996, Morillo 1994]. Thomson et al [1997] displayed impaired forearm vasoconstriction and paradoxical vasodilation during the application of subhypotensive lower body negative pressure (LBNP) in patients with tilt-tablepositive VVS compared with age-matched normal control subjects, suggesting possible reduced activation of cardiopulmonary baroreceptors. Morillo [1997] found no differences in vagal baroreflex responses between normal subjects and patients with VVS during supine rest but found impaired responses to pharmacologically induced arterial pressure changes below resting levels. Other studies have shown abnormal forearm resistance vessel vasodilator responses and impaired splenic vasoconstriction during dynamic leg exercise in patients with VVS, which may be the result of exaggerated cardiopulmonary mechanoreceptor activation on exercise [Thomson 1995, 1996, Sneddon 1994]. Overall these observations suggest that patients with VVS may have abnormal cardiac baroreflex responses during both central volume unloading and exercise. Nevertheless, patients in these studies tended to be young and no data are currently available on cardiac BRS in older patients with VVS.

1.1.3.8. Plasma catecholamines

A number of investigators have demonstrated a marked increase in plasma adrenaline levels prior to both spontaneous and induced VVS [Sra 1994, Sander-Jensen 1986, Abi-Samra 1988, Chosy 1965, Hackel 1991, Fitzpatrick 1992] with the increase antedating evident changes in BP and heart rate [Benditt 1997]. Changes in plasma

noradrenaline levels during pre-syncope are less clear with increases [Vingerhoets 1984, Chosy 1965] and no increases [Sander-Jensen 1986, Abi-Samra 1988] being reported. Adrenaline produces β_2 -adrenergic dilatation in both skeletal muscle and splanchnic resistance vessels at concentrations measured in humans under stress [Rowell 1989, Shepherd 1979]. Several studies have used intravenous administration of the non-selective β adrenergic agonist isoproterenol to enhance susceptibility to VVS [Grubb 1991b, Kapoor 1994], while propranolol is known to reduce the firing of cardiac mechanoreceptors with vagal afferents [Thames 1980]. These findings suggest that both endogenous and exogenous increases in catecholamines may exacerbate a tendency to VVS through both arterial vasodilatation and the Bezold-Jarish reflex mechanism [Mark 1983]. This is the rationale behind the widespread therapeutic use of β adrenergic blockers in patients with recurrent VVS [Abboud 1993, Almquist 1989], with some evidence that beneficial effects may be mediated by β_1 rather that β_2 adrenergic receptors [Waxman 1994].

1.1.3.9. Cerebral autoregulation

The advent of transcranial Doppler ultrasonography (TCD) has led to an explosion of interest in cerebral haemodynamics prior to VVS. Irrespective of the pathophysiology involved in VVS, it is thought that the loss of consciousness during syncope is caused by cerebral hypoperfusion [Van Lieshout 1991, Yonehara 1994, Mattle 1995]. Reductions in mean and diastolic cerebral blood flow velocity (CBFV), a surrogate for cerebral blood flow (see section 2.1.5.), with relative preservation of systolic CBFV have been consistently reported during pre-syncope [Schondorf 1997, Sung 2000, Grubb 1991a, 1998, Janosik 1992]. These findings led to the hypotheses that an
impairment of cerebral autoregulation (CA) may be contributing to the symptoms of VVS and that impaired CA may, in fact, be the primary problem in patients with recurrent VVS. A number of investigators have added weight to this hypothesis by using TCD with either head-up tilt or incremental lower body negative pressure to apparently show paradoxical cerebral vasoconstriction prior to syncope [Janosik 1992, Bondar 1995, Sung 2000, Grubb 1991a, 1998, Levine 1994, Giller 1992]. Indeed, a number of authors have described a form of 'cerebral syncope' whereby syncope, occasionally confirmed by electroencephalography, has occurred in the presence of apparent cerebral vasoconstriction but in the absence of systemic hypotension or bradycardia [Gomez 1999, Njemanze 1993, Daffertshafer 1995, Grubb 1998, Fredman 1995]. Most of these studies have, in the main, used Gosling's pulsatility index (GPI) as a marker for rises in cerebrovascular resistance (CVR) (see section 2.1.5.4.). As arterial BP is not taken into account in its calculation, GPI may, however, be poorly applicable to the cerebral circulation [Czosnyka 1996b, Schondorf 1997, Aaslid 1992]. More recently, others have used Poiseuille's Law to suggest that classical CVR (CCVR) falls prior to syncope and that CA is, therefore, preserved in VVS [Schondorf 1997]. Neither GPI nor CCVR, however, take into account the critical closing pressure (CrCP) of the cerebral circulation, i.e. the pressure below which flow in a blood vessel ceases, which can have a profound effect on cerebral blood flow [Dewey 1974, Panerai 1995, Dawson 1999a]. Resistance area product (RAP) is an index of CVR that takes CrCP into account (see section 2.1.5.4.) and may, therefore, be more appropriate for use when analysing the cerebral circulation [Panerai 1993, Evans 1988, Carey 2000].

1.1.3.10. Respiration

Respiratory patterns prior to VVS have been a matter of some debate. Lipsitz [1998] recently described rather convincingly an increase in respiratory amplitude during pre-syncope with no change in respiratory rate. Schondorf [1997] similarly reported no change in respiratory rate during pre-syncope but tidal volume was not measured. An increase in respiratory amplitude during pre-syncope is unsurprising, as an increase in respiratory depth is a well-known mechanism that results in an increase in BP through venoconstriction and a mechanical increase in preload [Shepherd 1981]. Gilliat [1948] has also reported a spinal vasoconstrictor reflex activated by deep inspiration which may lead to increases in systemic BP. Respiratory changes during pre-syncope may, therefore, be secondary to an established pathophysiological process but it is possible that they may also play a primary role in the process. In spite of the reported increase in respiratory amplitude and probable subsequent hyperventilation, carbon dioxide levels have not yet been assessed during presyncope. This is surprising considering the known marked effects of carbon dioxide on CBF (see section 1.3.3.1) which could account for the proposed cerebral vasoconstriction prior to syncope.

1.1.3.11. Endothelial factors

The role of endothelial factors in the pathphysiology of VVS is attracting increasing attention. Nitric oxide (NO) acts not only as an endothelial mediator of vascular smooth muscle relaxation, but it is now well established that NO is also a modulator of sympathetic and parasympathetic nervous system activity in animals [Chowdhary 1999]. Histochemical staining techniques have identified neuronal populations that contain NO synthetase within medullary cardio-regulatory sites and their peripheral autonomic pathways [Zanzinger 1997, Dun 1994, Tanaka 1994, Klimaschewski 1992,

Anderson 1993, Chowdhary 1999]. NO has both pre- and post-ganglionic sympathetic nerve activity and may attenuate cardiovascular end-organ response to sympathetic stimulation [Chowdhary 1999]. Central attenuation of sympathetic outflow and baroreflex-mediated vagal control by NO has also been described [Chowdhary 1999, Sakuma 1992].

There is a scarcity, however, of information regarding the role of NO in the regulation of human cardiovascular autonomic control. Two groups of investigators used NO synthetase inhibitors to suggest that, in agreement with animal studies, NO significantly modulates sympathetic nerve activity in humans [Lepori 1998, Owlya 1997]. In addition, a NO synthetase inhibitor significantly potentiated the cardiac contractile response to dobutamine in patients with heart failure, but not in patients with normal cardiac function [Hare 1998].

There is even less information regarding possible endothelial dysfunction during VVS in humans. Kaufmann et al [1993] demonstrated a rise in NO activity during neurally mediated syncope which has the potential to explain the withdrawal of sympathetic activity prior to VVS discussed earlier. Others previously showed that HUT induces the release of endothelin-1 into plasma, probably from the neurohypophysis, and hypothesised that impaired endothelin-1 release could account for failure to maintain vascular tone during orthostasis [Matzen 1992, Kaufmann 1991].

In summary, the pathophysiology of VVS is complex and unclear and may involve autonomic, central, neurohormonal, humeral, venous, endothelial, plasma volume, respiratory and cerebrovascular mechanisms.

1.1.4. Clinical characteristics

VVS most commonly occurs in the upright position but may also occur when sitting [Thilenius 1992] and when supine [Verrill 1970]. Patients often report situations or events likely to trigger an event such as standing for long periods, warm environments, anxiety, physical pain, mental anguish, emotional distress and even the anticipation of physical pain or trauma. In most patients, the occurrence of VVS is usually preceded by a prodrome which is highly variable in duration, lasting anywhere from several seconds to several minutes [Lewis 1932, Wayne 1961]. Typical symptoms include dizziness, vertigo, diaphoresis, nausea, disturbances of vision, fatigue, weakness, headache, paraesthesiae, abdominal discomfort and visual and auditory hallucinations [Lewis 1932, Van Lieshout 1991]. Observers often report patients as becoming pale and diaphoretic with cold skin and dilated pupils [Wayne 1961].

Symptoms during this pre-syncopal period are often recognised by patients and offer the opportunity to take evasive action to prevent loss of consciousness ensuing. If evasive action is not taken, however, a variable period of loss of consciousness intervenes with diffuse loss of muscle tone and collapse. Urinary and faecal incontinence are not unknown and tonic-clonic movements due to transient cerebral hypoxia may lead to confusion with epilepsy. The post-syncopal period is often characterised by dizziness, nausea, headache and a general malaise and, once again, varies substantially in duration [Wayne 1961]. Many patients with recurrent VVS report a relapsing and remitting course [Grubb 1996a].

Many elderly patients have little or no prodrome prior to VVS leading to an increased risk of morbidity due to failure to take evasive action [Fitzpatrick 1991a, Grubb 1996a, Manolis 1990, Linzer 1991, Silverstein 1982]. Older patients may often display atypical features during VVS that pose diagnostic dilemmas. Confusion and

disorientation may occur during both the prodromal and recovery periods and may be particularly prolonged after the event [Grubb 1996a]. Focal neurological deficits, such as dysarthria, may lead to an incorrect diagnosis of a transient ischaemic attack, especially if hypotension and bradycardia are not sufficiently profound to induce loss of consciousness [Grubb 1993b]. Post-prandial VVS is not uncommon in the elderly, probably exacerbated by sequestration of blood by the mesenteric circulation [Lipsitz 1986]. This situation may be compounded by the consumption of alcohol with a meal due to an increase in venous pooling secondary to the vasodilatory effects of alcohol [Lipsitz 1986]. In addition, comorbidity and polypharmacy may complicate the diagnosis and treatment of an older patient [Linzer 1997].

1.1.5. Investigations

Investigators in the 1980's demonstrated that the cause of syncope could not be established in up to 45% of cases [Mathias 2001, Kapoor 1991]. It was hypothesised that a large proportion of these patients had VVS, but no confirmatory test existed. In the past, therefore, VVS was a diagnosis of exclusion with other causes of syncope (Table 1.1) excluded using a variety of investigations (Table 1.2). These investigations have a low diagnostic yield, but, in combination with careful history taking and physical examination, form the mainstay of diagnostic testing in patients presenting with syncope [Mathias 2001, Linzer 1997].

Head-up tilt (HUT) table testing has been used as an investigative tool in the pathophysiology of orthostatic stress for over fifty years [Hellebrandt 1943] before the initial demonstration of its utility in the diagnosis of unexplained syncope [Kenny 1986]. Since then, a number of investigators have confirmed its usefulness as a method of unmasking susceptibility to VVS and related disorders [Fitzpatrick 1989,

Grubb 1991b, Strasberg 1989, Raviele 1989, Abi-Samra 1988]. BP, HR and catecholamine changes and prodromal symptoms experienced prior to VVS induced by HUT are virtually identical to those experienced prior to spontaneous episodes [Grubb 1996a]. HUT was initially validated in younger patients but has recently been validated as a useful diagnostic test in patients ≥ 65 years [Bloomfield 1999]. The usefulness of the HUT test in clinical practice and as a research tool has been limited, however, by a lack of standard methodology [Kenny 2000, Kapoor 1994].

1.1.5.1. Normal cardiovascular, respiratory & cerebrovascular responses to HUT

Right atrial pressure falls immediately after HUT from 4-6 mmHg to 0-2 mmHg [Bridgden 1950] but a reduction in left ventricular stroke volume (SV) does not take place until after about six beats, possibly due to the reservoir of blood available in the lungs and heart [Wieling 1993]. A temporary small decline occurs in arterial BP, with a minimum pressure at around 7 seconds [Borst 1984, Wieling 1993], which may be greater after a long period of supine rest prior to HUT and is possibly attributable to an increase in the unstretched volume of the veins [Borst 1984]. An increase in systemic vascular resistance does not occur until about 6 seconds after HUT, but a subsequent increase of about 40% results in a net rise in mean arterial BP of approximately 10mmHg after 30 seconds. SV falls to a stable level of about 70% of supine after 30 seconds, by which time heart rate has increased by approximately 10 beats. A fast (1.5 seconds) speed of tilting evokes similar haemodynamic effects as the most commonly used speed of tilting (3 seconds) [Ten Harkel 1990] but much slower speeds of tilting may attenuate these responses [Smith 1994]. After about 5 minutes of HUT, increases have occurred in DBP (~10mmHg), MAP (5-10%), heart rate (~20-30%) and systemic vascular resistance (~30-40%) while

decreases have occurred in intrathoracic blood volume (~20-30%), SV (~30-45%) and cardiac output (~15-20%) [Wieling 1993]. In contrast, SBP changes very little. The decrease in intrathoracic blood volume is mostly accounted for by the accumulation of 500-1000ml of blood in the leg veins during orthostasis.

CBFV has been shown to fall significantly during orthostatic stress with HUT and lower body negative pressure (LBNP) in normal subjects [Bondar 1995, Giller 1992, Levine 1994, Schondorf 1997, Stoll 1999, Cencetti 1997]. Arterial, end-tidal and transcutaneous carbon dioxide (CO_2) levels have also been shown to fall significantly in normal subjects after HUT [Cencetti 1997, Serrador 1998, Anthonisen 1965, Bjurstedt 1962, Yoshizaki 1998] and this is probably due to an increase in tidal volume without any change in respiratory frequency [Cencetti 1997, Yoshizaki 1998]. Changes in CO_2 are well known to affect CBFV [Garnham 1999, Panerai 1999b] and Cencetti et al [1997] have shown a significant link between the declines in CBFV and CO_2 after HUT.

1.1.5.2. Indications, contraindications and risks of HUT testing

Any patient with recurrent syncope unexplained by history, physical examination and the neurological and cardiovascular investigations outlined above should be considered for HUT testing [Kenny 2000, Grubb 1996a]. Tilt table testing may also be useful for investigating elderly patients with recurrent unexplained falls or transient ischaemic attack-like symptoms [Grubb 1996a, Grubb 1993b]. A role for testing has also been proposed in the investigation of patients with recurrent, treatment-resistant unexplained seizures [Grubb 1996a, Zaidi 1998], recurrent vertigo [Grubb 1992c] and high risk patients with a single unexplained syncopal episode [Kenny 2000].

Caution should be exercised when subjecting patients with severe left ventricular outflow obstruction, mitral stenosis, coronary artery disease and carotid stenosis to HUT [Kenny 2000]. In spite of the marked BP and HR changes induced during HUT, very few adverse events and no fatalities have been attributed to the test in the literature [Kapoor 1994, Gatzoulis 1995] although many centres undoubtedly have anecdotal reports of same [Kenny 2000].

1.1.5.3. Equipment

Tilt tables should be of the foot-plate support type as physiological responses may differ with the use of saddle tables [Kenny 2000, 1996]. They are generally padded and calibrated to allow accurate placement at a chosen angle. Both manually operated and electrically automated tables are available but each should allow rapid changes of posture to both the tilted and supine positions. HR is monitored, usually continuously, with standard three-lead electrocardiography. It is recommended that BP is monitored on a beat-to-beat basis as sphygmomanometry is insufficient to assess rapid changes in arterial BP [Kenny 2000]. Non-invasive devices such as the Finapres (see section 2.2) and Portapres are ideal and obviate the need for invasive intra-arterial monitoring. Future classifications and treatments of VVS may be based on continuous BP assessment and sphygmomanometry must now be considered inadequate for use during HUT [Kenny 2000]. Intravenous cannulation may affect the specificity of the test [McIntosh 1994] and is, therefore, not routinely recommended, but advanced cardiac life support equipment should be on constant standby [Kenny 2000].

1.1.5.4. Prior instructions & environment

HUT was often initially performed in the fasting state, though this was inconsistently included as part of the protocol [Kapoor 1994]. It is now recognised that, in order to avoid dehydration and consequent hypotension, especially in older subjects, affecting the test result, patients should be fasted for no more than two hours before the procedure [Benditt 1996, Victor 1996, Kenny 2000]. Drugs affecting the cardiovascular and autonomic systems or that may affect intravascular volume are generally discontinued a minimum of five half-lives before HUT [Victor 1996], unless implicated in symptomatology, when testing is usually performed on medications [Kenny 2000]. A quiet, dimly lit, temperature controlled environment (20-23°C) is required to minimise autonomic stimulation. The test is usually supervised by a physician and nurse or cardiovascular technician familiar with the test protocol and complications [Kenny 2000]. After explanation of the protocol and an instruction to minimise lower limb movement to maximise venous pooling during tilt [Benditt 1996, Victor 1996], a period of supine rest of between 20-45 minutes is normally allowed prior to HUT [Kapoor 1994, Fitzpatrick 1991].

1.1.5.5. Tilt angle and duration

The angle and duration of HUT are the two primary determinants of the number of positive tests yielded by a particular protocol [Benditt 1996, Kapoor 1992, Mansourati 1996]. Angles of between 40° and 90° have been used, but the commonest angle employed is 60° [Kapoor 1994]. Angles < 60° produce significantly less haemodynamic effects than steeper angles, but increasing the tilt angle beyond 60° produces no apparent additional effect on cardiac output or sympathetic tone [Zaidi 2000]. Despite these findings, there have been suggestions that higher tilt angles may increase the yield of positive tests [Mansourati 1996, Kapoor 1994].

The duration of passive HUT, where pharmacological provocative agents are not used, varies widely from centre to centre ranging from 10 to 60 minutes [Kapoor 1994, Grubb 1996a, Fitzpatrick 1991, Kenny 2000]. The optimum duration of passive HUT is in doubt, but there may be a relationship between duration and the number of positive results [Kapoor 1994], probably due to a reduction in the number of false negative results [Kapoor 1994, Fitzpatrick 1991, Kenny 2000]. One study reported a mean time to syncope of 24 minutes [Fitzpatrick 1991], and if a normal distribution and 2 standard deviations are assumed, a HUT duration of between 30-45 minutes seems reasonable [Kenny 2000].

1.1.5.6. Provocation during HUT testing

A number of centres introduced provocative agents during HUT in an attempt to increase the sensitivity and shorten the duration of the test.

1.1.5.6.1. Isoproterenol

The most commonly used provocative agent is the non-selective β adrenergic agonist isoproterenol which is administered intravenously either by continuous or bolus infusion [Graham 2001, Almquist 1989, Waxman 1989, Grubb 1991b, Fitzpatrick 1996, Carlioz 1997, Kapoor 1992, Chen 1992]. Again, no standard protocol exists and the drug is widely administered at differing doses and at different times after HUT [Kapoor 1994, Kenny 2000]. It is thought that isproterenol shortens the time to syncope by augmenting the Bezold-Jarish reflex by stimulating cardiac β_1 adrenergic receptors but a central mechanism cannot be fully discounted [Mark 1983, Waxman 1994]. A comprehensive literature review of tilt table tests up to 1994 found that the

use of intravenous isoproterenol increased the yield of positive results by a factor of approximately 33% over passive HUT [Kapoor 1994]. The increase in positive results may have been attributable to higher tilt angles or intravenous cannulation rather than the drug itself, as studies performed at 60° demonstrated identical percentages of positive responses whether isoproterenol was used or not [Kapoor 1994]. Neither could any relationship be shown between the maximal dose of isoproterenol and the percentage of positive responses, but higher doses of isoproterenol are not recommended due to concerns about adverse effects on specificity [Kapoor 1992, Carlioz 1997, Fitzpatrick 1996, Kenny 2000]. Surprisingly, a literature review found no evidence for the generally accepted assumption that isoproterenol shortens the time to a positive response [Kapoor 1994]. Contraindications to the use of isoproterenol include ischaemic heart disease and left ventricular outflow obstruction, while known dysrhythmias are a relative contraindication. Side effects necessitating discontinuation are commoner in the elderly and include arrhythmias, chest pain, severe tremulousness and vomiting.

1.1.5.6.2. Sublingual nitroglycerin

More recently, sublingual nitroglycerin has emerged as a possibly more user-friendly provocative agent for use with HUT [Del Rosso 1999, Raviele 1995, Kurbaan 1999, Natale 1998]. It has been found to be as effective as intravenous infusions of isoproterenol for provoking VVS, but has a lower incidence of adverse events [Graham 2001]. Nitroglycerin is thought to act as a provocative agent by causing vasodilation of capacitance vessels [Mason 1971] thereby significantly increasing venous pooling [Raviele 1995]. Due to ease of administration, sublingual administration superceded earlier attempts to devise protocols using intravenous

methods [Raviele 1994]. A variety of doses have been used but, due to variable absorption and linear pharmacokinetics, metered doses of between 400-800 µg are currently recommended [Kenny 2000, Graham 2001].

1.1.5.6.3. Other pharmaceutical agents

A number of other pharmacological provocative agents have been tried with varying degrees of success, including sublingual isosorbide dinitrate [Zeng 1999], intravenous isosorbide dinitrate [Aerts 1999], edrophonium [Lurie 1993], clomipramine [Flevari 1998] and adenosine [Brignole 2000, Shen 1996, Perez-Paredes 1998]. More work is needed using these agents, however before they can be recommended for more general use.

1.1.5.6.4. Lower body negative pressure

Others have combined HUT with lower body negative pressure to provide a more physiological increase in orthostatic stress [El-Bedawi 1994]. This method has been shown to have high reproducibility and specificity [Hainsworth 1994] but is not yet in widespread use.

1.1.5.7. Positivity criteria

Since the advent of HUT as a clinical tool, a wide variety of criteria have been proposed for deeming positivity [Kapoor 1994]. A test is normally described as positive if a patient's original symptoms are reproduced in association with hypotension and/or bradycardia. The occurrence of hypotension and/or bradycardia in the absence of symptom reproduction is not currently regarded as a positive test [Kenny 2000]. The degree of hypotension and bradycardia required for a positive test are again a matter of some debate and have varied with the use of provocative agents [Kapoor 1994], but a classification of VVS has been proposed according to BP and heart rate responses during HUT-induced pre-syncope (Table 1.3.) [Sutton 1992]. A positive response is usually described as cardioinhibitory if a marked bradycardia (\leq 40 bpm for > 10 seconds) or prolonged asystole (>3 seconds) occur in association with symptoms [Sutton 1992]. A positive response is generally defined as vasodepressor if symptoms occur with SBP < 80mmHg and HR maintained at a level no more than 10% below its peak value [Sutton 1992]. A positive response is normally regarded as mixed if SBP is <80mmHg and HR does not fall below 40bpm, falls below 40bpm for < 10 seconds or asystole of <3 seconds occurs in association with symptoms [Sutton 1992]. However, as mentioned above, these are descriptive classifications of a syndrome that is incompletely understood and are likely to undergo refinement as the precise pathophysiology becomes clearer.

1.1.5.8. Sensitivity, specificity & reproducibility

The sensitivity of HUT testing is difficult to elucidate as no 'gold standard' currently exists for the diagnosis of VVS with which HUT testing can be compared. Although the percentage of positive responses in patients with syncope of unknown origin is often referred to by the term 'sensitivity', this is incorrect due to the lack of an adequate reference standard. Approximately 50% of patients with unexplained syncope have a positive response to passive HUT testing but rates of between 26% and 90% have been reported [Kapoor 1994]. As mentioned above, higher tilt angles and an increased duration of testing may increase the yield of positive results [Kapoor 1994]. Isoproterenol appears to increase to number of positive responses to about 62% (range 39% to 87%), but assessment of the effect of isoproterenol is confounded

by the use of a higher tilt angle in most isoproterenol studies [Kapoor 1994]. Sublingual nitroglycerin has been reported to increase the positive yield from 34% to 73% [Kurbaan 1999], 25% to 51% [Raviele 1995], 45% to 79% [Natale 1998] and 11% to 59% [Del Rosso 1999].

The specificity of HUT results have been evaluated by performing the test in control subjects with no previous history of syncope. The situation is complicated by the fact that control subjects may have a predisposition to VVS but may not become symptomatic prior to enrolment in such studies [Kapoor 1994]. Specificity rates of between 0% and 100% have been reported for passive HUT testing, but the overall rate is approximately 90% [Kapoor 1994]. Specificity declines as longer durations of testing and higher doses of isoproterenol are used [Linzer 1997, Kapoor 1994]. Overall specificity of HUT testing with isoproterenol is approximately 75% (range 35% to 100%), with specificity worst in younger patients [Kapoor 1994]. The use of sublingual nitroglycerin with HUT testing has been reported to have a specificity of 94% [Raviele 1995, Del Rosso 1999] and does not appear to decrease specificity compared with a passive HUT test in older patients [Natale 1998]. False positive responses to HUT occur more commonly in younger subjects, possibly due to the age related decline in cardiopulmonary BRS [Grubb 1992b].

Reproducibility of the HUT test is a matter of importance with regard to assessing treatment regimes, but reproducibility studies are confounded by the widely differing protocols in existence. Nevertheless, negative responses may be more reproducible than positive responses (84% v 55%) when passive HUT is re-performed at 10 days [Ruiz 1996]. Immediate reproducibility of a positive passive test is reported to be approximately 75% [Fish 1992]. Negative HUT tests with isoproterenol have a reproducibility of 85% when performed 1-6 weeks apart, while positive test

reproduction may be as high as 90% [Sheldon 1992]. Immediate reproducibility of a negative isoproterenol HUT test result appears to be higher than a positive result (94% v 57% [de Buitleir 1993] and 100% v 80% [Chen 1992]). A negative HUT test with sublingual nitroglycerin provocation is 83% reproducible at 1-28 days while a positive test is 79% reproducible [Foglia-Manzillo 1999].

1.1.5.9. Testing in older patients

It was originally felt that VVS was an uncommon occurrence in older people, a hypothesis apparently supported by the demonstration of a reduced susceptibility of the elderly to syncope during postural tilt [Lipsitz 1989]. Indeed, physiological responses to HUT in healthy subjects may differ with advancing age. Stroke volume decreases and splanchnic vascular resistance increases to a greater degree than in younger subjects, while the increases in heart rate and forearm vascular resistance may be significantly less [Minson 1999]. Bootsma et al [1995], however, have demonstrated no age-related differences in the HR and BP responses of healthy subjects to HUT.

A number of studies have subsequently suggested that VVS is more frequent in older patients than generally recognised [Hackel 1991, Grubb 1992b,1993b, Marangoni 1996, Brembilla-Perrot 1996] and that HUT testing is a useful diagnostic tool in such patients [Marangoni 1996, McIntosh 1993, Bloomfield 1999, Brembilla-Perrot 1996]. When supervising HUT tests involving older subjects, however, the different symptomatology of the elderly during pre-syncope (see section 1.3.3) must be borne in mind [Grubb 1996a, Grubb 1993b, Lipsitz 1986].

Increasing age may be associated with different BP and HR behaviour prior to HUTinduced VVS, with greater falls in BP and smaller increases in HR being reported [Marangoni 1996].

The use of isoproterenol and sublingual nitroglycerin as provocative agents does not appear to affect the specificity of the HUT test in older age groups [Natale 1998, Bloomfield 1999], but nitroglycerin may yield more positive results than isoproterenol [Natale 1998]. The isoproterenol HUT test has been shown, however, to have excellent immediate reproducibility in older age groups [Kou 1997].

1.1.6. Treatment

Assessment of the efficacy of any intervention used to treat recurrent VVS is fraught with difficulties due to the relapsing and remitting nature of the condition, the fact that any tolerable intervention is unlikely to eliminate all events and symptoms and the variable reproducibility of the HUT test.

The majority of patients need no intervention other than reassurance and education regarding avoidance of precipitating factors and evasive action should pre-syncopal symptoms occur [Benditt 1999]. Discontinuation of medications with antihypertensive effects that may exacerbate vasodepression may improve matters [Benditt 1999]. Should recurrent episodes of VVS affect quality of life, further intervention may be considered including elastic support stockings [Grubb 1996a], fludrocortisone [Benditt 1999, Grubb 1996b, Scott 1995, Balaji 1994], increased dietary salt intake [El-Sayed 1996], β adrenergic blockers [Klingenheben 1999, Mahanonda 1995, Scott 1995, Lurie 1992, Sra 1992], midodrine (an α adrenergic agonist) [Klingenheben 1999, Sra 1997, Grubb 1996, Benditt 1998, Ward 1998],

serotonin reuptake inhibitors [Grubb 1993a, 1994, 1996a, Di Girolamo 1999b] and stress and anxiety management through biofeedback therapy [Benditt 1999, McGrady 1986]. Little benefit appears from using disopyramide [Morillo 1993], transdermal scopolamine [Lee 1996, Benditt 1999], ephedrine, dihydroergotamine [Benditt 1999, Grubb 1996b, Deal 1997], etilephrine [Raviele 1999], or theophyllines [Nelson 1991]. Permanent pacemakers with rate hysteresis have recently been shown to be useful in patients with the cardioinhibitory form of the syndrome [Ammirati 1998, Connolly 1999] and moderate exercise [Mtinangi 1998, 1999] and tilt training programs [Ector 1998, Di Girolamo 1999a] may be useful in patients with resistant forms of VVS.

A number of studies have demonstrated a progressive amelioration of symptoms related to VVS over time [Brignole 1992, Natale 1996, Morillo 1993]. It is unclear, however, if this is the natural history of the disorder or related to counselling and reassurance after diagnosis.

1.2. Cerebral blood flow

The human brain has a high metabolic demand for energy and, uniquely in humans, uses glucose as its sole substrate for energy metabolism [Warlow 1996]. Approximately 0.125kg of glucose is metabolised daily in the brain via the glycolytic pathway and tricarboxylic acid cycle [Warlow 1996]. Two molecules of pyruvate are formed from each molecule of glucose by the process of glycolysis. Two glycolytic pathways exist whereby molecules of adenosine triphosphate (ATP) are formed by a series of mitochondrial reactions either in the presence (aerobic glycolysis) or absence (anaerobic glycolysis) of oxygen. Aerobic glycolysis is the more efficient pathway, yielding 36 molecules of ATP from each molecule of glucose, while anaerobic

glycolysis yields only two. Inhibition of pyruvate oxidation during anaerobic glycolysis causes the reduction of pyruvate and the consequent accumulation of intracellular and extracellular lactic acid, inhibiting the ability of mitochondria to sequester calcium. To maintain integrity, neurones in the brain require a constant supply of ATP so that concentrations of the major intracellular cation, potassium, and major extracellular cations, calcium and sodium, are kept constant [Warlow 1996]. As the brain is unable to store energy, a constant supply of oxygenated blood and glucose is essential to preserve function and structural integrity.

The average human brain comprises approximately 2% of total body weight (~1.4 kg in a 65 kg adult) but receives 15-20% of the total cardiac output – a cerebral blood flow (CBF) of around 0.5L/kg/min. This equates to a cerebral metabolic rate of oxygen consumption of approximately 0.33 - 0.35 L/kg/min, or about 0.045 L/min of oxygen – 20% of the total oxygen consumption of the body at rest [Warlow 1996]. CBF and oxygen consumption is higher in people < 20 years and lower for those > 60 years.

In the resting brain, CBF is closely matched with the metabolic demands of cerebral neurones. Grey matter has a higher metabolic rate than white matter and, therefore, has a higher regional CBF. Local CBF may vary by between 10-20% depending on local brain function but, under steady state conditions, global CBF remains essentially constant [Lassen 1977, 1974, Aaslid 1987]. The physiological mechanism behind the close coupling of metabolism with CBF is unknown, but brain activity may produce local metabolites or neuronal discharges affecting the vasoactivity of local blood vessels.

Whole blood viscosity is mainly determined by haematocrit, and CBF has been shown to be inversely proportional to both haematocrit and whole blood viscosity, as well as

fibrinogen levels [Thomas 1982, Ameriso 1990]. It is thought that the inverse relationship of CBF with haematocrit is not due to sluggish blood flow at higher viscosities, but rather that the higher oxygen content of high-haematocrit blood allows the maintenance of oxygen delivery in the face of lower CBF.

CBF is primarily determined by the cerebral perfusion pressure (PP) at the base of the brain and by the cerebrovascular resistance (CVR), which is influenced mainly by the diameter of intracranial arteries as well as blood viscosity. In most body tissues, PP is the difference between the forward pressure exerted by the arterial system and the venous backpressure. This, however, is complicated by the fact that the brain is effectively contained in an enclosed space that retains its own intrinsic pressure, intracranial pressure (ICP). ICP, usually about 8mmHg in a supine adult, affects venous distension and arterial filling and has the effect of limiting CBF. CBF, therefore, can be represented by the equations:

$$CBF = \frac{PP}{CVR} = \frac{MAP - VP - ICP}{CVR}$$

where *MAP* is mean arterial pressure and *VP* is intracranial venous pressure (usually about 10mmHg).

From these equations, it is clear that, as intracranial PP changes, a constant CBF may be maintained by variations in CVR, mainly through alterations in the diameter of small intracranial arterioles or arteries, a process known as cerebral autoregulation.

1.3. Cerebral autoregulation

1.3.1. Historical perspective

Autoregulation refers to the intrinsic ability of an organ to maintain a constant blood flow despite changes in perfusion pressure (PP). Changes in BP and, hence, PP are counteracted by changes in vascular resistance, providing the organ with a constant supply of blood containing oxygen and vital nutrients.

The phoenomenon of autoregulation was first described by Bayliss [1902] with the aid of experiments on isolated canine hindlimbs. Lowering hindlimb PP caused a passive decrease in limb volume but restoration of PP caused an increase in limb volume to levels greater than baseline. The investigators suggested that these findings could be explained by local vasodilatation occurring during the period of hypoperfusion. This hypothesis of the intrinsic ability of an organ to alter vessel diameter was supported by Wacholder [1921] who demonstrated contraction of isolated segments of carotid artery 8-20 seconds after an increase in PP. Rein confirmed the ability of the canine renovascular bed to autoregulate in 1931 by showing preservation of renal blood flow throughout a considerable systemic BP range. The concept of a lower limit of PP, below which the ability of an organ to autoregulate is compromised, was described by Selkurt in 1946 using the canine kidney. As PP fell below 80mmHg, a pressure passive relationship developed whereby blood flow became dependent on PP. Shipley described the upper limit of the autoregulatory process using canine renovascular beds in 1951. Active autoregulation was subsequently demonstrated in a number of other organs including the heart, retina, intestine, liver, skeletal muscle and, perhaps the most studied organ, the brain.

Cerebral autoregulation (CA) was first described by Fog [1937] who demonstrated variations in the directly measured diameter of cat pial arterioles in response to manipulations of systemic BP. Such methods, however, involved long surgical

procedures which damaged the brain, often leading to the finding of impaired or absent CA and stimulating debate as to whether the process actually existed. The nitrous oxide gas inhalation technique was introduced by Kety in 1948 allowing the measurement of CBF under more physiological conditions and, for the first time, in humans. Harper [1965] used this method to demonstrate that CBF did not change significantly in normocapnic dogs when MAP was lowered to 90mmHg. In 1959, Lassen used a number of small studies performed in the early 1950's with Kety's inert gas method to describe cerebral blood flow and oxygen consumption in man. These studies used a variety of patients, including those with essential hypertension and toxaemia of pregnancy, and a variety of methods, including the infusion of pressor and depressor pharmaceutical agents, tilting and spinal anaesthesia, to demonstrate that CA is also active in man. A number of these studies were methodologically flawed, as the effect of essential hypertension, toxaemia and many pharmaceutical pressor and depressor agents on CBF and CA are even still unknown. Nevertheless, these studies suggested that CBF remained constant at approximately 0.5L/kg/min while MAP was within the range 60-150mmHg but that a pressure passive relationship developed outside this range. Using these concepts, Lassen constructed the now classic autoregulatory plateau (Figure 1.1) Some of these original studies combined CBF measurement with arteriovenous oxygen extraction assessment. This technique demonstrated that, although CBF passively followed MAP below the autoregulatory range, brain function was not affected until CBF fell below 0.2L/kg/min, due in the main to increased oxygen extraction by the brain. It soon became clear that, below this critical level, symptoms of cerebral hypoperfusion, such as dizziness, pallor and sweating, occurred and that, unless CBF was rapidly restored to higher levels, cerebral infarction would occur. If

the upper limit of the autoregulatory range was exceeded, a passive increase in CBF with MAP occurred, causing disruption of the blood-brain barrier with consequent brain oedema and impairment of function.

1.3.2. Mechanisms of Cerebral Autoregulation

Changes in MAP can be classified as either static or dynamic [Aaslid 1989, Tiecks 1995]. Static BP changes allow CA to be studied over a prolonged period, and, therefore, allow assessment of the overall efficiency of the system [Paulson 1990]. Dynamic BP changes, however, are followed by restoration of CBF to baseline within 8-20 seconds when CA is intact and such rapid step changes in BP may be used to assess the latency of the CA system [Paulson 1990]. This distinction between cerebral autoregulatory responses as either static or dynamic CA is important because responses may be effected by different physiological mechanisms and affected differently by pathological conditions such as acute ischaemic stroke [Tiecks 1995, Dawson 2000]. Baumbach [1985] also explored the temporal heterogenecity of CA and demonstrated that the blood-brain barrier was more susceptible to hypertensive disruption after rapid, rather than step-wise, increases in systemic BP. Precisely how the brain alters vessel diameter to effect the process of CA remains unclear, but a number of theories have become well established.

1.3.2.1. Tissue perfusion theory

This early theory attempted to explain the phoenomenon of autoregulation through the response of venules to external pressure [Johnson 1964]. It was hypothesised that an increase in MAP, and hence CBF, increased extracellular tissue fluid production which, in turn, encroached on low pressure veins causing an increase in back pressure

to the arterial system, reducing PP and returning CBF to baseline levels. Such a passive mechanism suggests that CA should be immune to external manipulation with drugs, which does not appear to be the case (see section 1.3.3.5.). A number of other findings are also incompatible with the theory. Mathematical modelling has demonstrated that a pressure dependent mechanism could not explain the autoregulatory plateau [Koch 1964]. The theory does not account for Fog's findings of variations in the directly measured diameter of cat pial arterioles [1937] and the role of osmotic pressure on extracellular fluid formation has not been considered. Consequently, the tissue perfusion pressure theory as an underlying mechanism of CA has fallen from favour, but may have some merit in explaining the effects of ICP on PP.

1.3.2.2. Metabolic theory

Changes in cerebral PP lead to accumulation or clearance of vasoactive metabolites in the perivascular space. Oxygen and metabolites produced during cellular function, e.g. carbon dioxide, adenosine and hydrogen, potassium and calcium ions, have vasoactive effects, predominantly dilatory. The metabolic theory of CA suggests that periods of local cerebral hypoperfusion result in the release and accumulation of vasoactive chemicals that produce vasodilatation of cerebral vessels. Hypoxia [Kontos 1978b] and hypercapnia are two of the most powerful stimuli increasing global CBF. Hypocapnia reduces CBF and widens the plateau region of the static pressure autoregulation curve, representing an improvement in autoregulatory capacity [Paulson 1990]. Hypercapnia has the opposite effect with an upward shift of the static autoregulatory curve and disappearance of the plateau at very high levels of CO₂ [Harper 1966]. More recently, hypercapnia has also been shown to reduce

[Zhang 1998a, Panerai 1999b, Birch 1995] and hypocapnia to enhance [Birch 1995] the efficiency of dynamic CA.

Again, a number of weaknesses exist in the metabolic theory. Adenosine has been shown to increase during periods of hypoperfusion, but caffeine, an adenosine antagonist, has not been shown to date to change CBF [Kontos 1985]. Hypotension per se has not been shown to alter potassium and hydrogen ion concentrations [Kuchinsky 1978]. Both hypoxia and hypercapnia would require the release of a mediating substance to affect CVR and this would probably take longer than the 8-20 second time period seen for the dynamic CA response. Therefore, the metabolic theory cannot be completely rejected, especially when considering static CA, but other mechanisms must be explored to satisfactorily explain the phoenomenon of dynamic CA.

1.3.2.3. Myogenic theory

The myogenic theory states that the smooth muscle of cerebral vessels is responsive to changes in transmural pressure with small arteries and arterioles constricting or dilating in response to increases or decreases respectively in transmural pressure. In other words, the ability to vasoconstrict with hypertension and vasodilate with hypotension is an intrinsic property of cerebral blood vessels. This theory was first proposed by Bayliss in 1902 following examination of the effects of altered PP on denervated isolated canine hindlimbs. A myogenic mechanism would be rapid and help explain the response of the cerebrovasculature to dynamic BP changes, but precisely how the mechanism works is unclear. Stretch of smooth muscle in vascular walls could alter membrane permeability to ions such as calcium, cause changes in membrane potential and ultimately modify the status of actin and myosin filaments in

the smooth muscle [Paulson 1990]. Evidence for this hypothesis exists in the form of an increase in frequency and amplitude of spontaneous action potentials in rat arteries in response to rises in MAP [Harder 1985]. In addition, Thorin-Trescases et al [1997] demonstrated an inverse relationship between the diameter of human pial arterioles and the level of tone at an intraluminal pressure range of 60-90mmHg, a finding that also supports the involvement of a myogenic process in CA.

Wagner et al [1985], however, suggested that CA was a function of the cerebral PP gradient and could not be fully explained by a myogenic mechanism.

1.3.2.4. Neurogenic theory

The neurogenic hypothesis states that the autoregulatory adjustment of the calibre of resistance vessels is mediated by vascular innervation. This theory assumes that there is a reflex arc by which changes in cerebral PP control the diameter of resistance vessels via efferent extrinsic vasomotor nerves. There is substantial histological evidence for a dense neurological innervation of extraparenchymal and intraparenchymal tissues [Rosenblum 1971, Harper 1975]. In general, larger arteries are densely innervated with plexuses diminishing in size as vessels become smaller. The extrinsic nerves originate mainly from the cranial ganglia and consist predominantly of adrenergic fibres, especially around the anterior cerebral, middle cerebral, internal carotid and posterior communicating arteries. Constrictor α receptors and dilatory β_1 receptors have been identified in these nerves [Heistad 1978a] and in vitro and in vivo studies have demonstrated vasoconstriction of cerebral vessels in response to direct and intravenous administration of noradrenaline and inhibition of this vasoconstriction by the α -adrenergic blocker phenoxybenzamine

[Rosenblum 1971]. Stimulation and denervation of α -adrenergic perivascular sympathetic nerves at the level of the superior cervical ganglion results in changes in cerebral blood volume, intracranial pressure and the formation of cerebrospinal fluid [Heistad 1978a]. These findings may, however, be due to changes mediated in the large extracranial vessels rather than those concerned with CA and few studies have imaged small vessels to directly document calibre changes. Carotid baroceptor stimulation does not cause a change in CBF [Heistad 1976] and sectioning of the superior cervical sympathetic trunk distal to the superior cervical ganglion does not alter basal CBF [Heistad 1978b].

A large number of animal and human experiments have examined the role of the sympathetic system in CA by assessing CA and cerebral reactivity in response to pharmacological and non-invasive manipulation of BP, exsanguination and CO₂ inhalation [Eklöf 1971, Skinhøj 1972, 1973, Heistad 1978a, Hernandez-Perez 1975, Harper 1975, Roatta 1998]. Overall, these studies have demonstrated no effect of the sympathetic nervous system on CBF within the normal range (60-140mmHg) of systemic MAP. There is some evidence, however, that activity of the sympathetic nervous system extends the autoregulatory plateau at both extremes and may thus play a role in protecting the brain from extremes of hypo- and hyper-perfusion [Harper 1975, Hernandez-Perez 1975, MacKenzie 1977, Sadoshima 1985, Talman 1994]. Others have found that pharmacological blockade of the sympathetic nervous system does not alter the upper BP level of the autoregulatory plateau [Skinhøj 1973]. More recently, Roatta et al [1998] used the cold pressor test to demonstrate that a moderate sympathetic stimulus may affect both large and small cerebral vessels well below the accepted upper limit of the autoregulatory plateau.

The role of the parasympathetic system, which originates from the sphenopalatine and otic ganglia, is unclear. Parasympathetic stimulation in sympathetically denervated rats does not cause an increase in CBF, but parasympathetic denervation may extend the autoregulatory plateau at lower BP levels in rodents [Morita 1994]. In conclusion, the parasympathetic system appears to have little role to play in the control of CA. The sympathetic system also seems to play a minor role at physiologically normal systemic BP levels, but may, however, extend the range of CA by its action on extraparenchymal vessels.

1.3.2.5. Endothelial derived factor theory

This hypothesis states that endothelial derived factors mediate CA through increases or decreases in endothelial derived relaxing factor (EDRF) and endothelial derived constriction factor (EDCF). EDRF has recently been identified as nitric oxide (NO) [Moncada 1988] and has been proposed as playing an important role in the regulation of CBF [Cohen 1995, Iadecola 1994a, 1994b, Brian 1996, Toda 1996, Bryan 1995, Kontos 1993, Okamoto 1997a]. The findings that inhibitors of NO synthetase reduce basal CBF, and that NO concentrations do not appear to be affected by changes in oxygen or glucose metabolism, imply a role for NO in the maintenance of basal tone [Gardiner 1990, Prado 1992, Faraci 1991, Iadecola 1994a, 1994b]. NO may also mediate cerebral vascular responses to acetylcholine, substance P, bradykinin and α_2 adrenergic receptor agonists [Brian 1996, Toda 1996, Bryan 1995]. There is also evidence that NO contributes to the rise in CBF produced by hypercapnia [Iadecola 1994a, 1994b, Toda 1996, Okamoto 1997a] and inhalational anaesthetics [Koenig 1994, McPherson 1993, Okamoto 1997b], though NO synthetase inhibitors do not seem to inhibit the hypercapnic hyperaemic response in humans [White 1998]. Moreover, several studies have indicated that inhibitors of NO synthetase attenuate the CBF changes in response to electrical stimulation [Akgoren 1996], seizures [Toda 1996] and nociceptive stimulation [Cholet 1996]. Neuronal NO may also be released directly by nerve endings into the vascular adventitia resulting in the relaxation of vascular smooth muscle in the cerebral circulation [Bredt 1990, Rodrigo 1997, Yoshida 1994]. The vasodilator actions of NO in the cerebral circulation are generally thought to be due to the stimulation of guanylate cyclase / Ca²⁺-activated K⁺ channels in vascular smooth muscle cells [Cohen 1995, Loscalzo 1995]. Activation of these channels hyperpolarises vascular smooth muscle and limits Ca²⁺ influx through voltage gated channels [Cohen 1995]. Recent studies have used inhibitors of guanylate cyclase to suggest that the vasodilator response to NO in pial arterioles in vivo and in vitro is mediated by cGMP-dependent and cGMP-independent pathways [Alonso-Galicia 1999, Faraci 1999, Cohen 1995, Weisbrod 1998].

The role of NO in CA is contraversial. Some investigators have suggested that NO synthetase inhibitors abolish CA [Kobari 1994a, Tanaka 1993] but this is disputed by others [Iadecola 1994a, 1994b].

At least two other important endothelial products have been identified, prostacyclin (a vasodilator) and endothelin-1 (a vasoconstrictor) [Cohen 1995]. How these endothelial products affect CBF and CA is unclear, but endogenous endothelin-1 may have a role to play in the control of the vascular tone of human pial arteries [Thorin 1998]. Moreover, intravascular infusions of low doses of endothelin-1 cause cerebral microvessel vasoconstriction in cats, but infusions of larger doses cause vasodilation, probably through the induction of NO in cerebrovascular endothelium [Kobari 1994b]. Inhibitors of cyclooxygenase, such as indomethacin, reduce basal CBF probably by interfering with the vasodilatory actions of prostacyclin [Markus 1994].

It is not clear which of these mechanisms if any is predominantly responsible for CA. It may be that neurogenic, endothelial and metabolic factors modulate an intrinsic vascular myogenic response by influencing the rapidity of the response, the slope of the pressure-flow relationship and range of the autoregulatory plateau. It is also possible that cerebral blood vessels autoregulate by different mechanisms according to their location and size. This is supported by the findings of Baumbach et al [1985] who, suggested that CA displayed regional and segmental heterogenecity as well as the temporal heterogenicity mentioned earlier. Regional differences in CA were confirmed by the finding that brainstem CA was more effective than cerebral CA in the face of increases and decreases of cerebral PP. The same investigators also suggested that segmental CVR indicated that small cerebral vessels (<200µm in diameter) made significant contributions to CA in the MAP range of 80-100mmHg while the role of large cerebral arteries (>200µm in diameter) became increasingly important at higher MAP levels.

1.3.3. Factors that may affect cerebral autoregulation

CBF and CA are thought to be influenced not only by pathological states, but also by a number of other physiological and non-physiological parameters and it is essential that these parameters are considered when studying CA.

1.3.3.1. Carbon dioxide partial pressure

Arterial pCO_2 and, to a lesser extent cerebrospinal fluid pCO_2 , have profound effects on CBF. As mentioned earlier, CO_2 has marked vasodilatory effects on the cerebrovasculature which add credence to the metabolic hypothesis of CA. Indeed,

inhalation of fixed concentrations of CO₂ (usually 5%) has been widely used as a method of measuring cerebral vasodilatory reserve capacity, also referred to as cerebral reactivity (CR) [Paulson 1972, Markwalder 1984, Ameriso 1994]. CR cannot be compared with CA, the response to changes in MAP, and, indeed, CR and CA may be affected differently by various processes - so called "dissociated vasoparalysis". Studies of CR have, however, shown that CBF increases with rises in pCO_2 and that there may be an exponential relationship between the two [Paulson 1972, Markwalder 1984, Ameriso 1994]. Previous animal work has suggested that the CBF response may be blunted at extremes of pCO₂ [Harper 1965]. In addition to effects on CBF, changes in pCO_2 have also been shown to profoundly affect CA. Hypocapnia reduces CBF and widens the plateau region of the static pressure autoregulation curve, representing an improvement in autoregulatory capacity (Figure 1.1.) [Paulson 1990]. Hypercapnia has the opposite effect with an upward shift of the static autoregulatory curve and disappearance of the plateau at very high levels of CO₂ (Figure 1.1.) [Harper 1966]. Aaslid et al confirmed the impairment of CA during hypercapnia and augmentation during hypocapnia with the use of the thigh-cuff technique in 1989. More recently, hypercapnia has also been shown to reduce [Zhang 1998a, Panerai 1999b, Birch 1995] and hypocapnia to enhance [Birch 1995] the efficiency of dynamic CA.

As pCO_2 has such profound effects on CA, we believe that CO_2 assessment is vital in any study of CA.

1.3.3.2. Age

The cardiovascular system demonstrates many age-related changes as evidenced by the well-recognized increase in systolic blood pressure (SBP) [Kannel 1978] and decreases in systemic artery compliance, endothelial NO release and cardiac BRS with age [Fleg 1986, Gribbin 1971, Dawson 1999b, Tschudi 1996]. Aging, through the use of the nitrous oxide and Xenon clearance methods, has also been shown to be associated with a demise in CBF and CBF volume [Kety 1956, Naritomi 1979, Shaw 1984, Matsuda 1984]. Transcranial Doppler ultrasound has appeared to confirm these findings by demonstrating reduced cerebral blood flow velocities with advancing years [Martin 1994, Krejza 1999]. In addition, it has been suggested that CR also decreases with age and that coincident vascular risk factors augment the decrease [Yamamoto 1980, Yamaguchi 1979]. Davis et al [1983] contradicted these findings, however, by finding no age-related differences in CR assessed by CO₂ inhalation, although, contrary to others, they also claimed no reduction in cerebral white matter blood flow. A recent study used the breath-holding index to show an age-related decrease in CR between pre- and post-menopausal women [Matteis 1998]. No difference was shown between young and older men, however, and the findings in women may, therefore, reflect the influence of hormonal factors rather than age per se (see next section).

Senescent rats demonstrate a pressure passive relationship between CBF and BP, suggesting impairment of CA [Hoffman 1981], and this is accentuated in spontaneously hypertensive breeds, implying that age-related BP changes may be more important than age per se. The effects of age on CA in humans, however, are unknown and the hypothesis that the integrity of CA diminishes with age forms the basis of one of our studies. Nevertheless, in view of the known effects of age on CBF and CR, age should be a controlled factor in studies of CA until further evidence of the effect of age on CA is available.

1.3.3.3. Gender

Women have been shown to have higher levels of CBF until the menopause, a finding potentially attributable to female hormones, such as oestrogens or progestogens, or to the lower haematocrit, a factor compounded by menstrual blood loss [Kety 1956, Martin 1994, Karnik 1996]. Two studies assessing CR, one with CO₂ inhalation and the other with acetazolamide injection, demonstrated increased vasodilatory responses in pre-menopausal women [Karnik 1996, Kastrup 1998]. The effects of gender on CA, however, are unknown and, in view of the known effects on CBF and CR, gender should be a controlled factor in studies of CA.

1.3.3.4. Blood pressure

There is considerable experimental evidence that vascular histology changes as systemic BP levels increase. Smooth muscle hypertrophy and thickening of the extracellular matrix, through alterations in laminin, fibronectin and collagen IV, lead to an increase in the media to lumen ratio and changes in vascular resistance through reduced vascular stress (Laplace's Law) [Harper 1984, Faraci 1990, Slivka 1991, Nag 1997]. The effects of these changes on CBF are, however, unclear. No differences in CBF were demonstrated between spontaneously hypertensive rats and normotensive Wistar Kyoto rats [Hoffman 1981]. CBF was diminished, however, particularly in frontal cortical areas, in stroke-prone spontaneously hypertensive rats when compared to normotensive Wistar Kyoto rats and stroke-resistant spontaneously hypertensive rats [Yamori 1977]. These latter findings may suggest an influence of genotype or vascular disease on CBF rather than an influence of hypertension per se. Human studies have been equally inconclusive. A number of investigators have found no change in CBF in hypertensive patients [Kety 1947, Faraci 1990, Thulin 1993]. The study of Kety et al [1947] also demonstrated that the oxygen extraction fraction was unchanged, but that cerebrovascular resistance increased with BP. Conversely, others have implied that CBF is reduced in hypertensive patients after finding reduced oxygen consumption with the use of an arteriovenous oxygen extraction technique [Strandgaard 1973]. Untreated hypertensive patients were shown to have lower CBF when compared with age and sex matched well-controlled hypertensives and normotensives [Nobili 1993]. This finding was confirmed by PET scanning and internal jugular monitoring [Fujishima 1995, Lambert 1996] methods which also uncovered regional differences in CBF, consisting of selectively larger decreases in CBF in the frontal cortex and basal ganglia.

The effect of hypertension on cerebral reactivity (CR) is also disputed. Tominaga et al [1976] found no alteration in the cerebrovascular response to inhaled CO_2 or hyperventilation in the pCO₂ range of 20-55mmHg, while more recent TCD studies have suggested that CR is impaired in hypertension [Maeda 1994].

CA has, however, been shown to be affected by hypertension, predominantly in the form of a shift in the static autoregulatory curve to the right (Figure 1.1.) [Strandgaard 1973, 1976, 1989, Hoffman 1981, Fujishima 1984]. There is evidence that this shift increases with longer durations of hypertension and that effective treatment may reverse the trend [Strandgaard 1976, 1989, Harper 1984, Slivka 1991, Nag 1997]. The shift in the autoregulatory curve to the right may have a protective function against severe hypertension, but it is unclear if the changes become permanent when present for a length of time or how soon after the instigation of treatment that reversal of the changes may be seen [Strandgaard 1976].

Despite a clear demonstration of a shift in the static autoregulatory curve to the right, it is unclear if CA is actually impaired by the shift. Faraci et al [1987] demonstrated

increased resistance in large and small cerebral vessels in cats, but only small vessels showed an increase in resistance in the brainstem. Severe hypertension, however, caused a fall in resistance in the cerebrum, suggesting impaired CA, but maintenance of high resistance in the brainstem vessels, again indicating possible regional differences in the effect of BP on CA.

A recent study in humans suggested that dynamic CA is preserved in untreated hypertensive patients [Blake 1999], but data concerning the effect of BP on dynamic CA is otherwise scarce.

In view of the uncertainty of the effects of BP on CA, BP should be a controlled factor in studies of CA.

1.3.3.5. Drug therapy

May different classes of drug affect CBF, CA and CR and much work has been done assessing the effects of antihypertensive agents.

Diuretics lower systemic BP without affecting CBF [Venkata 1987, Semplicini 1993, Landmark 1995], but do not appear to improve impaired CR in hypertensive subjects [Traub 1982]. Calcium antagonists also lower systemic BP without affecting CBF [Thulin 1993, Landmark 1995, Pandita-Gunawardena 1999] and, in addition, appear to correct the autoregulatory shift induced by hypertension (see section 1.3.3.4) [Fujishima 1995]. β adrenergic receptor blockers may increase resistance in the small cerebral vessels responsible for CA and lower the vasodilatory capacity to CO₂ challenge and hypotensive stress [Mathew 1973, Fujishima 1995]. Longer term administration at higher doses, however, may increase CBF through a vasodilatory action on the cerebral circulation [Globus 1983]. α adrenergic receptor blockers reduce systemic BP with debate over whether CBF is subsequently increased [Mathew 1973, Fujishima 1995] or unchanged [Venkata 1987]. Angiotensin converting enzyme (ACE) inhibitors reduce systemic BP in humans without affecting CBF [Waldemar 1989, 1990, Semplicini 1993, Démolis 1993] and may correct the shift of the autoregulatory curve to the right [Waldemar 1989] resulting in an improvement in vasodilatory reserve to acetazolamide [Démolis 1993]. In addition, inhibitors of ACE may reduce the upper limit of static CA leading to a shortening of the autoregulatory plateau [Squire 1994].

Finally, the general anaesthetic agent propofol does not affect CA but isoflurane and desflurane may alter dynamic and then static CA [Strebel 1995].

1.3.3.6. Posture

Posture has been shown to have an effect on CBF. Transcranial Doppler ultrasonography has consistently shown reduced CBFV after HUT [Schondorf 1997, Stoll 1999, Cencetti 1997, Grubb 1991a], during lower body negative pressure [Bondar 1995,1994, Levine 1994, Giller 1992] and during active standing [Savin 1995] and higher CBFV during head-down tilt [Savin 1995]. Changes in CBFV may be due to a variety of different reasons, including altered systemic sympathetic activation [Bondar 1994, Levine 1994] or changes in the diameter of the MCA, although recent work has suggested that MCA calibre does not change during simulated orthostasis [Serrador 2000]. Furthermore, a direct relationship has been shown between the physiologically lower CO₂ levels after HUT and reduced CBFV [Cencetti 1997]. In spite of the CBF changes demonstrated on posture, little information exists on the

effect of posture on CA. Using frequency domain analysis and lower body negative pressure, Zhang et al [1998b] suggested that CA may deteriorate during orthostatic

stress. Conclusions were based, however, on the transfer-function gain between pressure and velocity, the interpretation of which is a matter of some debate. In addition, coherence and phase results from the same work do not support worsening of CA with orthostatic stress. Methodology using rapid thigh cuff deflation has failed to demonstrate any deterioration in dynamic CA with 40° HUT [Leftheriotis 1998] but this angle of tilt is significantly lower than angles currently used for HUT testing [Kenny 2000]. More recently, infared spectroscopy has been used to demonstrate a significant decrease in cerebral oxygenation during active standing in healthy elderly subjects with no change in the cerebral oxygenation of younger subjects [Mehagnoul-Schipper 2000]. This could result from inferior CA in elderly subjects on standing, but cerebral oxygenation is not a surrogate for CA and only frontal oxygenation was measured in the study.

The precise effect, therefore, of changing posture on CA is unclear and requires further investigation.

1.3.3.7. Miscellaneous factors

Tobacco smokers have significantly lower CBF [Shaw 1984], a 48% reduction in vasodilatory capacity to CO₂, and a 24% reduction in vasoconstriction to 100% oxygen compared with healthy controls [Rogers 1984]. The act of smoking, however, increases CBF [Boyajian 2000] through mechanisms that are unclear. Xenon inhalation studies have demonstrated reduced CBF in subjects with heavy alcohol consumption [Shaw 1984, Rogers 1983]. Patients with transient ischaemic attacks and hyperlipidaemia have been shown to have significantly lower CBF than similar patients with normal lipid profiles, but
normal subjects with hyperlipidaemia appear to have no significant trend towards reduced CBF [Meyer 1987].

Whole blood viscosity is mainly determined by haematocrit, and CBF has been shown to be inversely proportional to both haematocrit and whole blood viscosity, as well as fibrinogen levels [Thomas 1982, Nelson 1956, Ameriso 1990]. It is thought that the inverse relationship of CBF with haematocrit is not due to sluggish blood flow at higher viscosities, but rather that the higher oxygen content of high-haematocrit blood allows the maintenance of oxygen delivery in the face of lower CBF.

Although the above factors have been shown to affect CBF, their influence on CA is unknown. In view of these uncertainties, it is sensible to control for as many of these factors as possible when studying CA. Figure 1.1. Autoregulatory curves of a normal subject at rest (black) and during hypercapnia (solid blue) and hypocapnia (dotted blue) and of a normocapnic hypertensive subject (red).



52

Reflex mediated	-	Vasovagal				
	-	Situational	-	ne Deglu	tition	
		21100010101	-	Cough	1	
			-	Mictu	rition	
			-	Defec	ation	
	-	Neuralgia				
Orthostatic hypotension						
Cerebral						
Psychogenic						
Neurological		Migraine				
-	-	Transient isch	nsient ischaemic attacks			
	-	Seizures				
	-	- Subclavian steal syndrome				
Cardiac -	Organic heart disease -		; -	Aortic stenosis		
			-	Hyper	trophic cardiomyopathy	
			-	Pulmo	onary embolism	
			-	Myxo	ma andial information	
			-	Coron	ardial infarction	
			-	Tamp	onade	
			-	Aortic	dissection	
-	- Arrhythmias					
		Tachyarrhyth	mias	-	Ventricular tachycardia	
				-	Torsades de pointes	
				-	Supraventricular	
		Bradyarrhyth	/thmias	-	Sinus node disease	
				-	$2^{nd}/3^{rd}$ degree heart block	
				-	Pacemaker malfunction	

Table 1.1. Causes of Syncope

Iatrogenic

Idiopathic

Table 1.2. Investigations for Syncope

.

<u>Investigation</u>	Diagnostic Yield
Electrocardiography	5%
"Routine blood tests"	2-3%
24-hour Holter monitoring	19%
Echocardiography	5-10%
Tilt-table test & Carotid sinus massage	26-90%
Loop recorder	24-47%
Electroencephalography	1-2%
Electrophysiological studies	10-50%
Computed tomography of the brain	4%
Magnetic resonance imaging of the brain	4%
Exercise stress test	<1%
Thallium stress test	<1%
Psychiatric evaluation	5-24%
Carotid Doppler ultrasonography	6%
Transcranial Doppler ultrasonography	Unknown

1

Table 1.3. Classification of Vasovagal Syncope

Type 1 (mixed)

Systolic BP <80 mmHg and heart rate >40bpm

<u>or</u>

Systolic BP <80mmHg and heart rate <40 bpm for <10 seconds

<u>or</u>

Systolic BP <80mmHg and asystole for <3 seconds

Type 2 (cardioinhibitory)

Heart rate ≤ 40 beats per minute for >10 seconds <u>or</u> asystole for >3 seconds

- 2A BP falls prior to fall in heart rate
- 2B BP falls at or after onset of fall in heart rate

Type 3 (vasodepressor)

Systolic BP <80mmHg and heart rate >90% of its peak value at syncope

CHAPTER TWO

Methods

2.1. Measurement of Cerebral Blood Flow

Cerebral blood flow (CBF), as discussed in section 1.2, is primarily dependent on perfusion pressure (PP) and cerebrovascular resistance (CVR), but recent developments have suggested that the critical closing pressure (CrCP) of the cerebral circulation, i.e. the pressure below which blood flow in a vessel ceases, may have an important role to play [Dewey 1974, Dawson 1999a)]. CBF was first measured in animals using invasive methods such as bubble flow meters, but less invasive techniques had to be developed for use in human subjects. In this section, the most widely used techniques of measuring CBF in humans will be discussed, some of which are now largely of historical interest only, whilst concentrating on the currently most widely used technique, transcranial Doppler ultrasonography (TCD).

2.1.1. Nitrous Oxide Method

Initially described by Kety et al in 1948, this method employed the Fick principle and was the first method that could be applied to unanaesthetised subjects. The Fick principle states that the "quantity of any substance taken up in a given time by an organ from the blood that supplies it is equal to the amount of the substance carried to the organ by the arterial inflow minus the amount removed by the venous drainage in the same period", or:

 $(Q_b)_u = (Q_a)_u - (Q_v)_u$

where $(Q_b)_u$ is the quantity of substance taken up by whole brain, $(Q_a)_u$ is the quantity contained in the arterial inflow and $(Q_v)_u$ is the quantity drained by the venous system. Nitrous oxide (N₂O), an inert gas, became the substance of choice for use with this technique because of its favourable solubility and diffusion characteristics. A low concentration was inhaled and CBF calculated using arterial and venous blood samples acquired when equilibrium had been established, approximately 10 minutes after inhalation, according to the equations:

$$(Q_a)_u = TF \int_0^u Adt$$
$$(Q_v)_u = TF \int_0^u Vdt$$
$$\therefore TF = \frac{(Q_b)_u}{\int_0^u (A - V)dt}$$

where A is the arterial concentration of N_2O , V the venous concentration and TF the total cerebral blood flow per minute. This technique was the first to approximate normal blood flow to be in the region of 0.5L/kg/minute, but had three inherent drawbacks. No compensation could be made for any inequalities that might exist in venous drainage by the two internal jugular bulbs, contamination of the venous drainage by extracerebral structures was an established risk and temporal resolution was extremely poor.

2.1.2. Xenon Inhalation / Injection Method

Devised in the 1960's, this method initially employed intracarotid bolus injections of radioactive ¹³³Xenon (Xe), but was later refined to utilise either inhaled or intravenously injected ¹³³Xe [Mallet 1965, Obrist 1975, Overgaard 1981, Bouma 1990]. Clearance curves of the γ radiation were measured by extracranial detectors, usually 8 in number, attached to the surface of the cranium. The intracarotid technique fell out of favour, not just because of its invasive nature, but also because it only allowed measurement of CBF in one anterior henisphere, unlike the inhalation and intravenous techniques which allowed examination of bilateral anterior and posterior circulations. All techniques employing ¹³³Xe were, however, limited by

poor temporal resolution and two other main limitations. Detectors only picked up changes occurring in the superficial brain structures and, therefore, yielded little in the way of anatomical information. In addition, hyperventilation often triggered by the process necessitated adjustment of CBF values for CO₂ levels.

2.1.3. Xenon enhanced computer tomography

This technique proved initially very attractive as it combined the use of computerised tomography to increase anatomical detail, especially of deep seated structures, with the use of non-radioactive Xe gas [Gur 1982]. Unlike earlier studies that assessed elimination, this technique assessed Xe uptake in different cerebral areas approximately 4 minutes after inhalation when equilibrium was thought to exist between end-tidal and arterial Xe levels. The high atomic number and high permeability across the blood-brain barrier the element allowed enhancement on CT scans at quite low inhaled concentrations. Calculation of CBF with this method, however, assumed uniform uptake throughout the brain substance but fast and slow flow areas led to inevitable error. While some mathematical compensation could be made for variations in uptake, substantial errors around pathological regions, e.g. between an infarct and its penumbra, severely limited its usefulness. Raised intracranial pressure was a recognised side effect of Xe and, once again, adjustment of CBF calculations for hyperventilation induced CO₂ changes was necessary. Temporal resolution was poor, and, in addition, difficulties with prototype CT scanners, such as high radiation dose, movement artefact and poor image resolution, compromised the safety and usefulness of the procedure.

2.1.4. Other radiological techniques

Positron emission tomography (PET), single photon enhanced computerised tomography (SPECT) and, more recently, phase contrast magnetic resonance angiography (PC MRA) [Patrick 1996] have emerged as exciting tools as they give information on cerebral metabolism as well as CBF with excellent morphological and anatomical detail. Expense and poor temporal resolution have, however, precluded their widespread use to date in the study of CBF and autoregulation.

2.1.5. Transcranial Doppler Ultrasonography

2.1.5.1. Physical and technical aspects

Christian Doppler first described the physical phoenomenon that has been named the Doppler principle when he noted that light waves were reflected at different frequencies when disturbed by a moving object. Satomura and Karubo first used the principle to study blood flow in 1960 and by 1965 Kato was using a transcranial Doppler (TCD) ultrasound technique to study CBF during neurosurgical procedures. Ultrasound refers to sound frequencies higher than 20 KHz. Current TCD techniques use a transducer (i.e. probe) with a peizoelectric crystal that generates ultrasound waves of a known frequency which are focused into a beam by a lens. When the beam is directed at a stream of moving red blood cells, the ultrasound waves are scattered and reflected back with a changed frequency to the transducer that also acts as a receiver. The change in frequency is known as the Doppler shift and is calculated according to the formula:

$\Delta f = \frac{2(v\cos\theta)f_0}{c}$

where Δf is the Doppler shift, v is the magnitude of scatter velocity, θ is the angle between the ultrasound beam and the direction of motion, f_0 is the original frequency of the transmitted ultrasound beam and c is speed of propagation of ultrasound in soft tissues.

Because the sample volume contains a large number of cells causing different amounts of scatter, a spectrum of frequencies is relayed to the transducer. The maximum Δf is achieved by the highest cell velocity in the given time sample and the mean Δf reflects the mean velocity in the time sample [Gill 1985]. Accuracy is improved by the use of a Doppler beam that is uniform over the entire area of the insonated vessel [Evans 1982], and spatial resolution increased by the use of pulsed wave, rather than continuous wave, Doppler. The interface between the probe and the tissue being insonated is provided by a water-based coupling gel. Initial use of the technique for CBF measurement was hindered by the fact that most Doppler probes emitted a frequency that was absorbed or scattered by the cranial bones to such an extent that blood flow detection was impossible. However, 1982 saw the introduction of the PEDOF 2-MHz pulsed Doppler device which overcame these problems [Aaslid 1982] to allow measurement of blood flow velocity in large basal intracranial arteries. In addition, Aaslid appreciated the non-uniformity of the cranial bones and described the acoustic windows over the temporal, occipital and orbital bones. The introduction of a gated probe further improved the overall technique by allowing alteration of the depth of insonation and determination of the direction of blood flow either towards or away from the probe [Newell 1992].

Insonation may, however, prove technically impossible if an adequate acoustic window is not present, a finding commoner in older women and people of certain ethnic racial origins including the Chinese and Afro-Caribbean populations [Halsey 1990]. Detection rates in such subjects may be enhanced by increasing the probe power from the usual 100mW/cm^2 to as much as 800mW/cm^2 , but scalp discomfort and even burns may result from the use of such high powers [Halsey 1990].

2.1.5.2. Vessel identification and reproducibility

Aaslid [1982] described in detail the methods of insonation of the basal cerebral arteries comprising the Circle of Willis (Figure 2.1.) through the various acoustic windows (Figure 2.2). In doing so, he employed a 2-MHz probe with a high pass filter of 100 MHz, a low pass filter of 3.4-9 KHz, an emitted power of 350mW, a burst repetition of 6.8-18 KHz, a pulse length of 10µs and an emitting area of 1.5cm², i.e. 10 times the cross sectional area of the middle cerebral artery. For the studies in this thesis, the proximal segment of the middle cerebral artery

(MCA) was insonated, bilaterally when possible, through the transtemporal acoustic window using a SciMed QVL 120 TCD apparatus (Figure 2.3.). This is a purpose built, dual-channel machine as the use of two separate machines is practically impossible due to the asynchronicity of pulse repetition and frequency of both machines.

The transtemporal acoustic window lies immediately superior to the zygomatic arch, is comprised of a natural thinning of the temporal bone due to attenuation of the temporal squamosa, and is divided into three main functional areas - anterior, middle and posterior (Figure 2.2.). The posterior window is often the most successful point for MCA insonation and requires anterior and superior direction of the beam. Although as much as 39% of an ultrasound beam may be absorbed at this point, satisfactory insonation can be achieved in over 75% of subjects. In addition to the MCA, the terminal internal carotid artery (TICA), anterior and posterior cerebral arteries (ACA and PCA respectively) and the anterior and posterior communicating

arteries (ACoA and PCoA respectively) may be identified through the transcranial window. Accurate identification of vessels can be augmented by knowledge of the anatomy (Figure 2.1.), direction of flow, sampling depth and likely velocities [Aaslid 1982, Arnolds 1986, Martin 1994] of each vessel as well as the response of flow in each vessel to ipsilateral common carotid artery compression (Table 2.1.). Using such methods, TCD has proven accurate in vessel identification and, if interhemispheric asymmetry is kept to a minimum, both inter- and intra-observer reproducibility is very good. Inter- and intra-observer reproducibility at 2 to 24 hours is <8% [Padayachee 1986, Totaro 1992, Demolis 1993, Baumgartneer 1994, Bay-Hansen 1997] and intra-observer reproducibility 13% at 2 months [Bay-Hansen 1997].

2.1.5.3. Validation as a method of measuring cerebral blood flow

CBF can be calculated from velocity measurements using the equation:

$$CBF = \sum_{i} v_i . \Delta A_i$$

where ΔA_i is the cross sectional area of the insonated vessel and v_i is the velocity of blood at the point cross section measured.

This calculation, however, assumes that there is no significant change in the diameter of the artery being insonated, that the angle of insonation is less than 30° (to reduce error in the calculation of velocity to <15%) and that the territory supplied by the artery in question is also constant. The angle of insonation is important as it can lead to significant underestimations of velocity a fact apparent in the equations:

$$\cos\theta^1 = \frac{\cos\theta}{\cos\eta}$$

$$V_t = \frac{V_{obs}}{\cos\theta}$$

where θ^1 is the measured angle, θ is the true angle, η is the angle of the vessel crossing the plane of imaging, V_t is the true velocity and V_{obs} is the observed velocity. The angle of insonation is generally assumed to be zero in most studies using TCD as it is impossible to measure it.

Changes in blood pressure and CO₂ and O₂ concentrations do not significantly alter the diameter of the MCA and other large basal cerebral arteries (<4%) but small distal vessels actively concerned with CA undergo considerable variation (>29%) [Hubner 1967, Giller 1993, Poulin 1996, Serrador 2000]. Attempts have been made to use Doppler spectral power to assess MCA vessel diameter during thigh cuff application and release [Aaslid 1989] and during hypocapnia and hypercapnia [Poulin 1996], but a recent study has cast doubt on the usefulness of this method of assessing vessel diameter [Deverson 2000]. A significant change in MCA diameter is unlikely, however, during thigh cuff inflation and release [Newell 1994b], lower body negative pressure [Serrador 2000], Valsalva manoeuver [Giller 1993, Tiecks 1996] or supine rest.

Because of the multiple assumptions, calculations of CBF from TCD measurements do not correlate well with values derived using the traditional methods of CBF measurement e.g. correlation between intravenous Xe clearance studies and TCD is 0.424 [Bishop 1986]. The correlation between CBF and velocities is much better, however, being approximately 0.63 between SPECT and TCD [Lindegaard 1987, Dahl 1992, Newell 1994b]. Even better correlation scores have resulted from comparisons between TCD velocities and CBF measurements during manoeuvres e.g. correlation between TCD velocities and Xe clearance is as high as 0.85 when testing cerebrovascular reactivity [Bishop 1986, Kontos 1989, Dahl 1992, Larsen 1994, Sugimori 1995]. Therefore, although TCD does not lend itself well to the absolute calculation of CBF, velocity and changes in velocity correlate well with CBF results obtained using traditional techniques. Consequently, TCD-derived velocities from the main basal cerebral arteries (usually the MCA) have been widely used as a surrogate for CBF and to reflect changes in hemispheric CBF.

TCD machines can now display continuous beat-to-beat velocity profiles of basal cerebral arteries by performing power spectral analysis of the received signal using fast Fourier transformation (FFT) [Attinger 1966]. The information is displayed as an X-Y representation, the X-axis representing time and the Y-axis representing velocity in cm/s, with the intensity of the reflected signal represented by the brightness of the displayed velocity (Figure 2.4.). Conversion by spectral analysis yields an analogue output similar to the output from a non-invasive BP monitor, thus enabling measurement of beat-to-beat changes and the assessment of dynamic changes with a remarkable degree of temporal resolution [Aaslid 1982].

2.1.5.4. Calculation of indices of cerebrovascular resistance & critical closing pressure using transcranial Doppler ultrasound measured cerebral blood flow velocity

As discussed in section 1.2, cerebrovascular resistance (CVR), is one of the primary determinants of CBF and transcranial Doppler ultrasound has been used to derive various indices of CVR.

Gosling's pulsatility index (GPI) [1974] and Pourcelot's Resistance Index (PRI) [Planiol 1974] were among the first indices described:

$$GPI = (V_{\rm s} - V_{\rm d}) / V_{\rm m}$$

$$PRI = (V_{\rm s} - V_{\rm d}) / V_{\rm s}$$

where V_s , V_d and V_m represent systolic, diastolic and mean CBFV respectively.

The accuracy and tendency of these indices are practically identical in most applications, but GPI has probably been more widely used [Gomez 1999, Njemanze 1993, Daffertshafer 1995, Grubb 1998, Fredman 1995]. The attractiveness of these indices is that they may be derived from TCD recordings alone and do not require simultaneous measurement of arterial BP. The main drawback, however, is that both indices are small in magnitude and easily overshadowed by central cardiovascular factors [Aaslid 1992]. In addition, both indices were derived to reflect the interplay between the resistance and compliance components of the cerebrovascular bed, and neither index, therefore, directly reflects any physiological property such as CBFV, CVR or vasomotor tone [Aaslid 1992]. Moreover, a recent animal study has suggested that there may, in fact, be an inverse relationship between GPI and CVR [Czosnyka 1996a].

Classical CVR (CCVR) may be a more accurate index of CVR as it is calculated using the Poiseuille equation which takes arterial BP into account:

CCVR = Mean ABP / Mean CBFV

CCVR, unlike GPI and PRI, is therefore less likely to be overshadowed by central cardiovascular factors. Calculation of CCVR, however, makes the assumption that CBFV always intersects the ABP axis at 0mmHg (i.e. that blood flow continues even at very low levels of ABP), ignoring the critical closing pressure (CrCP) of the cerebral circulation (Figure 2.5.).

CrCP refers to the pressure below which blood flow in a vessel ceases and may have a profound effect on CBF [Burton 1951, Aaslid 1992, Dewey 1974, Dawson 1999a, Panerai 1995, 1996, 1999b, Garnham 1999, Carey 2000]. CrCP has an inverse relationship with CO₂ levels [Aaslid 1992, Panerai 1999b. Garnham 1999] and can change over a matter of seconds [Carey 2000]. Perhaps the most well described

method of calculating CrCP involves plotting the instantaneous relationship between CBFV and ABP for each cardiac cycle and performing linear regression to the ABP axis [Aaslid 1992, Panerai 1999b, 1995, Carey 2000, Garnham 1999, Dawson 1999a] (Figure 2.5., line 2). In addition, this method has been validated for very low levels of ABP [Carey 2000]. The point of intercept of the arterial BP axis represents the CrCP while the inverse of slope of the line represents the resistance-area product (RAP). RAP, therefore is a similar resistance index to CCVR, but probably superior as it allows for changes in CrCP. RAP may, therefore, be the best index of CVR as it allows for changes in both CrCP and ABP and it truly reflects the beat-to-beat changes in the instantaneous relationship between CBFV and arterial BP. Examination of Figure 2.5. should make it clear that changes in cerebral perfusion may occur through changes in CrCP without changes in RAP (line 3) and vice versa (line 1).

In conclusion, TCD is a relatively simple, well-tolerated, non-invasive technique for the assessment of CBF using basal cerebral artery blood flow velocities as a surrogate for CBF. It has been validated against traditional techniques, has a high degree of reproducibility and excellent temporal resolution, enabling assessment of both dynamic and static changes.

2.2. Non-invasive continuous blood pressure monitoring

Assessment of cerebral autoregulation (CA) requires measurement of the beat-to-beat response in CBF/MCAV to beat-to-beat changes in arterial BP. Intra-arterial monitoring remains the "gold standard" for continuous measurement of BP, but this is invasive, painful and entails inherent significant risks of thromboembolism,

haemorrhage, local and systemic infection and arterial dissection [Mangato 1979]. Servo-controlled plethysmomanometry permits non-invasive, continuous beat-to-beat recording of peripheral arterial BP and is based on the principles of "vascular unloading" and the "arterial volume clamp" first reported by Penaz [1973]. The development of fully automated instruments was a long awaited step forward in the technique of BP measurement. A number of commercially available, non-invasive, continuous BP monitors currently exist, including the Finapres 2300 system (Ohmeda, Colarado, USA) (Figure 2.6.).

The Finapres employs an infared finger photoplethysmograph surrounded by a fluid filled pressure cuff. An appropriately sized cuff is placed around each subject's finger, usually the middle finger of the non-dominant hand, and automatically inflated until the maximal plethysmographic finger pulsation is detected (Figure 2.6.). A finger volume is then calculated and the cuff aims to keep this value constant by a negative feedback loop (servo-adjust mechanism). Any adjustment reflects a change in arterial BP which is duly recalculated on a beat-to-beat basis. Results can be downloaded onto a computer, printed out or recorded directly onto a digital analogue tape (Figure 2.6.).

A number of studies have validated this system against intra-arterial brachial measurements in a variety of situatuions. Imholz et al [1988] found that the Finapres device reproduced intra-arterial BP patterns at rest faithfully, with values underestimated by 1 ± 10 mmHg, 9 ± 7 mmHg and 4 ± 6 mmHg for SBP, DBP and MAP respectively. Others have also examined the reliability of the Finapres in a variety of invasive and non-invasive physiological settings [Parati 1989, Lal 1995, Rongen 1995]. Manoeuvres used included the Valsalva manouevre, isometric hand grip, cold pressor test, mental arithmetic, passive leg tilt, lower body negative

pressure and passive and active standing, some of which have been employed during this research project. The Finapres tended to underestimate readings, particularly for SBP, in those studies but detected the same trends of change with accurate temporal representation. McAuley [1997], however, demonstrated that the Finapres SBP level was consistently and significantly higher than the intra-brachial value and that the difference increased during the cold pressor test. Omboni [1993] demonstrated that standard deviations of DBP, MAP and pulse interval were similar when assessed by Finapres and intra-arterial monitoring, whereas standard deviation of SBP was significantly overestimated by the Finapres. Bos [1992] found that Finapres measurements in patients with peripheral vascular disease were not always equal to intra-brachial measurements and that, during the Valsalva manoeuvre, the magnitude of BP response was sometimes over- or under-estimated by the Finapres device. Of particular interest, Jellema [1996] demonstrated that the Finapres device could be reliably used in clinical settings as a monitor of sudden changes in BP such as those induced by head-up tilt. Although arterial pressure waveforms change with head-up tilt, Petersen [1995] demonstrated that non-invasive plethysmography correlates very well with intra-arterial pressure recordings during head-up tilt. Most studies validating the Finapres device have been performed in subjects younger than 75 years, but validation has also been performed in the 71-83 years age group with comparable results [Rongen 1995].

The accuracy of the Finapres may be influenced by a number of factors. Jones [1993] demonstrated that SBP levels were underestimated when the finger cuff was applied too tightly and overestimated when applied too loosely. Finapres SBP may be significantly affected by fingertip temperature, probably as a result of local vasoconstriction of arteriovenous shunts, and finger warming may be a useful

procedure to improve the reliability of Finapres readings [Tanaka H 1993]. Hildebrandt [1991], however, found that heating of the fingertip distal to the Finapres cuff significantly decreased BP readings and concluded that heat-induced vasodilation may make Finapres readings unrepresentative of systemic BP. Other possible explanations for BP over- and under-estimation in the above studies include finger cuff malapplication, inappropriate cuff size, finger vasospasm and failure to position the hand at atrial level. In our subjects, an appropriate cuff size was chosen for each subject in agreement with the manufacturer's instructions and the hand and arm were supported at atrial level at all times using a custom designed arm-rest. All recordings were performed in a temperature controlled laboratory and BP readings were only accepted if they were within 15/10 mmHg of contralateral brachial readings.

In conclusion, the Finapres is currently the device of choice for non-invasive, continuous beat-to-beat BP monitoring, especially when measurement of relative changes, rather than absolute values, of BP is desired. It has been validated against intra-arterial measurements in older age groups and during a variety of manoeuvres, including head-up tilt.

2.3. Measurement of carbon dioxide partial pressure

As discussed previously, CO_2 has profound effects on CBF, through its stimulation of vasodilation, its depressive effect on critical closing pressure, and upward effect on the static autoregulatory curve. In any studies of CBF or CA, therefore, monitoring of CO_2 is essential. Arterial blood gas sampling is regarded as the "gold standard" of carbon dioxide partial pressure (p CO_2) measurement. However, this yields results for a single point in time only, and, like intra-arterial BP measurement, is not without

complications if repeatedly performed [Downs 1973, Bedford 1977]. Two recognised alternatives to arterial blood gas sampling are transcutaneous and end-tidal CO_2 measurement.

2.3.1. End-tidal carbon dioxide (ETCO₂) measurement

Capnography is a non-invasive technique allowing continuous measurement of expired or end-tidal CO₂ levels with the aid of infared spectroscopy [Ledingham]. Capnography uses microprocessor technology and the principle of infared absorption to measure ETCO₂ levels. Infared energy is a continuous band of electromagnetic radiation which borders the lower frequencies of visible light. CO₂ absorbs infared energy in direct proportion to the concentration of CO₂ present, and the amount of absorption may be measured by a photometer. Exhaled gas is channelled to the photometer via nasal canulae or, as in the case of our subjects, via a closely fitting face mask. A continuous beam of infared energy emitted by the photometer passes through the gas sample which absorbs frequencies specific to the individual gases present. A detector measures the reduced energy which has passed through the gas sample. A microprocessor analyses the detected values and displays CO₂ as a continuous waveform which may be displayed on a monitor or recorded onto a digital analogue tape (Figure 2.7.).

ETCO₂ levels may vary with changes in the rate of CO₂ production which is dependent on factors such as the work of breathing, CO₂ transport and elimination, nutritional support, cardiac output and the administration of exogenous CO₂ [Napolitano 1999, Weil 1985]. In subjects with healthy lungs, due to the effect of physiological dead space, an arterial/end-tidal CO₂ difference (usually \leq 6mmHg) exists but in conditions of cardiovascular stability ETCO₂ levels bear a constant relation to arterial CO₂ levels [Napolitano 1999]. Others have found that, in patients with normal pulmonary function and matching of ventilation-perfusion, ETCO₂ provides an accurate estimation of arterial pCO₂ [Hillier 1990, Tobias 1994, Campbell 1994, Napolitano 1999, Schmitz 1995, Rozycki 1998]. Capnography may, however, lose accuracy in very ill subjects [Schmitz 1995, Tobias 1997, Napolitano 1999], subjects with lung disease or unequal ventilation perfusion matching [Epstein 1985, Hand 1989, Sivan 1992], subjects with low tidal volumes [Tobias 1997] and during large decreases in cardiac output that may compromise alveolar blood flow [Weil 1985]. Capnography has been used in conjunction with the tilt table test [Serrador 1998, Yoshizaki 1998, Cencetti 1997, Novak 1998a, 1998b, Naschitz 1997] but there are unresolved issues regarding its use during a test that involves marked and sudden changes in cardiac output.

2.3.2. Transcutaneous carbon dioxide (TCO₂) measurement

A transcutaneous method (TINA, Radiometer, Copenhagen) [Larson 1993] of measuring CO₂ levels has been validated against arterial blood gas and arterialised ear lobe samples in a variety of situations [Trempler 1981, Mahutte 1984, Carter 1989, Sridhar 1993, Dawson 1998]. In neonates and infants with respiratory failure, transcutaneous CO₂ (TCO₂) monitoring has provided more accurate estimation of arterial pCO₂ than ETCO₂ monitoring [Tobias 1997, Rozycki 1998]. Prior to application, the TINA electrode is calibrated according to the manufacturer's instructions and heated to between 43° and 45° C, the recommended temperature range for adults. In our studies, a temperature of 43°C was chosen to minimise the risk of thermal injury to the skin. Optimal areas for TCO₂ measurement are the abdomen and chest due to high capillary density, ample capillary blood flow, thin epidermis and no shunting effects. The chosen site is cleaned with alcohol and the electrode attached with an adhesive fixation ring and contact electrolyte fluid (Figure 2.8.). Typical physiological stabilisation time of a subject is 3-7 minutes for pCO₂ measurements but this may be increased by lower electrode temperatures. The TINA electrode combines a heating element, two resistors acting as temperature sensors, a Clark-type oxygen electrode and a Severinghaus-type CO₂ electrode in a single unit. The temperature of the electrode is sensed by the resistor incorporated in the Ag/AgCl reference electrode. Due to high thermal conductivity of the silver body, the resistor responds quickly to any changes in temperature. The thermostating system maintains the electrode at the preset temperature. When the electrode is attached to the skin, the generated heat is transferred from the heating element via the silver body to the skin surface. The heating produces local vasodilatation and increases the permeability of the skin to oxygen and CO₂, rendering a measurement on the skin surface possible. The pCO_2 measurement is, in fact, a pH measurement. As CO₂ is released from the skin, it diffuses through the membrane into the electrolyte where it reacts with water forming carbonic acid which immediately dissociates into bicarbonate and hydrogen ions according to the equation:

 $H_2O + CO_2 \Leftrightarrow H_2CO_3 \Leftrightarrow H^+ + HCO_3^-$

The changes in H^+ in the electrolyte imply changes in pH. As the pH in the electrolyte changes, the voltage between the glass electrode and the reference electrode changes. The pH change is converted to a pCO₂ reading on the basis of the linear relationship between pH and log pCO₂, as expressed by the Henderson-Hasselbach equation:

$$pH = pK + \log \frac{\left[HCO_3^-\right]}{a.pCO_2}$$

where pK is the dissociation constant of carbonic acid, $[HCO_3^-]$ is the concentration of bicarbonate ions, *a* is the solubility coefficient of dissolved CO₂ and pCO_2 is the partial pressure of CO₂. As no charged molecules can penetrate the membrane, the change in pH is strictly due to the CO₂ diffusion into the electrolyte. The potential measured across the combined electrode is fed into the pCO₂ channel where it is digitised. This digitised signal is then passed onto the microcomputer where it is converted to display pCO₂ in mmHg or kPa.

The biggest drawback with TCO₂ monitoring is poor temporal resolution, with gas diffusion contributing to a response time of up to 50 seconds [Steurer 1997]. This limitation may be surmounted, however, by monitoring for periods in excess of this time lag after an expected change. The accuracy of TCO₂ measurements during changes in cardiac output is also a matter of some debate [Steurer 1997] and cutaneous hypoperfusion during periods of hypotension or hypocapnia could potentially reduce CO₂ diffusion transcutaneously and affect accuracy [Steurer 1997]. Technical variables affecting the accuracy of TCO₂ readings, such as trapped air bubbles, damaged membranes and improper or inappropriate placement techniques, were avoided in this work by careful use of the equipment by well trained staff. Problems associated with patients which could potentially affect reading accuracy, such as the presence of oedema and the administration of vasoconstricting drugs, did not arise in subjects recruited for these studies.

In conclusion, in skilled hands, $ETCO_2$ and TCO_2 measurement provide acceptable alternatives to invasive techniques for arterial pCO₂ estimation.

2.4. Assessment of Cerebral Autoregulation

2.4.1. Assessment of static cerebral autoregulation

Lack of temporal resolution in measurements of CBF with traditional techniques meant that most studies of CA prior to the advent of TCD were of static CA. Original animal studies usually induced static BP changes by pharmacological manipulation, haemorrhage or reperfusion with autologous blood. Human studies have induced pressor changes of the order of 20 mmHg through the use of intravenous phenylephrine or noradrenaline [Larsen 1994, Tiecks 1995, Strebel 1995], and depressor changes with passive head-up tilt [Grubb 1991a, Schondorf 1997] or lower body negative pressure [Bondar 1994, 1995, Giller 1992, Levine 1994]. Cerebral responses to static pressor and depressor changes are usually considered separately, as cerebral vasoconstriction and vasodilation may involve different physiological mechanisms and be affected differently by pathological processes. Studies of static CA usually involve calculation of an autoregulatory index (ARI) similar to that described by Tiecks et al [1995] where the percentage change in cerebrovascular resistance (CVR) per percentage change in BP is calculated. In our studies, data were manually selected to include a one-minute baseline period before the pressor or depressor stimulus as well as the subsequent period of static MAP

change during the stimulus. CVR can be estimated for each window and the static ARI calculated, as previously described by Tiecks et al [13], using:

$$CVR = \frac{MAP}{MCAV}$$

% $\Delta CVR = \frac{CVR_{end} - CVR_{base}}{CVR_{base}}$
% $\Delta MAP = \frac{MAP_{end} - MAP_{base}}{MAP_{base}}$
StaticARI = $\frac{\%\Delta CVR}{\%\Delta MAP} * 100\%$

where MAP_{end} and CVR_{end} represent values at the end of the period of static BP rise, and MAP_{base} and CVR_{base} represent values during the baseline period. A change in CVR that would fully compensate for the MAP change yields a result of 100%, and no change in CVR, i.e. no autoregulation, would give a result of 0%. Tiecks [1995] considered a value of >50% to be normal. These principles are illustrated in Figure 2.9.

2.4.2. Assessment of dynamic CA

As discussed earlier, dynamic CA is a relatively new concept and is usually considered as a separate entity from static CA as it may be effected by different physiological mechanisms. Assessment of dynamic CA is enabled by the excellent temporal resolution of TCD which allows examination of CBF changes in response to pressor and depressor BP changes occurring over a matter of seconds. Methods of assessing dynamic CA using spontaneous BP changes at rest have theoretical advantages over the use of induced dynamic BP changes, as stimuli used to induce dynamic BP changes may also induce sympathetic stimulation or changes in CO₂ levels, both of which may affect the integrity of CA (see section 2.5.).

2.4.2.1. Aaslid's Method

Perhaps the most widely used method of assessing dynamic CA involves the principle of inducing a dynamic BP change and assessing the length of time it takes for CBF to recover to baseline levels [Aaslid 1989, Newell 1994b, Tiecks 1995]. The rate of recovery (dROR) is assessed by:

$$dROR = \frac{\Delta CVR}{\Delta Time} \div \Delta ABP$$
$$\Delta CVR = \frac{\Delta ABP}{\Delta CBF}$$

The dROR represents the percentage recovery per second, with most normal subjects achieving a dROR of \geq 20% [Tiecks 1995]. The first studies described methods using rapid thigh-cuff deflation to induce a dynamic depressor change, but the methods are potentially applicable using other dynamic pressor and depressor stimuli, including spontaneous BP changes at rest.

Aaslid and Tiecks proposed a mathematical model to represent this principle to enable an autoregulatory index (ARI) to be calculated ranging from 0 (absent) to 9 (most efficient) [Tiecks 1995]. For an ABP change P(t), normalised by baseline values preceding the change, the relative pressure change dP(t) is given by [Tiecks 1995]:

$$dP(t) = \frac{P(t)}{1 - CrCP}$$

where *CrCP* is the critical closing pressure assumed to be normalised by the baseline pressure. The relative velocity change predicted by the model is given by:

$$V_{M}(t) = 1 + dP(t) - K \cdot x_{2}(t)$$

where K is a parameter reflecting system autoregulatory gain [Tiecks 1995] and $x_2(t)$ is an intermediate variable obtained from the following pair of equations:

$$x_{1}(t) = x_{1}(t-1) + \frac{dP(t) - x_{2}(t-1)}{f.T}$$
$$x_{2}(t) = x_{2}(t-1) + \frac{x_{1}(t) - 2D.x_{2}(t-1)}{f.T}$$

where f, D and T represent the sampling frequency, damping factor and time constant parameters respectively.

For each segment of data P(t), the model generated a corresponding predicted velocity $V_M(t)$ that was compared with the original data V(t) (Figure 2.10). Ten possible combinations of parameter values, corresponding to ARI values ranging from 0 to 9,

were tested and the one leading to the minimum square error corresponded to the solution adopted [Tiecks 1995].

A modification of these methods was subsequently described whereby the closest match is selected on the basis of the highest correlation coefficient [Dawson 2000] and fractional ARI values are obtained by parabolic interpolation around the point with least square error [Dawson 2000]. Unlike the least-squares method used by Aaslid and Tiecks [1995], this method does not rely on a specific value of critical closing pressure to select the closest match.

In this thesis, when applying these methods to induced and spontaneous ABP changes, data were marked manually at the point of initiation of the dynamic BP change. Only induced BP changes ≥ 10 mmHg and spontaneous BP changes ≥ 5 mmHg at rest were accepted as adequate dynamic stimuli, in keeping with the criteria adopted by others using similar methodology [Dawson 2000, Tiecks 1995, Strebel 1995, White 1997]. Subjects had to exhibit a synchronous and physiological change in BP and CBFV for all manual marks. A sample window of 30 seconds from the manual mark was used in all cases. In addition, these methods were also applied to one-minute segments of data from subjects when supine and in the head-up tilted positions.

These methods yield dynamic ARI values for right and left middle cerebral arteries for each individual for each dynamic BP change. As autoregulatory responses do not appear to depend on the cerebral hemisphere examined [Dawson 2000, Tiecks 1995, Strebel 1995], the mean of the two dynamic ARI values thus obtained was taken to be the ARI for each individual for each dynamic BP stimulus.

2.4.2.2. Frequency Domain Analysis

A number of methods have been recently devised, including frequency domain analysis, that allow quantification of dynamic CA from spontaneous BP changes at rest [Panerai 1998a, 1998b, 1999a, 1999b, Zhang 1998a,1998b]. Such methods of CA estimation are proving increasingly attractive, as mentioned previously, due to the uncertainty of the effects of induced dynamic BP changes on the integrity of CA. The use of spontaneous BP changes is also less noxious to subjects and poses less risk to patients potentially at risk from the induction of acute hypotensive or hypertensive changes e.g. patients post acute stroke or subarachnoid hemorrhage and those with carotid stenosis. In addition, assessment of CA at rest is possible over longer periods than with induced BP changes and has the potential, therefore, to yield more accurate results.

In this thesis, frequency domain analysis was performed on supine rest recordings of CBFV and ABP. Neglecting the effects of intracranial pressure and critical closing pressure [Dewey 1974, Panerai 1993], at a stable operating point, the relationship between CBFV and ABP can be written as:

$$V_o = \frac{P_o}{R_o} \tag{1}$$

where R_o represents the resistance-area product [Evans 1988]. For a small change in ABP, Δp , CA increases R_o by an amount Δr , and CBFV also changes by an amount Δv .

For small values of Δr and Δv , the product $\Delta r \cdot \Delta v$ can be neglected, and it is possible to write:

$$V_o + \Delta v = \frac{P_o + \Delta p}{R_o + \Delta r}$$
(2)

$$\frac{\Delta v}{V_o} \cong \frac{\Delta p}{P_o} - \frac{\Delta r}{R_o}$$
(3)

With perfect autoregulation, Δr compensates the Δp change and $\Delta v = 0$. At the other extreme, in the absence of autoregulation, $\Delta r = 0$, and the relative change in velocity is the same as in ABP. In either case, the above approximation becomes an equality, because the product $\Delta r \cdot \Delta v = 0$.

Given the linear properties of Fourier transform [Bendat 1986], equation 3 can be written in the frequency domain as:

$$\frac{V(f)}{V_{o}} = \frac{P(f)}{P_{o}} - \frac{R(f)}{R_{o}}$$
(4)

Assuming that changes in resistance are the result of changes in CBFV, it is possible to postulate a dynamic relationship between these two quantities in the form of a transfer function G(f), or:

$$R(f) = G(f).V(f)$$
⁽⁵⁾

Substituting in equation 4 and rearranging:

$$\frac{V(f)}{P(f)} = \frac{1}{R_o + V_o.G(f)} = H(f)$$
(6)

where H(f) is the transfer function between ABP and CBFV. The function G(f) can be regarded as the feedback gain of autoregulation [Panerai 1996]. When G(f) = 0, autoregulation is absent, and changes in ABP are completely transmitted to CBFV. At the other extreme, when the gain is high, H(f) is small, and the CBFV changes are equally small. Reynolds et al [1997] have provided a detailed analysis of the power spectra of V(f) and P(f) in neonates.

In our studies, cross spectral analysis of supine rest recordings of CBFV and ABP was performed with a segment of 512 samples selected from each record corresponding to the region with maximum ABP variability. For a segment of data with 256 samples, a 20% cosine tapered window was applied to v(n) and p(n), and the corresponding discrete Fourier transforms V(f) and P(f) computed with a FFT algorithm. The cross-spectrum was computed as [Bendat 1986]:

$$G_{PV}(f) = E[P(f).P(f)]$$
⁽⁷⁾

where the expected value of the complex product E[P(f).P(f)] was obtained by smoothing the spectra with a nine-point triangular moving average window and by averaging two segments of data. Similarly, the power spectra of p(n) was computed as:

$$G_{PP}(f) = E[P(f).P(f)]$$
(8)

The squared coherence function was estimated by [Bendat 1986]:

$$\gamma^{2}(f) = \frac{|G_{PV}(f)|^{2}}{G_{VV}(f).G_{PP}(f)}$$
(9)

and the transfer function was given by:

$$H(f) = \frac{G_{PV}(f)}{G_{PP}(f)}$$
(10)

From the real and imaginary parts of H(f) the amplitude and phase of the frequency response was calculated as:

$$|H(f)| = \left[H_R(f)^2 + H_I(f)^2\right]^{\frac{1}{2}}$$
(11)

$$\phi(f) = \tan^{-1} \frac{H_I(f)}{H_R(f)}$$
(12)

The frequency response was multiplied by a cosine-shaped low-pass filter with cut-off frequency at 1.0 Hz and the impulse response $h_{PV}(n)$ was obtained by the inverse Fourier transform using the same FFT algorithm. The step response was obtained by integration of $h_{PV}(n)$ for positive values of time.

Frequency domain analysis has been used to demonstrate impairment of dynamic CA in premature neonates, hypercapnia and severe carotid artery stenosis [Panerai 1998a, 1998b, 1999b]. High coherence between MCA BP and CBFV suggests that BP changes are transmitted passively to the cerebral circulation, indicating impaired CA. On the other hand, active CA will attenuate CBFV changes in response to BP changes and coherence will fall. Reduced coherence at very low frequencies (<0.1 Hz) has been shown to be indicative of active dynamic CA [Panerai 1998a, 1999b, Zhang 1998a, 1998b] (Figure 2.11a.). In subjects with normal CA, oscillations in CBFV lead BP oscillations by a positive phase angle i.e. changes in CBFV occur before those in systemic BP [Panerai 1998a, 1999b, Zhang 1998a, 1998b]. In subjects with impaired dynamic CA, the phase angle becomes less positive [Panerai 1998a, 1998b, 1999b, Zhang 1998a, 1998b] (Figure 2.11b.) and deterioration of CA may be reflected more markedly in the phase rather than in the amplitude frequency response [Panerai 1998a, 1998b] which may demonstrate an increase at low frequencies if CA is impaired (Figure 2.11c.).

The impulse response function represents the CBFV response to a very short, impulse-like disturbance in ABP. Impulse response plots show an initial positive wave and subsequent negative deflection in CBFV when dynamic CA is active, with impairment leading to blunting or disappearance of the negative deflection [Panerai 1998a, 1998b, 1999, Zhang 1998a, 1998b] (Figure 2.11d.). Step response plots demonstrate a return of CBFV to baseline values when dynamic CA is intact and higher values when impaired [Panerai 1998a, 1998b, 1999b, Zhang 1998a, 1998b] (Figure 2.11e.). In summary, both induced and spontaneous pressor and depressor BP changes can be used with Aaslid's model and frequency domain analysis to assess the integrity of dynamic CA.

2.5. Methods used to induce static and dynamic BP changes

In this work, a number of methods were used to induce pressor and depressor BP changes so that CA could be investigated (Table 2.2.). In this section, I will discuss the practical aspects and physiology of the various tests, along with any known effects of the tests themselves on CBF.

2.5.1. Thigh-cuff inflation and release - static pressor and dynamic depressor stimuli (Figure 2.12.)

This technique was first described by Aaslid [1989] as a non-invasive, nonpharmacological method of effecting a rapid fall in MAP for the purpose of CA assessment. The original description of the technique involved inflation of bilateral thigh-cuffs to 30mmHg above systolic BP for 3 minutes with subsequent rapid release achieving a mean fall in MAP of >20mmHg lasting >15 seconds. Subsequent work has validated this technique in a number of subject groups [Tiecks 1995, Strebel 1995, Dawson 2000], while illustrating that inflation to systolic BP levels or just above for 90 seconds produces BP falls of similar magnitude without causing as much patient discomfort.

In addition, it became clear that the period of thigh-cuff inflation caused a static rise in MAP, with a less marked rise in heart rate, presumably reflecting sympathetic stimulation due to discomfort. Consequently, this technique can be used as a means of inducing static and dynamic BP changes, but there are a number of drawbacks. Sympathetic stimulation, as discussed in section 1.3.2.4., may have effects on CA and compromise results but there is evidence that changes in sympathetic nervous system activity may not affect CBF in the BP range likely to be examined by thigh-cuff inflation and release i.e. 60-140mmHg [Roatta 1998, Eklöf 1971, Skinhøj 1972, 1973, Heistad 1978a, Hernandez-Perez 1975, Harper 1975]. A build up of blood rich in CO₂ and other potential metabolites in the legs during inflation with subsequent arrival in the brain after release is another potential confounding factor, but this may be minimised by using the shorter inflation period. The shorter inflation period may also help to minimise factors related to the subject that may also affect CA results, such as inadvertent breath holding or Valsalva manoeuvre performance during the test.

2.5.2. Valsalva manoeuvre - a dynamic pressor stimulus (Figure 2.13.)

The Valsalva manoeuvre, or forced expiration against a closed glottis, causes marked systemic haemodynamic changes that may be described in four separate phases. Phase I represents the start of the strain and is associated with a rise in ABP because of increased intrathoracic pressure. During phase II, as atrial filling decreases, a compensatory sympathetic response leads to an increase in heart rate, cardiac output and BP. When release of the strain causes a sudden fall in intrathoracic pressure (phase III), a sharp fall in BP is rapidly followed by a large BP overshoot due to the persistent sympathetic activity and high peripheral resistance (phase IV). BP levels gradually return to baseline over the next 30-60 seconds [Hamilton 1936]. The MCAV responds to these BP changes and the velocity changes can be used to calculate an autoregulatory index [Tiecks 1996].

The standard method of performing the manoeuvre is to blow into a syringe fitted with a constant bleed device which is attached to a transducer to record intrathoracic pressure. Subjects aim to achieve 40mmHg pressure for 15 seconds after a normal inspiratory breath [Palmero 1981, Smith 1987].

Phase IV of the manoeuvre is easily recognisable and reproducible, and, therefore, has been used as a dynamic pressor stimulus to derive indices of dynamic CA [Tiecks 1996]. Again, there are potential drawbacks when using the manoeuvre for calculating autoregulatory indices as the influences of raised intracranial pressure during strain and hypercapnia while breath holding on such indices are unknown.

2.5.3. Lower body negative pressure and release - static depressor and dynamic pressor stimuli (Figure 2.14.)

Lower body negative pressure (LBNP) is achieved by placing a subject's lower body in an air-tight box and sealing at the level of the iliac crests. Negative pressure is then applied with a pump, such as a domestic vacuum cleaner, to create negative pressure relative to the rest of the body in the lower body. The pressure is monitored with the aid of a manometer in series and can be varied according to the protocol required to be hypotensive (usually >-20mmHg) or non-hypotensive (usually < -10 to -15mmHg). Negative pressure has the effect of triggering venous pooling in the lower limbs and pelvis and causing a central hypovolaemic state with a fall in cardiac output and stroke volume [Stevens 1965]. Pooling of blood during LBNP of between -30 and -50 mmHg has been estimated at between 0.5 to 1.0 litres [Wolthuis 1970]. Nonhypotensive LBNP activates arterial high pressure baroreceptors and extraarterial low pressure cardiopulmonary mechanoreceptors leading to maintenance of MAP through an increase in heart rate and total peripheral resistance (TPR) [Stevens 1965, Wolthuis 1974]. As the degree and duration of LBNP increase, the capacity of these mechanisms to compensate is exceeded and MAP starts to decline (hypotensive LBNP), often to the point of syncope [Stevens 1965, Bondar 1994, 1995]. Gender and age related differences have been demonstrated in the responses to LBNP, with men demonstrating more calf pooling and women more thoracic impedance [Frey 1988]. Heart rate and TPR responses have been shown to diminish with age [Frey 1988] while athletes have a reduced tolerance to LBNP, possibly due to a high vagal tone or a decreased sympathoadrenal response [Raven 1993,1984].

If pressure is carefully controlled, a static decline in BP can be achieved. Rapid release of LBNP causes a sudden rise in circulating volume and, in conjunction with the persistent elevation in TPR, results in a dynamic rise in BP.

Many studies have examined the effect of LBNP on CBF, primarily in fit young men [Giller 1992, Ueno 1993, Bondar 1994, 1995, Levine 1994, Balldin 1996, Zhang 1998b]. Most investigators have demonstrated a decrease in MCAV with increasingly negative pressures and some have used indices of cerebrovascular resistance to suggest that the decrease in MCAV is due to cerebral vasoconstriction [Giller 1992, Levine 1994, Bondar 1995]. Consequently, hypotheses of sympathetic induced vasoconstriction overriding CA gained favour, especially as vasoconstriction was also demonstrated in the renal vascular bed [Hirsch 1989]. Cerebral vasoconstriction during LBNP may due to hypocapnia rather than sympathetic nerve activity [Zhang 1998b]. It is possible, therefore, that CA may actually improve during LBNP, due to CO₂ changes, rather than being compromised (see section 1.2.3.1). CO₂ and sympathetic changes, therefore, may limit the usefulness of LBNP as a BP stimulus for CA assessment.
In summary, the above manoeuvres may be employed as static and dynamic pressor and depressor stimuli to assess CA. It is important to use pressor and depressor and static and dynamic stimuli as CA responses to these different stimuli may be different and be affected differently by pathology. It is clear that the above BP stimuli may stimulate the autonomic nervous system and evoke changes in pCO_2 to some degree, making interpretation of results difficult. For these reasons, spontaneous pressor and depressor BP changes at rest may prove more useful when studying CA.

2.6. Study protocol

The initial part of the protocol of all studies described in this thesis was identical and is outlined here.

Subjects avoided caffeine containing products, nicotine and alcohol for at least 12 hours prior to the studies. They wore loose comfortable clothing and attended in the morning between 09.00 and 11.00 a minimum of 2 hours after a light breakfast. All recordings were made in a dedicated research room kept at constant ambient temperature (21-24°C) with external stimuli minimised.

Subjects lay supine on a padded table (Figure 2.3.) capable of being manually tilted with their head supported by two pillows. After lying supine for 10 minutes, three semi-automated BP readings were taken 1minute apart (Omron 711, Japan). The mean of the last 2 readings, providing pressures differed by less than 10 mmHg, was taken as the baseline casual BP measurement. A surface 3 lead ECG was fitted for measurement of heart rate and arterial BP was measured continuously and non-invasively from the middle finger of the non-dominant hand using a servo-controlled plethysmograph (Finapres 2300, Ohmeda, USA, Figure 2.6, Section 2.2). An

appropriate cuff size was chosen and the hand supported at the level of the right atrium using a custom-made arm rest.

Transcutaneous carbon dioxide (CO_2) partial pressure was measured using a transcutaneous gas monitor (TINA, Radiometer, Copenhagen, Figure 2.8, Section 2.3.2) validated previously [Dawson 1998]. The probe was applied, as outlined in section 2.3.2., to the skin of each subject at heart level in the anterior axillary line (Figure 2.8.).

To measure CBFV, both middle cerebral arteries were identified and insonated as described by in section 2.1.5. [Aaslid 1982] using bilateral 2MHz pulse transcranial Doppler ultrasound (SciMed QVL 842X, Bristol, UK). Both Doppler probes were held in place using a custom designed adjustable head frame and the velocity waveform spectra were visually displayed to aid positioning (Figure 2.3.). The Doppler frequency shift and the other parameters were recorded onto a digital audio-tape (Sony Digital Instrumentation Cassette Recorder PC-108M). Once the subjects had rested supine for a minimum of 30 minutes and the Finapres and TCO₂ traces had stabilized (<10% variation over 5 minutes), recording was started. Subjects were asked to lie still and to refrain from talking during the recordings.

2.7. Data analysis

Preliminary analysis of data recorded onto digital audio tapes was identical in all studies described in this thesis and is described here. Data were downloaded in real time onto a dedicated personal computer. A Fast Fourier Transform method was used to convert the Doppler signals into maximum frequency velocity envelopes. A window of 5ms was used to achieve temporal resolution. The Finapres, TCO_2 and ECG output signals were directly converted at 200 Hz.

Data were visually inspected, the BP trace was calibrated and artefactual data spikes were removed by linear interpolation. Each cardiac cycle was automatically marked to determine the R-R interval from the ECG tracing with ectopic beats being manually marked and removed by linear interpolation. Ectopic beats occurring with a frequency of more than one per 30 seconds led to rejection of the data. Using spline interpolation (equivalent to a third order polynomial) and resampling the data at 0.2 seconds, a uniform time base for all the data was achieved. Estimates of systolic, diastolic and mean BP, systolic, diastolic and mean CBFV, pulse interval and TCO₂ were made for each cardiac cycle.

Figure 2.1. Circle of Willis



ACA	=	anterior communicating artery		
PCA	=	posterior communicating artery		
BA	=	basilar artery		
VA	=	vertebral artery		
ICA	=	internal carotid artery		
PComA	=	posterior communicating artery		
AcomA	=	anterior communicating artery		
MCA	=	middle cerebral artery		
PICA	=	posterior inferior cerebellar artery		
AICA	=	anterior inferior cerebellar artery		

- 91





F	=	frontal
A	=	anterior
М	=	middle
Р	_	posterior

Figure 2.3. A subject undergoing transcranial Doppler ultrasound with the SciMed 120 QVL apparatus







Figure 2.5. Method of Calculation of CrCP and RAP



Figure 2.6. The Ohmeda 2300 Finapres device with cuff applied to the middle finger of the non-dominant hand





Figure 2.7. End-tidal carbon dioxide monitoring with the Capnogard infa-red capnograph and face-mask

Figure 2.8. TINA transcutaneous gas monitor with probe attached to the skin at heart level in the anterior axillary line







Time





MAP

Figure 2.10. Aaslid's Model



Mathematical model curves of dynamic cerebral autoregulation in response to a step change in BP.





Figure 2.11b. Phase frequency responses of a subject with normal CA (black) and a subject with impaired CA (red)



Figure 2.11b.



Figure 2.11c. Amplitude plots of a subject with normal CA (black) and a subject with impaired CA (red)

Figure 2.11d. Impulse response plots of a subject with normal CA (black) and a subject with impaired CA (red)



Figure 2.11d.





Figure 2.12. A volunteer undergoing thigh-cuff inflation

Figure 2.13. A volunteer performing a Valsalva manoeuvre





Figure 2.14. Application of lower body negative pressure

Artery	Transducer Position	Depth of sampling (mm)	Direction of Flow	Spatial Relationship ACA/MCA Bifurcation	Mean Velocity (cm/sec)	Response to Ipsilateral Carotid Compression
MCA	Transtemporal	30-60	Toward	Same	55 ± 12	÷
ACA/MCA Bifurcation	Transtemporal	55-65	Bidirectional	-	-	ţ
ACA	Transtemporal	60-80	Away	Anterior & Superior	50 ± 11	ł
Proximal PCA	Transtemporal	60-70	Toward	Posterior & Inferior	39 ± 10	↔↑
Distal PCA	Transtemporal	60-70	Away	Posterior & Inferior	40 ± 10	↔↓
TICA	Transtemporal	55-65	Toward	Inferior	39 ± 9	Ļ
OA	Transorbital	40-60	Toward	-	21 ± 5	Ļ
VA	Transforaminal	60-90	Away	-	38 ± 10	· _
BA	Transforaminal	80-120	Away	-	41 ± 10	-

 Table 2.1.
 Identification of basal cerebral arteries

ACA	=	anterior cerebral artery
MCA	=	middle cerebral artery
PCA	=	posterior cerebral artery
TICA	=	terminal internal carotid artery
OA	=	ophthalmic artery
VA	=	vertebral artery
BA	=	basilar artery

Table 2.2. Static and Dynamic Pressor & Depressor Changes

Stimulus	Induced	Static	Dynamic	Pressor	Depressor
Thigh cuff inflation	~	~	×	~	×
Thigh cuff release	~	x	~	×	~
Lower body negative pressure	~	~	×	×	~
Lower body negative pressure release	~	×	1	1	×
Phase IV of the Valsalva manoeuvre	~	×	1	1	×
Spontaneous BP changes at rest	×	×	1	1	~

CHAPTER THREE

Effect of Ageing on Cerebral Autoregulation

_____ 112

3.1. Summary

Normal ageing is associated with marked changes in the cardiovascular and cerebrovascular systems. Although cerebral autoregulation (CA) is impaired in certain disease states, the effect of age per se on CA in humans is unknown. 27 young subjects (≤ 40 years) and 27 older subjects (≥ 55 years), matched for sex and systolic blood pressure (BP), underwent measurement of cerebral blood flow velocity using transcranial Doppler ultrasound and non-invasive beat-to-beat arterial BP measurement during induced and spontaneous static and dynamic BP stimuli. Standard static and dynamic autoregulatory index (ARI) values were derived for each spontaneous and induced static and dynamic BP stimulus as well as cardiac baroreceptor sensitivity (BRS) and step response values for the two groups. The mean age of the young group was 29 ± 5 years and older group was 68 ± 5 years. Cardiac BRS was reduced in the older group $(8.6 \pm 4.5 \text{ v} 16.9 \pm 8.8 \text{ ms/mmHg})$ p<0.0001). However, no age-related differences were demonstrated in step response plots or in ARI values for pressor (0.80) or depressor (0.48) static or dynamic (p=0.62) BP stimuli (p=0.62). Mean dynamic ARI values for all stimuli combined were 4.9 ± 1.8 for the young group and 5.0 ± 2.3 for the older group. Mean static ARI values were $54 \pm 30\%$ for the young group and $56 \pm 32\%$ for the older group. Although increasing age is associated with a decrease in cardiac BRS and CBFV, dynamic CA, as assessed using step response analysis as well as cerebral blood flow responses to transient and induced static and dynamic BP stimuli, is unaffected by ageing.

3.2. Background

The cardiovascular system demonstrates many age-related changes as evidenced by the well-recognised increase in systolic blood pressure (SBP) [Kannel 1978] and decreases in systemic artery compliance and cardiac BRS with age [Fleg 1986, Gribbin 1971]. Ageing may also be associated with changes in cerebrovascular haemodynamics with both cerebral blood flow volume and cerebral blood flow velocities being reported as declining with advancing years [Matsuda 1984, Krejza 1999].

Cerebral autoregulation (CA) refers to the inherent ability of the cerebral blood vessels to keep cerebral blood flow (CBF) constant for a wide range of systemic BP levels [Lassen 1959, Paulson 1990]. CA occurs with a substantial degree of temporal heterogeneity in that physiological adjustments of CBF occur both quickly and slowly. Dynamic CA refers to the ability to maintain CBF in the face of BP changes occurring over a matter of seconds and reflects the latency of the cerebral vasoregulatory system. Static CA refers to CBF adjustments in response to more prolonged BP changes and is a measure of the overall efficiency of the system. Dynamic CA is impaired in a variety of disease states including post head-injury [Czosnyka 1996, Smielewski 1997, Steiger 1994], subarachnoid haemorrhage [Smielewski 1995], acute ischaemic stroke [Dawson 2000] and severe carotid artery disease [White 1997]. It has been suggested that static and dynamic CA may have different control mechanisms, and that dynamic CA may be more susceptible to damage in pathological states [Tiecks 1995] as seen following acute ischaemic stroke [Dawson 2000]. As disease states likely to affect CA are more common in the elderly, it is important to know if ageing per se affects CA.

The advent of transcranial Doppler ultrasonography has facilitated the non-invasive measurement of CBF haemodynamics by allowing cerebral blood flow velocity

(CBFV) changes in response to static and dynamic BP changes, and hence CA, to be calculated [Tiecks 1995, Aaslid 1989]. Methods of CA estimation using spontaneous BP changes may have advantages over methods using induced BP changes. Such methods include the new step response methodology [Panerai 1998a, 1999b, Zhang 1998a], and do not induce sympathetic stimulation or hypocapnia, both of which may affect the integrity of CA (see sections 1.3.2.4. & 1.3.3.1.). The use of spontaneous BP changes is also less noxious to subjects and poses less risk to patients potentially at risk from the induction of acute hypotensive or hypertensive changes e.g. patients post acute stroke or subarachnoid haemorrhage and those with carotid stenosis. Although the effect of ageing on CA has been studied in animals [Hoffman 1981], the effects of physiological ageing on CA in humans are unknown. This study was designed to assess the effect of human ageing on static and dynamic CA to spontaneous and induced pressor and depressor stimuli.

3.3. Objectives

- 1. To assess the effect of normal ageing on static cerebral autoregulation.
- 2. To assess the effect of normal ageing on dynamic cerebral autoregulation.
- 3. To examine the effect of the magnitude of pressor and depressor changes on static and dynamic CA.

3.4. Methods

3.4.1. Subjects

Subjects were recruited from a volunteer register in the department and from departmental staff. All were healthy, on no medications and free from cardiovascular, cerebrovascular or autonomic disease as based on history and clinical examination. Significant silent carotid stenosis was excluded on the basis of bilateral neck auscultation and the absence of typical transcranial Doppler ultrasound findings [Wilterdink 1997]. Subjects were assigned by age to a young group (\leq 40 years) or an older group (\geq 55 years) being pair-matched for sex, body mass index (BMI) (within 2.0 kg/m²) and systolic BP (within 10mmHg) using BP measurements made according to the protocol described in section 2.6.

3.4.2. Study Protocol

All subjects underwent the initial preparatory protocol outlined in section 2.6. After resting supine for a minimum of 30 minutes and when the Finapres and TCO_2 traces had stabilized (<10% variation over 5 minutes), two baseline recordings of at least 5 minutes each were made during supine rest for each subject. Subsequently, the pressor and depressor stimuli were applied in random order with a baseline recording of 60 seconds before and after each test.

3.4.2.1. Static pressor stimulus

Bilateral thigh cuffs were applied and inflated to suprasystolic pressures for 90 seconds with thigh cuff pressure monitored using a mercury sphygmomanometer, as described in section 2.5.1.

3.4.2.2. Static depressor stimulus

The lower limbs of the subjects were sealed at the level of the iliac crests in a custom designed box and lower body negative pressure (LBNP) applied (range 15-58

mmHg), as described in section 2.5.3, for 5 minutes to cause a fall of at least 10 mmHg in systolic BP.

3.4.2.3. Dynamic pressor stimuli

 LBNP release – After 5 minutes of the static depressor BP stimulus outlined above (3.4.2.2.), atmospheric pressure was suddenly restored to the box by disconnecting the pressure box.

II) Valsalva manoeuvre (phase IV) - Subjects were asked to perform a Valsalva manoeuvre lasting 15 seconds whilst supine by blowing into a tube connected to a transducer, as described in section 2.5.2, which registered intrathoracic pressures to aid compliance. After completion, subjects lay quietly for one minute while recording continued. This procedure was repeated twice so that each subject performed 3 Valsalva manoeuvres one minute apart.

III) Three spontaneous rises in BP >5mmHg were selected at random from each of the two 5-minute rest recordings. A total of six spontaneous pressor changes were, therefore, selected for each subject and the mean value taken.

3.4.2.4. Dynamic depressor stimuli

 Thigh cuff release – After 90 seconds of the static pressor BP stimulus outlined above (section 3.4.2.1), the cuffs were rapidly deflated while recording continued for a further minute. This test was repeated in some patients to increase the likelihood of obtaining a BP fall of >10 mmHg.

II) Three spontaneous falls in BP >5mmHg were selected at random from each of the two rest recordings. A total of six spontaneous depressor changes were, therefore, selected for each subject.

3.4.3. Data analysis

Data were initially analysed as outlined in section 2.7. In addition, ectopic beats occurring in the 15-second window prior to or after induced and spontaneous pressor and depressor BP changes led to the rejection of the data.

3.4.4. Determination of cerebral autoregulation

3.4.4.1. Determination of static CA

Indices of static CA ranging from 0% (absent) to 100% (most efficient) were derived using the methods outlined in section 2.4.1. Data were manually selected to include a one-minute baseline period before the pressor or depressor stimuli as well as the subsequent period of static MAP change during the stimuli.

3.4.4.2. Determination of dynamic CA using Aaslid's model

Dynamic CA was graded by generating a dynamic autoregulatory index (ARI), ranging from 0 (absent) to 9 (most efficient), for each subject for each dynamic pressor and depressor stimulus using the methods outlined in sectioned 2.4.2.1. Data were marked manually at the point of initiation of the dynamic BP change. Only induced BP changes \geq 10mmHg and spontaneous BP changes \geq 5mmHg at rest were accepted as adequate dynamic stimuli. Subjects had to exhibit a synchronous and physiological change in BP and MCA CBFV for all manual marks. A sample window of 30 seconds from the manual mark was used in all cases. Each dynamic BP stimulus yielded an ARI for right and left middle cerebral arteries for each individual. The mean of the two ARI values thus obtained was taken to be the ARI for that individual for that dynamic BP stimulus. For thigh cuff release, phase IV of the Valsalva manoeuvre and spontaneous pressor and depressor changes the mean of the ARI values derived for each adequate stimulus was taken to be the ARI for that dynamic BP stimulus.

3.4.4.3. Determination of dynamic CA using step response analysis

The step response of both young and older groups was derived using frequency domain analysis as outlined in section 2.4.2.2.

3.4.5. Determination of cardiac baroreceptor sensitivity

The PI and SBP series were analyzed by means of power spectral analysis with a fast Fourier transform with 512 samples. The data segments used were extracted under visual inspection from the most stable segment of each rest recording. The beat-tobeat series of PI and SBP were interpolated with a third-order polynomial and resampled with an interval of 0.5s to produce signals with a uniform time axis. The power spectra were obtained as the average of three recordings for each subject and were smoothed with a 13-point triangular window. This produced estimate of power spectra of PI and SBP, coherence function and frequency response between PI and SBP with 58 degrees of freedom. Coherence between BP and PI variability reflects the amount of linear coupling between the two spectra and is, therefore, comparable to the correlation coefficient in regression analysis [Robinson 1997]. A coherence value >0.4 was considered significant [Panerai 1997]. Power spectral analysis estimates of cardiac BRS were obtained by calculation of the α -index (square root of the powers of PI to BP) for the low frequency band (0.05 to 0.15 Hz). To correct for variability in total and very low frequency (0.02 to 0.05 Hz) powers, the low frequency power spectra for PI and SBP were calculated in normalised units [Montano 1994] as follows:

Power = Absolute power / (total power - very low frequency power)*100

3.4.6. Statistical Analysis

Data were analysed using SAS version 6.12 and Minitab 12 software packages. Results are presented as mean \pm standard deviation (range) and statistical significance set at p<0.05 level.

With all 5 dynamic BP stimuli, it was assumed that missing data were lost in random fashion i.e. that the probability of a missing ARI was not a function of the subject's ARI. A normal distribution was assumed and a good approximation to this was demonstrated by a plot of residuals. Using mixed modelling age by stimulus interaction, age effect, estimated difference (young v old) and stimulus effect were calculated.

As the two groups were pair-matched, Student's paired t-tests were used to compare ARI values derived for each BP stimulus between the two groups to see whether any individual test demonstrated a difference.

The differences between ARI values derived from the 5 different stimuli were estimated using a mixed model for repeated measures data. Different covariance patterns were investigated using Akaike's Information Criterion. Multiple pairwise testing was adjusted for using Tukey's method.

Mean step responses and their standard errors were calculated for both young and older groups [Panerai 1998a, 1999b]. The step responses at 5 seconds were compared between the two groups using Student's paired t-tests.

Static CA indices were compared between the young and older groups using Student's paired t tests.

3.5. Results

Demographic and baseline data for the young (n=27) and older groups (n=27) are contained in Table 3.1. Despite matching for systolic BP and BMI, both parameters were significantly higher in the older group (p < 0.001 and p=0.003 respectively), but mean BP and diastolic BP were similar in the two groups. Baseline mean CBFV was significantly higher in the younger group (p=0.006) as was mean baseline cardiac BRS (p<0.0001).

The average BP changes achieved by the different static and dynamic stimuli are shown in Table 3.2. Regression analysis failed to show a significant correlation between ARI values and the magnitude of dynamic (p=0.82) or static (p=0.39) BP change.

No differences were demonstrated between the young and older groups for static pressor or depressor ARI values (Table 3.2.). Mixed modelling did not demonstrate any variation of the effect of age on ARI values according to the dynamic stimulus used (age by stimulus interaction), p=0.36.

When a constant effect across all dynamic stimuli was assumed, no age effect was shown (p=0.62) with the estimated difference in ARI between young and older groups being -0.18 (95% CI; -0.93 to 0.56). No significant differences were shown between the two groups for ARI values derived using any of the 5 dynamic BP stimuli (see Table 3.2.).

When age effect was excluded from the final model and the effect of dynamic stimulus on ARI values was estimated for young and older subjects combined, a significant effect was demonstrated (p=0.033). Further comparison, using Tukey's method of adjusting p values for multiple comparisons, showed that ARI values derived from thigh cuff release were significantly greater those from LBNP release

(difference=1.27, 95%CI 0.03-2.50, p=0.042), though no other significant differences were found.

When all dynamic ARI values were combined, the young group had a mean ARI value of 4.9 ± 1.8 compared to 5.0 ± 2.3 for the older group (difference = -0.1; 95% CI -0.6 t 0.7; p=0.88).

Mean step response plots for the two groups, including standard errors, are displayed in Figure 3.1. No significant differences were shown between the step responses of the groups at 5 seconds (103.5 v 102.9, difference=0.6, p=0.37).

3.6. Discussion

Certain important aspects of cerebral haemodynamics have been shown to change with age, with CBF volume and CBFV decreasing and cerebral arterial vessels widening in diameter [Matsuda 1984, Krejza 1999]. The reasons behind the decline in CBF and CBFV with age are unknown but may be due to the increase in arterial vessel diameter [Matsuda 1984] or decreasing metabolic demand [Murphy 1996]. Dynamic, as opposed to static, CA has been shown to be the more vulnerable component to impairment in certain disease states [Dawson 2000, Tiecks 1995] and, as pathological changes become more prevalent with age, it is important to know if age per se causes deterioration in CA. That such changes take place in old age may have clinical importance and help explain, for example, the poor correlation between postural symptoms and systemic BP changes to standing in the elderly and in certain syncope syndromes [Wynne 1996]. In this study, no change was demonstrated in static CA with age and, despite the use of two different methodologies (Aaslid's model and frequency domain analysis), pressor and depressor BP stimuli and induced and spontaneous static and dynamic BP changes, no differences were demonstrated in static or dynamic CA with age. Similar methods have detected impaired dynamic CA in carotid artery stenosis [Panerai 1998b], after acute ischaemic stroke [Dawson 2000], in neonates [Panerai 1998a] and during hypercapnia [Panerai 1999b]. The older group had lower cardiac BRS and mean CBFV values than their younger counterparts, suggesting that the "normal" ageing process in other cardiovascular and cerebrovascular parameters was present in the subjects recruited [Gribbin 1971, Krejza 1999, Fuledsi 1997].

3.6.1. Pressor and depressor changes

Thigh cuff inflation, as a static pressor stimulus, and thigh cuff release and the Valsalva manoeuvre, as dynamic pressor and depressor stimuli respectively, have previously been used to assess CA [Dawson 2000, Tiecks 1995, 1996, White 1997]. Thigh cuff release is probably the most widely applied dynamic BP stimulus, but it is unknown if ARI values are dependent on the type of stimulus used, the direction of BP change (pressor or depressor) or the magnitude of BP change induced. The author is unaware of any previous reports of the use of LBNP release or spontaneous transient changes in BP at rest with Aaslid's model to assess CA. Although we have demonstrated significant differences between ARI values derived using different dynamic BP stimuli (thigh cuff release and LBNP release), no age-related differences were found.

3.6.2. Autoregulatory indices

Although Aaslid's dynamic ARI is an arbitrary model, it does give important clinical information, e.g. acute ischaemic stroke patients have dynamic ARI values in the order of 2.0 lower than age and BP matched controls [Dawson 2000]. Similar

methods have shown dynamic ARI values to be a mean of 2.6 lower than control values in patients with carotid artery disease [White 1997]. The static and dynamic ARI values in this study are consistent with those published previously [Dawson 2000, White 1997, Tiecks 1995, Mahony 2000] and demonstrate a dynamic ARI difference of only -0.18 between young and older groups with a 95% confidence interval of -0.93 to 0.56. These confidence intervals are of a magnitude that suggest that there are is no clinically significant age-related deterioration in dynamic CA.

3.6.3. Magnitude of blood pressure changes

The magnitudes of dynamic BP changes were significantly greater in the young for phase IV of the Valsalva manoeuvre but no differences were shown for any of the other dynamic BP stimuli. No significant correlation was found, however, between the magnitude of static and dynamic BP changes and ARI values. It is also worth noting that phase IV of the Valsalva manoeuvre caused the largest dynamic BP changes in both groups but yielded neither the highest nor lowest ARI values. Static BP changes between the two static stimuli are not comparable due to the protocol of LBNP application adopted.

3.6.4. Step response analysis

Step response analysis has been used to demonstrate impairment of dynamic CA in premature neonates [Panerai 1998a], hypercapnia [Panerai 1999b] and severe carotid artery stenosis [Panerai 1998b]. Step response plots characteristically demonstrate a return of CBFV to baseline values when dynamic CA is intact and higher values when impaired. The step response plots were remarkably similar for the two groups and indicate an active autoregulation in both groups.
3.6.5. Age and cerebrovascular function

Kastrup [1998] found that cerebrovascular CO_2 reactivity was not affected by ageing in men but decreased with age in women. A sub-analysis of the female subjects in this study demonstrated no differences in dynamic CA with age. It has to be remembered, however, that CO_2 reactivity and dynamic CA are different entities, the former possibly representing only one of the elements of the latter. If post-menopausal differences in CO_2 reactivity truly exist, they do not appear to affect the integrity of dynamic CA.

This is the first study to show that ageing does not impair CA while demonstrating that ageing does alter other cardiovascular and cerebrovascular haemodynamic parameters i.e. cardiac BRS and CBFV. CA is a function of multiple physiological processes including metabolic, myogenic, neuronal and, possibly, nitric oxide (NO) and endothelin-mediated endothelial mechanisms [White 1998, Thorin 1998]. It is interesting that, although arterial compliance [Fleg 1986], CBFV [Krejza 1999], CBF [Matsuda 1984] and, possibly, endothelium-mediated vascular relaxation [Hongo 1988] decline with age, the ability of the cerebrovasculature to compensate for BP changes remains undiminished. This suggests that the CA system may have an inbuilt reserve that allows for degeneration of the individual elements with time. If this is the case, it is possible that CA reserve may deteriorate with age and that impairment of CA may only be uncovered if the reserve is stretched to capacity e.g. at BP or age extremes, with large BP changes or in disease states. CBFV values in the young and older groups were comparable to values reported elsewhere and demonstrated the significant age-related decline one would expect [Krejza 1999, Fuledsi 1997]. It is unknown if the magnitude of CBFV has any effect on CA, but it would appear not to

be the case in this study as ARI values were similar in our young and old groups even though mean CBFV was significantly lower in the older group.

3.6.6. Limitations

Despite matching, the older group as a whole had a small though significantly higher systolic BP than the young group but mean BP and diastolic BP were well matched between the group pairs. Increasing BP levels cause a shift in the static autoregulatory curve to the right [Strandgaard 1973, 1976, 1989, Hoffman 1981, Fujishima 1984], but there is no evidence that increases in systemic BP levels impair or facilitate dynamic CA [Blake 2000]. Similarly, there is no evidence that increases in BMI are associated with changes in dynamic CA. If such changes did occur with increasing BMI, one could hypothesise that they would be mediated by increases in arterial BP, but this parameter has been reasonably well controlled.

The static and dynamic BP stimuli used in this study for assessing CA probably did not produce big enough BP changes to exceed the upper and lower BP limits within which CA is active, and the possibility of a narrowing with age of the BP plateau could not be addressed. In addition, only two static BP stimuli were employed resulting in quite large confidence intervals and the conclusions that can be drawn regarding the effect of ageing on static CA are, therefore, somewhat limited. The major limitation of this work, and indeed, any work using similar methodology, is that we have measured cerebral blood flow velocity rather than cerebral blood flow. Changes in CBF can only be reliably deduced from CBFV changes if the diameter of the insonated vessel remains unchanged [Aaslid 1982], but reliable, non-invasive assessment of vessel diameter has proven notoriously elusive. A number of authors have used the spectral power of Doppler signals to show that MCA diameter does not

change during thigh cuff application and release [Aaslid 1989] and during hypocapnia and hypercapnia [Poulin 1996], but a recent study has cast doubt on the usefulness of this method of assessing vessel diameter [Deverson 2000]. Significant change in MCA diameter is unlikely, however, during thigh cuff application and release [Newell 1994], Valsalva manoeuvre [Tiecks 1996], LBNP [Serrador 2000] or supine rest. Nevertheless, MCA diameter was not directly measured and caution must therefore be exercised in interpreting the results.

In addition to inducing BP changes, the different static and dynamic stimuli may result in other important physiological changes such as hypocapnia secondary to hyperventilation, hypercapnia during the Valsalva manoeuvre due to apnoea and, possibly, varying degrees of increased sympathetic nervous system activity between stimuli. Hypocapnia has been shown to augment dynamic CA [Aaslid 1989], but no age-related differences in CO₂ levels were found at rest. The effect of increased systemic SNSA on dynamic CA is unclear (see section 1.2.2.4.), nor is it known if the effect of sympathetic nerves on cerebral blood vessels changes with ageing. An agerelated deterioration would, however, probably impair CA and result in lower ARI values in the older age group. As the degree of change in CO_2 and sympathetic nervous system activity, and the effects on CA, of thigh cuff application and release, Valsalva manoeuvre and LBNP application and release are unknown, the statistical methods used allowed for the fact that ARI values derived from different dynamic stimuli may not be directly comparable. In view of these uncertainties, methods of assessing dynamic CA using spontaneous BP changes at rest, using Aaslid's model or step response analysis, may appear more attractive.

3.7. Conclusions

1. Static and dynamic CA, as assessed using induced and spontaneous pressor and depressor changes, are unaffected by ageing.

2. No relationship exists between the magnitude of induced and spontaneous BP changes and indices of static and dynamic CA.

3. Dynamic ARI values derived using thigh cuff release are significantly greater than those derived using LBNP release.

4. Further studies using other methods are needed to precisely define the autoregulatory curve in older persons.

Figure 3.1. Mean step responses (solid lines) and standard errors (dotted lines) of young group (black lines) and older group (red lines)



Figure 3.1.

	Young group	Older group	Difference (95%CI)	p value
n	27	27	n/a	n/a
M:F ratio	15:12	15:12	n/a	n/a
Mean age (years)	29 ± 5 (20-39)	68 ± 5 (55-79)	n/a	n/a
Mean systolic BP (mmHg)	124 ± 11 (107-145)	130 ± 9 (107-147)	6 (5, 7)	<0.001
Mean diastolic BP (mmHg)	76 ± 8 (61-90)	76 ± 7 (58-89)	0 (-3, 3)	0.98
MAP (mmHg)	92 ± 8 (76-104)	94 ± 8 (74-105)	2 (-4, 1)	0.10
Baseline heart rate (bpm)	63.2 ± 10.2 (45.2-76.4)	59.6 ± 7.6 (42.7-81.1)	3.6 (-1.4, 8.5)	0.15
Baseline mean CBFV (cms ⁻¹)	54.3 ± 14.8 (37.4-81.5)	42.5 ± 9.9 (20.6-55.2)	11.8 (3.5, 18.6)	0.006
Mean BMI (kgm ⁻²)	23.3 ± 1.6 (19.8-28.3)	25.6 ± 3.8 (19.5-34.6)	2.3 (0.9, 4.0)	0.003
Baseline CO ₂ (mmHg)	41.4 ± 5.8 (35.4-48.9)	40.9 ± 4.4 (35.0-45.9)	0.5 (-3.4, 4.4)	0.80
Cardiac BRS (ms/mmHg)	16.9 ± 8.8 (7.0-44.0)	8.6 ± 4.5 (1.9-22.1)	8.4 (4.4, 12.3)	<0.0001

Table 3.1.Demographic information and baseline characteristics of the young and older groups. Values given are mean ± standard
deviation with ranges in parentheses for the group as a whole. Differences between groups are listed as mean differences
with 95% confidence intervals in parentheses.

B	BP stimulus	n	Younger BP change (mmHg)	Older BP change (mmHg)	Difference (95% CI)	p value	Younger mean ARI	Older mean ARI	Difference (95% CI)	p value
Static	Thigh cuff inflation	27	19±8	22 ± 9	3 (-3, 6)	0.44	57 ± 32%	55 ± 35%	2 (-13, 17)	0.80
	LBNP	14	-18 ± 6	-20 ± 7	-2 (-5, 2)	0.31	52 ± 29%	57 ± 31%	-5 (-15, 10)	0.48
Dynamic	Spontaneous pressor changes	27	9±3 *	8±2 *	1 (-0.5, 2.5)	0.16	4.4 ± 1.5	4.0 ± 2.3	0.4 (-0.9, 1.5)	0.34
	Spontaneous depressor changes	27	-9±3 *	-9±4 *	0 (-1.3, 2.1)	0.67	5.0 ± 1.9	4.7 ± 2.5	0.3 (-1.0, 1.4)	0.63
	Thigh cuff release	19	$-16 \pm 3 $ †	-16±6 †	0 (-1.3, 4.8)	0.91	5.4 ± 1.1	6.2 ± 2.4	-0.8 (-1.9, 1.5)	0.13
	LBNP release	11	17 ± 7	14 ± 3	3 (-0.9, 7.6)	0.11	4 .1 ± 2.0	4.5 ± 2.6	-0.4 (-2.3, 1.5)	0.62
	Valsalva manoeuvre (phase IV)	18	28 ± 14 ‡	$20 \pm 6 \ddagger$	8 (1.7, 13.9)	0.02	5.2 ± 2.6	5.2 ± 2.7	0 (-1.9, 2.0)	0.94

Table 3.2.Differences between groups in MAP changes and ARI values for each BP stimulus.
n = the number of matched pairs where data were available.
Comparisons were made using the Student's paired t-test.

* Represents the mean of all values > 5mmHg derived from the rest recordings.

† Represents the mean of all values > 10 mmHg derived from the thigh cuff recordings.

‡ Represents the mean of all values > 10 mmHg derived from the three Valsalva recordings.

CHAPTER FOUR

Baseline Cerebral Autoregulation in Patients With

Recurrent Vasovagal Syncope

4.1. Summary

Cerebral vasoconstriction has been reported prior to VVS, suggesting that CA may be impaired during pre-syncope. It is unclear, however, whether patients with recurrent VVS have abnormalities of CA at rest as well as during orthostatic stress.

17 patients with recurrent VVS and 17 normal control subjects were pair-matched for age, sex and systolic blood pressure (BP). Bilateral middle cerebral artery blood flow velocities were measured supine using transcranial Doppler ultrasound and BP measured by non-invasive beat-to-beat monitoring. Static and dynamic autoregulatory indices were calculated for each subject using spontaneous BP changes at rest and pressor and depressor BP changes induced by thigh-cuff inflation and release. Frequency domain analysis of rest recordings was also used to investigate dynamic CA.

No differences were demonstrated in static autoregulatory indices between the patient $(56\pm39\%)$ and control $(59\pm37\%)$ groups (p=0.7). Dynamic autoregulatory indices derived from mean values for both thigh-cuff release and spontaneous BP changes were similar for both patient (5.6 ± 1.7) and control (5.7 ± 1.5) groups (difference 0.2; 95% CI -0.6 to 0.4; p=0.56) and frequency domain analysis also failed to find any differences between groups.

Static and dynamic CA are preserved at rest in patients with recurrent VVS and cannot be used to predict the presence of recurrent VVS.

4.2. Background

Vasovagal syncope (VVS) is the commonest form of syncope encountered in medical practice [Kapoor 1990, 1994, Driscoll 1997]. The presence of VVS does not

necessarily imply a specific underlying cardiovascular pathology and can occur in normal individuals in a variety of circumstances [Van Lieshout 1991, Jorgensen 1993, el-Bedawi 1994a, Lightfoot 1995, Bondar 1995]. Prior to VVS, systolic blood pressure (BP), diastolic BP and total peripheral resistance decline in association with pre-syncopal symptoms such as dizziness, nausea, diaphoresis and diplopia until syncope and loss of consciousness occur [Van Lieshout 1991, Novak 1996, de Jong-Vos van Steenwijk 1995, Novak 1996, Hainsworth 1991, Fitzpatrick 1993] (see section 1.1.3.2.).

Dynamic CA refers to the ability to maintain cerebral blood flow in the face of BP changes occurring over a matter of seconds whereas static CA refers to cerebral blood flow adjustments in response to more prolonged BP changes (see section 1.3.). Irrespective of the pathophysiology involved in VVS, it is thought that the loss of consciousness during syncope is caused by cerebral hypoperfusion [Van Lieshout 1991, Sung 2000, Grubb 1991a, 1998, Fredman 1995, Giller 1992, Levine 1994, Schondorf 1997]. It is possible, therefore, that an impairment of CA may be contributing to the symptoms of VVS. In support of this hypothesis, a number of studies have used transcranial Doppler ultrasound with either head-up tilt (HUT) or incremental lower body negative pressure to suggest that paradoxical cerebral vasoconstriction may occur prior to VVS [Driscoll 1997, el-Bedawi 1994a, Sung 2000, Grubb 1991a, 1998, Fredman 1995, Giller 1992, Levine 1994]. More recently, however, others have suggested that cerebrovascular resistance (CVR) falls prior to syncope and that CA is, therefore, preserved in VVS [Schondorf 1997]. Transcranial Doppler ultrasound has facilitated the non-invasive measurement of cerebral blood flow velocity (CBFV) changes in response to static and dynamic BP changes, allowing indices of CA to be calculated [Tiecks 1995, Aaslid 1989] (see

section 2.4.). Methods of CA estimation from spontaneous BP and CBFV changes, e.g. frequency domain analysis [Panerai 1998a, 1999b, Zhang 1998a], may have advantages over methods using induced BP changes (see section 2.4.2.2.) as they do not induce sympathetic stimulation or changes in arterial CO₂, both of which may occur with other methods, such as the Valsalva manoeuvre, and may affect the integrity of CA. The use of spontaneous BP changes is also less noxious to subjects and poses less potential risk from the induction of acute hypotensive or hypertensive changes, particularly in patients with carotid artery stenosis and post acute ischaemic stroke.

As CA may be impaired during pre-syncope, induced and spontaneous static and dynamic changes in BP were used to investigate the hypothesis that abnormalities of CA detected during supine rest can predict recurrent episodes of VVS.

4.3. Objectives

1. To assess if differences exist during supine rest in static cerebral autoregulation between patients with recurrent VVS and controls.

2. To assess if differences exist during supine rest in dynamic cerebral autoregulation between patients with recurrent VVS and controls.

4.4. Methods

4.4.1. Subjects

17 patients with a history of recurrent VVS were recruited from a specialist outpatient syncope clinic (mean age = 48 ± 22 years; range 15-82). The diagnosis was made on the basis of reproduction of syncope in association with the symptoms of referral and the characteristic haemodynamic profile [Van Lieshout 1991] by a 70° HUT test of 30-minutes maximum duration without pharmacological provocation. Mean time to syncope was 816 ± 550 seconds (range 65-1790 seconds). In addition, each patient had other possible causes of syncope excluded by a combination of careful history and physical examination, full neurological assessment, standard laboratory tests, 12-lead surface ECG, supine and orthostatic BP measurements, bilateral supine and erect carotid sinus massage, 24-hour Holter recording, 2dimensional echocardiography and standard autonomic function tests [Ewing 1981]. Other cardiac and neurological investigations, such as electroencephalography, exercise stress testing, angiography, Doppler flowmetry of cervical vessels, electrophysiologic study and computed tomography or magnetic resonance imaging of the brain, were performed when clinically indicated. In addition, inclusion criteria included: at least 3 syncopal episodes in the previous 2 years; the last syncopal episode occurring within 6 months of recruitment; an interval of >6 months between the first and last episode.

The patient group was pair-matched for age, sex and systolic BP within 10 mmHg with 17 control subjects recruited from a volunteer register in the department and from departmental staff. Control subjects were healthy and had no previous history of syncope or pre-syncope (as assessed by history and physical examination, full neurological assessment and standard autonomic function tests) and had remained asymptomatic during a previous 30-minute 70° HUT test without pharmacological provocation. None of the patients or control subjects was taking any medication known to affect the cardiovascular system.

4.4.2. Study Protocol

All subjects underwent the initial preparatory protocol outlined in section 2.6.

The servo-adjust mechanism of the Finapres was then switched off and a baseline recording of 10 minutes was made with the subjects lying quietly. The servo-adjust mechanism of the Finapres was then switched on and bilateral thigh cuffs applied. The servo-adjust mechanism was again switched off and recording recommenced. After one minute, the thigh cuffs were inflated to 20mmHg above systolic BP for 90 seconds with thigh cuff pressure monitored using a mercury sphygmomanometer (see section 2.5.1). The cuffs were then rapidly deflated while recording continued for a further minute. After the recording, the Finapres servo-adjust mechanism was again switched off, the thigh cuffs reapplied, and the inflation/deflation cycle repeated in all subjects for a second time.

4.4.3. Data analysis

Data were initially analysed as outlined in section 2.7. In addition, ectopic beats occurring in the 15-second window prior to or after the point of thigh cuff release led to the rejection of the data.

4.4.4. Determination of Dynamic Cerebral Autoregulation using Aaslid's Method

Data were manually marked at the point of thigh cuff release and, in addition, four spontaneous dynamic pressor and four spontaneous dynamic depressor changes were marked using the 10-minute rest recording. Dynamic CA was graded by generating an autoregulatory index (ARI) ranging in value from 0 (absent) to 9 (most efficient) using the methods outlined in section 2.4.2.1. Only a dynamic BP fall \geq 10mmHg after thigh cuff release and spontaneous BP changes \geq 5mmHg at rest were accepted as adequate dynamic stimuli. Subjects also had to exhibit a synchronous and physiological change in BP and CBFV for all manual marks.

Each dynamic BP stimulus yielded a dynamic ARI for the right and left MCA of each individual. The mean of the two dynamic ARI values thus obtained was taken to be the ARI for that individual for that dynamic BP stimulus. The mean of the two dynamic ARI values thus derived from the two thigh cuff releases for each subject was taken as the dynamic ARI for that subject for thigh cuff release. Similarly, the mean of the four dynamic ARI values derived using spontaneous pressor and depressor changes was taken as the dynamic ARI for each subject for spontaneous pressor and depressor changes.

4.4.5. Determination of Static Cerebral Autoregulation

ARI values were derived for each subject for the static pressor response induced by thigh cuff inflation as described in section 2.4.1. Data were manually selected to include the one-minute baseline period before thigh cuff inflation and 90 second period of static mean arterial BP (MAP) rise after inflation. A BP rise ≥ 10 mmHg was considered to be an adequate static stimulus. Each static BP rise yielded an ARI for the right and left MCA of each subject. The mean of these two ARI values was taken to be the static ARI for each static pressor stimulus. The mean of the two static ARI values thus derived from the two thigh cuff inflations for each subject was taken as the static ARI for that subject.

4.4.6. Frequency Domain analysis

Frequency domain analysis was performed using the methods outlined in section 2.4.2.2.

4.4.7. Statistical analysis

As patients and controls were pair matched, demographic details and baseline characteristics of the two groups were compared using the Student's paired t-test. A normal distribution of static and dynamic ARI values was assumed and a good approximation to this was demonstrated by a plot of residuals. Dynamic ARI values were compared between the three dynamic BP stimuli and between the two groups using two-way analysis of variance. Static ARI values were compared between the two groups using the Student's paired t-test.

For frequency domain analysis, Student's paired-t tests were applied to individual spectral harmonics of coherence, amplitude and frequency response in the range 0.02 - 0.2 Hz, and also for the trough values of the impulse response, and at different times along the step response.

Data were analysed using Minitab 12 software package. Results are presented as mean \pm standard deviation (range). Statistical significance was set at p< 0.05 level.

4.5. Results

The demographic details and baseline charateristics of the two groups are contained in Table 4.1. No significant differences were demonstrated between the groups for age, body mass index, baseline mean CBFV, TCO_2 or casual systolic and diastolic blood pressure, though baseline heart rate tended to be slightly higher in the control group $(68 \pm 8 \text{ v} 63 \pm 10 \text{ beats per minute; p=0.06}).$

No significant differences were demonstrated between ARI values derived from the three dynamic BP changes (Table 4.2.), though ARI values derived from thigh cuff release tended to be higher than those derived from spontaneous pressor (difference = 0.9; 95% CI –0.1 to 1.8) and depressor (difference = 0.7; 95% CI –0.3 to 1.6) changes at rest (p=0.07). Overall, mean dynamic ARI values were 5.6 ± 1.7 for the patient group and 5.7 ± 1.5 for the control group, with no significant difference between the two groups (difference = -0.2; 95% CI –0.6 to 0.4; p=0.56). No differences were found between the groups for static ARI values or for dynamic ARI values derived from individual dynamic stimuli (Table 4.2.).

Mean values of coherence, amplitude and phase responses for the two groups were similar throughout the ranges studied, as shown in Figures 4.1., 4.2. and 4.3. respectively. The impulse and step responses of the two groups were also similar (see Figures 4.4. & 4.5.).

4.6. Discussion

Using both static and dynamic models of CA [Tiecks 1995, Panerai 1998a, 1999b, Zhang 1998a], as well as induced and spontaneous BP changes, no differences were demonstrated during supine rest between patients with stringently diagnosed recurrent VVS and pair-matched healthy controls. Using similar methods and volunteer numbers, others have previously detected impaired dynamic CA in patients with severe carotid artery stenosis [White 1997], after acute ischaemic stroke [Dawson 2000], in neonates [Panerai 1998b] and during hypercapnia [Panerai 1999b]. In addition, the confidence intervals for dynamic ARI values suggest that with this study size there are no clinically significant differences in dynamic CA between the two groups.

4.6.1. Cerebral autoregulation during pre-syncope

In view of the fact that the loss of consciousness during syncope is probably caused by cerebral hypoperfusion [Van Lieshout 1991, Sung 2000, Grubb 1991a, 1998, Fredman 1995, Giller 1992, Levine 1994, Schondorf 1997], the hypothesis that CA is impaired in patients with recurrent VVS, and that this impairment may be the primary problem in such patients, is an attractive one. Recent studies have appeared to support this hypothesis by demonstrating paradoxical increases in CVR during presyncope [Driscoll 1997, el-Bedawi 1994a, Sung 2000, Grubb 1991a, 1998, Fredman 1995, Giller 1992, Levine 1994]. More recently, however, others have suggested that CA may indeed be preserved prior to VVS [Schondorf 1997]. These results suggest that CA is normal at rest in patients with recurrent VVS and baseline CA measurements do not, therefore, predict the presence of VVS.

4.6.2. Autoregulatory Indices

Although Aaslid's method of calculating an ARI is an arbitrary model, it does give important clinical information, e.g. acute ischaemic stroke patients have dynamic ARI values in the order of 2.0 lower than age and BP matched controls [Dawson 2000]. Similar methods have shown dynamic ARI values to be a mean of 2.6 lower than control values in patients with carotid artery disease [White 1997]. The static and dynamic ARI values in this study are consistent with those published previously [Tiecks 1995, Dawson 2000, White 1997, Mahony 2000] and suggest that CA in patients with recurrent VVS is normal when supine.

4.6.3. Static & Dynamic Blood Pressure Stimuli

Thigh cuff release is probably the most widely applied dynamic BP stimulus, but it is unknown if ARI values are dependent on the type of dynamic stimulus used or the direction of BP change (pressor or depressor). It is also unclear whether induced BP changes alter physiological parameters that are known to affect CA, such as arterial CO_2 levels and systemic sympathetic nervous system activity, and whether these changes could potentially affect ARI values. In view of the uncertainties surrounding methods involving induced dynamic BP changes, both induced and spontaneous BP changes and two different methodologies of assessing dynamic CA were employed in this study but still failed to show any differences between patients and controls.

4.6.4. Frequency Domain Analysis

Frequency domain analysis has been used to demonstrate impairment of dynamic CA in premature neonates, hypercapnia and severe carotid artery stenosis [Panerai 1998a, 1998b, 1999b]. High coherence between arterial BP and CBFV suggests that BP changes are transmitted passively to the cerebral circulation, indicating impaired CA. On the other hand, active CA will attenuate CBFV changes in response to BP changes and coherence will fall. Reduced coherence at very low frequencies (<0.05 Hz) has been shown to be indicative of active dynamic CA [Panerai 1998a, 1999b, Zhang 1998a] but both patient and control groups demonstrated similar coherence values at all frequencies. In subjects with normal CA, oscillations in CBFV lead BP oscillations by a positive phase angle i.e. changes in CBFV occur before those in systemic BP [Panerai 1998a, 1999b, Zhang 1998a]. In subjects with impaired dynamic CA, the phase angle becomes less positive [Panerai 1998a, 1998b] but similar phase frequency responses were present in both groups. Amplitude frequency

responses were also similar in both groups, but Panerai [1998a, 1998b] observed that deterioration of CA is reflected more markedly in the phase rather than in the amplitude frequency response. Impulse response plots show an initial positive wave and subsequent negative deflection in CBFV when dynamic CA is active [Panerai 1998a, 1998b, 1999b, Zhang 1998a] with impairment leading to blunting or disappearance of the negative deflection [Panerai 1998a, 1998b, 1999b, Zhang 1998a]. Step response plots demonstrate a return of CBFV to baseline values when dynamic CA is intact and higher values when impaired [Panerai 1998a, 1998b, 1999b, Zhang 1998a]. The impulse response and step response plots were similar for patients and controls and indicate active autoregulation in both groups.

4.6.5. Limiting Factors

Only one pressor stimulus (thigh cuff inflation) was used to assess static CA and the confidence intervals are consequently not narrow enough to exclude clinically significant differences in static CA between the two groups. This is unlikely, however, as dynamic CA is probably more susceptible to impairment than static CA [Tiecks 1995, Dawson 2000] and this was normal in both groups. Nevertheless, the findings with regard to static CA need to be confirmed using other established methodologies.

This study did not attempt to assess CA changes during pre-syncope and, in view of apparently contradictory evidence in this regard [Driscoll 1997, el-Bedawi 1994a, Sung 2000, Grubb 1991a, 1998, Fredman 1995, Giller 1992, Levine 1994, Schondorf 1997], further investigation of CA during pre-syncope is warranted (see Chapters 5 and 6). Previous work in the area has relied on calculations of CVR to illustrate apparent cerebral arterial vasoconstriction or vasodilation but such indices may not

accurately reflect cerebrovascular changes [Schondorf 1997, Czosnyka 1996, Aaslid 1992]. It may be that the lower limit of CA occurs at a higher arterial BP in patients with recurrent VVS than normal controls (i.e. that the CA curve is shifted to the right) or that the autoregulatory plateau is narrowed in such patients, but the methodology of this study did not permit the testing of these hypotheses.

Although this study shows that both static and dynamic CA appear to be normal at rest in patients with recurrent VVS, the possibility that CA may deteriorate in patients with recurrent VVS during orthostasis was not addressed. Zhang [1998b] used lower body negative pressure and frequency domain analysis to suggest that CA deteriorates in normal subjects during orthostatic stress. Leftheriotis [1998], however, used thigh cuff release during HUT to suggest that dynamic CA is preserved during orthostasis, but the angle of tilt was only 40°. The hypothesis that orthostasis itself impairs CA in patients with recurrent VVS was not addressed by this study and this is an important area requiring further investigation.

The major limitation of this work, and indeed, any work using similar methodology, is that CBFV rather than CBF was measured. Changes in CBF can only be reliably deduced from velocity changes if the diameter of the insonated vessel remains unchanged [Aaslid 1982]. Significant changes in MCA diameter have not, however, been demonstrated during thigh cuff application and release [Newell 1994] or during supine rest. Nevertheless, MCA diameter was not directly measured in this study and caution must therefore be exercised in interpreting our results.

4.7. Conclusions

1. Static and dynamic CA, assessed using spontaneous and induced pressor and depressor BP changes, are normal during supine rest in patients with recurrent VVS.

Measurements of static or dynamic CA during supine rest cannot, therefore, be used to predict recurrent episodes of VVS.

Figure 4.1. Mean coherence values of the patient group (red line) and control group (black line). A representative sample standard error bar is included for the coherence values of the patient group at 0.06 Hz.



Figure 4.2. Mean values of phase frequency response of the patient group (red line) and control group (black line). A representative sample standard error bar is included for the phase values of the patient group at 0.06 Hz.



Figure 4.3. Mean values of amplitude frequency response of the patient group (red line) and control group (black line). A representative sample standard error bar is included for the amplitude values of the patient group at 0.06 Hz.



Figure 4.4. Mean impulse responses (solid lines) and standard errors (dotted lines) of the patient group (red lines) and control group (black lines).



Figure 4.5. Mean step responses (solid lines) and standard errors (dotted lines) of the patient group (red lines) and control group (black lines).



Parameter	Patients	Controls	Difference (95% CI)	p value
Age	48 ± 22	49 ± 20	-1	0.46
(years)	(15-82)	(19-90)	(-5.38, 2.55)	
Systolic BP	131 ± 21	129 ± 17	2	0.68
(mmHg)	(98-179)	(105-163)	(-9, 14)	
Diastolic BP	76 ± 11	79 ± 10	-3	0.18
(mmHg)	(60-98)	(63-103)	(-8, 3)	
Heart rate (bpm)	63 ± 10 (46-78)	68 ± 8 (52-82)	-5 (-10, 1)	0.06
Mean CBFV	53 ± 14	59 ± 12	-6	0.13
(cms ⁻¹)	(34-83)	(41-83)	(-14, 2)	
Transcutaneous CO ₂ (mmHg)	41.1 ± 4.7 (29.1-47.0)	41.8 ± 3.9 (34.5-47.9)	-0.7 (-3.8, 2.4)	0.61
BMI	22.0 ± 2.8	23.8 ± 3.7	-1.8	0.13
(kg/m ⁻²)	(16.9-27.8)	(16.8-30.3)	(-4.3, 0.6)	

Table 4.1.Demographic and baseline characteristics of the patient and controlgroups compared using Student's paired t-tests.

Values are displayed as mean \pm standard deviation with ranges in parentheses.

BP stimulus		ARI	Difference		
	Patients & Controls	Patients	Controls	(95% CI)	p value
Thigh cuff release (dynamic)	6.2 ± 1.4	6.1 ± 1.4	6.3 ± 1.5	-0.2 (-1.1, 0.7)	0.565
Rest pressor changes (dynamic)	5.3 ± 1.7	5.0 ± 1.8	5.5 ± 1.6	-0.5 (-1.7, 0.7)	0.396
Rest depressor changes (dynamic)	5.5 ± 1.7	5.6 ± 1.9	5.4 ± 1.4	0.2 (-1.0, 1.3)	0.738
Thigh cuff inflation (static)	N/A	56 ± 39 %	59 ± 37 %	-3 % (-18, 12)	0.702

Table 4.2.Dynamic and static ARI values of patient and control groups.

Values displayed are mean \pm standard deviation. Differences shown are for the values between patient and control groups.

CHAPTER FIVE

Cerebral Autoregulatory Responses To Head-Up Tilt

in Normal Subjects & Patients With Recurrent

Vasovagal Syncope

5.1. Summary

In chapter four, it was recorded that no abnormalities were detected in static or dynamic cerebral autoregulation (CA) in patients with recurrent vasovagal syncope (VVS) during supine rest. The effect of orthostatic stress, however, on dynamic CA in patients with recurrent VVS and normal subjects is unclear and in this chapter the dynamic CA responses of both groups to head-up tilt (HUT) are studied. 17 patients with recurrent VVS and 17 pair-matched control subjects underwent 70° HUT for up to 30 minutes. Bilateral middle cerebral artery blood flow velocities (CBFV) were measured using transcranial Doppler ultrasound along with noninvasive beat-to-beat blood pressure (BP), heart rate (HR) and transcutaneous (TCO₂) and end-tidal (ETCO₂) carbon dioxide concentrations. Indices of dynamic CA were derived for periods before, during and after HUT. Eight normal subjects with no previous clinical history of syncope but who developed VVS using an identical protocol were also studied.

CBFV, TCO₂ and ETCO₂ levels declined significantly after HUT in all groups (p<0.0001). Dynamic CA indices were unchanged throughout HUT in non-syncopal control subjects and were initially unchanged in patients, but deteriorated significantly in patients and syncopal controls in the sixty seconds before (p=0.027 and p=0.012 respectively) and sixty seconds after (p=0.002 and p=0.007 respectively) returning supine following symptom onset. In non-syncopal controls, changes in TCO₂ after HUT were significantly correlated with changes in mean CFBV (p=0.011) and inversely correlated with indices of dynamic CA (p=0.027).

Dynamic CA, therefore, is preserved in patients and control subjects initially after HUT. Cerebral autoregulatory function remains intact in non-syncopal control subjects during prolonged orthostasis, but deteriorates in patients and syncopal controls immediately before and after VVS. An inverse relationship exists between TCO_2 changes and indices of dynamic CA after HUT in normal subjects, suggesting that orthostasis-induced hypocapnia may help to preserve dynamic CA.

5.2. Background

Hypotension prior to VVS is probably precipitated by sympathetic nervous system withdrawal and increased vagal activity [Mosqueda-Garcia 1997, Morillo 1997, Lewis 1932] but loss of consciousness during syncope is probably caused by cerebral hypoperfusion [Van Lieshout 1991, Bondar 1995, Sung 2000, Grubb 1991a, Giller 1992, Levine 1994, Schondorf 1997], suggesting that an impairment of CA may contribute to the symptoms of VVS. In chapter four, however, no abnormalities were detected during supine rest in static or dynamic CA in patients with recurrent VVS. CBF velocity (CBFV) has been shown to fall significantly during orthostatic stress with HUT and lower body negative pressure (LBNP) in normal subjects [Bondar 1995, Sung 2000, Grubb 1991a, Giller 1992, Levine 1994, Schondorf 1997, Cencetti 1997]. Arterial, end-tidal and transcutaneous carbon dioxide (CO_2) levels have also been shown to fall significantly in normal subjects immediately following HUT [Cencetti 1997, Serrador 1998, Anthonisen 1965, Bjurstedt 1962, Yoshizaki 1998] and Cencetti [1997] has shown a significant link between the declines in CBFV and CO₂ after HUT. Hypocapnia reduces CBF, improves dynamic CA [Panerai 1999b] and widens the plateau region of the static pressure autoregulation curve, representing an improvement in autoregulatory capacity [Paulson 1990] (see section 1.3.3.1.). Despite the fall in CBFV during orthostatic stress, its effect on dynamic CA is unclear with both intact [Leftheriotis 1998, Lipsitz 2000] and impaired [Zhang 1998b] CA being reported in normal subjects. Leftheriotis [1998] demonstrated preserved

dynamic CA in normal subjects in response to rapid thigh cuff release during 40° head-up tilt. This angle of tilt exerts less orthostatic stress, however, than the angles of 60-90° in more widespread usage. Zhang [1998b] employed lower body negative pressure (LBNP) and an increased low-frequency transfer function gain to demonstrate deterioration in dynamic CA during high levels of orthostatic stress in normal subjects. The issue is complicated by the facts that there may be differences in cerebrovascular and cardiovascular responses between HUT and LBNP [Bondar 1995] and that the significance of changes in transfer function gain are a matter of some debate [Panerai 1998a]. More recently, Lipsitz [2000] looked at standing changes from sitting in elderly controls and previously treated hypertensives and found similar responses to those found in young, healthy controls.

This study investigated the dynamic CA responses to HUT-induced orthostatic stress in normal subjects and patients with recurrent VVS, hypothesising that dynamic CA would remain intact during orthostatic stress in normal subjects but would deteriorate in patients with recurrent VVS.

5.3. Objectives

1. To assess the dynamic cerebral autoregulatory responses of normal subjects and patients with recurrent VVS to HUT.

2. To assess the relationship between dynamic CA changes and physiological CO₂ changes in normal subjects after HUT.

5.4. Methods

5.4.1. Subjects

17 patients with a history of recurrent VVS according to the same inclusion and exclusion criteria outlined in section 4.4.1. were recruited from an out-patient specialist Syncope Clinic.

The patient group was pair-matched for age, sex and systolic BP within 10 mmHg with 17 control subjects (non-syncopal control group) recruited from a volunteer register in the department and from departmental staff. Control subjects were healthy (as assessed by history and physical examination, full neurological assessment and standard autonomic function tests), had no previous history of syncope or pre-syncope and had remained asymptomatic during a previous 30 minute 70° HUT. In addition, we studied 8 subjects who were otherwise similar to the controls but developed VVS (mean time to syncope = 1102 ± 612 seconds; range 185-1800 seconds) during HUT using an identical protocol to that outlined below (syncopal control group). None of the patients or control subjects was taking any medication known to affect the cardiovascular system.

5.4.2. Study Protocol

All subjects underwent the initial preparatory protocol outlined in section 2.6. In addition, end-tidal CO_2 was measured via a closely fitting face-mask and an infra-red capnograph (Capnogard, Novametrix) as described in section 2.3.1. (Figure 2.7). The vertical height in centimetres (Ht) from the point of insonation of the right MCA to the second intercostal space was also recorded for each subject.

After resting supine for a minimum of 30 minutes to obtain stable values (<10% variation over 5 minutes), a 5-minute baseline recording was made. The subjects then underwent HUT within three seconds to an angle of 70° for 30 minutes or until syncope was imminent. The Finapres cuff was kept at heart level at all times before,

during and after HUT with the aid of an adjustable arm-rest. In order to minimise discomfort and improve compliance with the study protocol, $ETCO_2$ measurements were discontinued 5 minutes after tilt in all subjects. The imminence of syncope was recognised by the occurrence of a subjective sensation of impending syncope in association with the typical hemodynamic profile [Van Lieshout 1991]. For ethical reasons, all pre-syncopal subjects were replaced supine prior to loss of consciousness, the point at which subjects were replaced supine being taken as the point of syncope. The point of syncope was synchronised for all subjects using a mark generated by an electrical device each time the tilt table passed through 45°. Recording continued for a further 5 minutes after returning supine.

5.4.3. Data Analysis

Data were initially analysed as outlined in section 2.7. with estimates of systolic, diastolic and mean arterial BP, systolic, diastolic and mean CBFV, pulse interval (PI), TCO_2 and $ETCO_2$ being made for each cardiac cycle.

Using methods described in section 2.4.2.1, an estimation of a dynamic autoregulatory index (ARI) ranging from 0 (absent) to 9 (most efficient) was made for the 1-minute period before tilt, first and third minutes after tilt, third last and last minutes before returning supine and first and third minutes after returning supine for each subject. MCA mean pressure (MCAMP) during HUT was estimated from MAP by subtracting the hydrostatic pressure (Ht * 0.735 * sin 70°). Detailed averaged profiles of all parameters were calculated for both groups for the 1-minute period before and 3minute period after HUT. The model allowed calculation of correlation coefficients assessing how closely measured CBFV fitted the model predicted velocity [Panerai 1999a] for each subject during each time period.

5.4.4 Statistical analysis

Demographic details and baseline characteristics of the three groups were compared using one-way analysis of variance. The between group and within subject changes of the patient and non-syncopal control groups were modeled for each outcome measure using a mixed model for repeated measures data. Model selection was by changes in the log likelihood and denominator degrees of freedom were calculated using Satterthwaite's method [Satterthwaite 1946]. Different covariance patterns were investigated using Akaike's Information Criterion [Akaike 1973]. To investigate the relationship between TCO₂ changes and ARI and mean CBFV changes after HUT in non-syncopal control subjects, changes in TCO₂ after HUT were treated as a covariate in a repeated measures model using CBFV and ARI values as the dependent variables. ARI values immediately before and after syncope were compared between the syncopal control group and other groups using Student's 2 sample t-tests. Data were analysed using SAS version 6.12 and Minitab 12 software packages. Statistical significance was set at p<0.05 level.

5.5. Results

No significant differences were demonstrated between the three groups in demographic details and baseline characteristics (Table 5.1.). All 17 patients developed syncope within 30 minutes of HUT (mean time to syncope = 816 ± 550 s; range 65-1790 s). All 17 control subjects remained asymptomatic during the 30 minutes of HUT. Actual data records for a non-syncopal control subject before and after tilting and a patient before and after syncope are displayed in Figures 5.1. and 5.2. respectively.

Changes in MCA mean CBFV from baseline for the patient and non-syncopal control groups are contained in Table 5.2, changes in MCAMP and MAP in Table 5.3, changes in pulse interval in Table 5.4. and changes in TCO₂ and ETCO₂ in Table 5.5. Mean MCAMP and mean CBFV changes for the 1-minute period before HUT and 3-minute period after HUT for both patient and non-syncopal control groups are displayed in Figure 5.3. Mean ETCO₂ and TCO₂ changes for the same period are displayed in Figure 5.4.

Mean CBFV was significantly lower in patients than non-syncopal controls 3 minutes before syncope (difference=8.9 cm/s; 95% CI 0 to 18; p=0.044) and at syncope (difference=24.6 cm/s; 95% CI 16 to 33; p<0.0001) but was similar at all other times (see Table 5.2.). MCAMP was lower in patients than non-syncopal controls at syncope (difference=52.5 mmHg; 95%CI 45 to 60; p<0.0001) but was similar at all other times (see table 5.3.). No differences were demonstrated between patients and non-syncopal controls in TCO₂ (p=0.31) and ETCO₂ (p=0.10) values at any stage.

Mean dynamic ARI values of the patient and non-syncopal control groups for the 7 chosen time points are contained in Table 5.6. ARI values were similar to pre-HUT values both during and after HUT in non-syncopal control subjects (Table 5.6.). Patient dynamic ARI values were similar to non-syncopal control values at baseline and initially after HUT but were significantly lower during the last minute before syncope and the first minute after syncope (Table 5.6.).
ARI values of the 8 control subjects who developed VVS were similar to patient values during the last minute before $(3.1 \pm 2.2 \text{ v} 3.6 \pm 3.0; \text{ difference} = -0.5; 95\% \text{ CI} - 1.3 \text{ to } 0.3; p=0.25)$ and first minute after $(2.1 \pm 1.7 \text{ v} 2.3 \pm 1.8; \text{ difference} = -0.2; 95\%$ CI –1.0 to 0.6; p=0.60) syncope and significantly lower than non-syncopal control values during the same periods (p=0.012 and p=0.007 respectively). Correlation coefficients reflecting model accuracy did not differ between groups and were similar to baseline coefficients pre- and post-syncope (Table 5.6.). A significant relationship was demonstrated between TCO₂ changes and mean CBFV changes (p=0.011; r² = 0.46) and a significant inverse relationship was shown between TCO₂ changes and ARI values (p=0.027; r² = 0.12) after HUT in non-syncopal control subjects.

5.6. Discussion

This study has not demonstrated any deterioration in dynamic CA during orthostatic stress in normal subjects and has shown that dynamic CA is initially preserved after HUT in patients susceptible to VVS. Dynamic CA has been shown, however, to deteriorate in patients and control subjects during pre-syncope and to remain impaired during the immediate post-syncopal period. In addition, the results suggest that an inverse relationship exists between CO₂ changes and changes in indices of dynamic CA after HUT in normal subjects, a finding consistent with the conclusions of others that hypocapnia physiologically improves autoregulatory function [Paulson 1990, Panerai 1999b]. Baseline ARI values in this study are comparable to values previously derived using similar methods [Panerai 1999a, 2001] and suggest that baseline dynamic CA is normal in all groups studied. As well as these new findings, this study confirms that CBFV and CO₂ levels decline significantly in normal subjects after passive HUT [Bondar 1995, Sung 2000, Grubb 1991a, Giller 1992, Levine 1994, Schondorf 1997, Cencetti 1997, Serrador 1998, Anthonisen 1965, Bjurstedt 1962, Yoshizaki 1998] and reveals that similar changes occur initially after HUT in patients with recurrent VVS. Finally, the results also appear to confirm the significant relationship between CO₂ and CBFV changes after HUT in normal subjects first demonstrated by Cencetti [1997].

5.6.1. Dynamic cerebral autoregulation during orthostatic stress

Zhang [1998b] used frequency domain analysis and LBNP to demonstrate that dynamic CA may deteriorate in normal subjects during high levels of orthostatic stress. An increased low-frequency CBFV/arterial BP transfer function gain was shown during pre-syncope, suggesting a closer relationship between CBFV and ABP and, therefore, impairment of dynamic CA. There may be important differences, however, between HUT and LBNP in cardiovascular and cerebrovascular responses [Bondar 1995] and the significance of changes in transfer function gain are the subject of some debate [Panerai 1998b]. For the latter reason, dynamic CA was assessed in this study employing ARI values derived from Aaslid's model rather than using frequency domain analysis.

More recently, Leftheriotis [1998] used rapid thigh cuff deflation to demonstrate preserved dynamic CA in normal subjects 5 minutes after 40° head-up tilt. In many ways, this study forms a link between the studies of Zhang [1998b] and Leftheriotis [1998] and shows that the findings of all three studies are compatible with each other. The finding of Zhang [1998b] that dynamic CA deteriorates in normal subjects during pre-syncope is supported by our demonstration of decreased ARI values during presyncope in the syncopal control group. In addition, we have demonstrated that similar changes occur in patients with recurrent VVS. The conclusion of Leftheriotis [1998] that dynamic CA is preserved in normal subjects at low levels of orthostatic stress is similarly supported by our initial data after HUT, which also demonstrate similar preservation of dynamic CA in patients with recurrent VVS.

5.6.2. Dynamic cerebral autoregulation and carbon dioxide changes after HUT

This work appears to confirm the findings of Cencetti [1997] that CBFV changes in normal subjects after HUT are significantly linked to physiological hypocapnia. Hypocapnia has previously been suggested as helping to facilitate dynamic CA [Paulson 1990, Panerai 1999b] and the suggestion of a significant inverse link between changes in TCO₂ and dynamic ARI values supports this hypothesis. It is possible, therefore, that the preservation of dynamic ARI values after HUT may be facilitated by the physiologically lower CO₂ levels seen during orthostasis but this warrants further investigation as the r^2 value reported (0.12) is unimpressive.

5.6.3. Dynamic cerebral autoregulation during pre-syncope

As loss of consciousness during syncope is probably caused by cerebral hypoperfusion [Van Lieshout 1991, Bondar 1995, Sung 2000, Grubb 1991a, Giller 1992, Levine 1994, Schondorf 1997], the hypothesis that impaired CA is the underlying problem in patients with recurrent VVS is an attractive one. The findings of preserved ARI values in patients during supine rest (see Chapter 4) and initially after HUT and similarly impaired ARI values in normal subjects immediately before and after syncope tend to refute this hypothesis. The reason for impairment of dynamic ARI values during the peri-syncopal period is unclear, but is most likely to result from MCAMP falling outside the proposed autoregulatory range of 60-150mmHg [Paulson 1990] during this period (Table 5.3.). Another potential reason for impairment of dynamic CA before and after syncope is a build up of cerebral metabolites (including CO₂) during a period of relative hypoxia, but our methods did not allow us to explore this hypothesis.

5.6.4. Model validity and accuracy

The model proposed by Aaslid and Tiecks [Tiecks 1995] was initially developed using hypotension induced by rapid thigh cuff deflation, but has been previously applied to spontaneous BP changes at rest [Panerai 1999a, 2001] and shown to be valid. Correlation coefficients between measured and model predicted CBFV were similar to previously reported coefficients [Panerai 1999a] and did not differ immediately before or after syncope, suggesting that the model is accurate and valid at low extremes of CBFV and MCAMP.

5.6.5. Limiting factors

The major limiting factor of this work is the indirect measure of MCAMP. Although arterial pressure waveforms change with head-up tilt, non-invasive plethysmography correlates very well with intra-arterial pressure recordings during head-up tilt [Petersen 1995]. A direct measurement of MCA pressure is impossible without very invasive procedures which would, in themselves, affect the interpretation of such work. As original work with the model [Panerai 1999a, 2001] used MCA pressure derived using Finapres monitoring as the input parameter and assumed that fluctuations in perfusion pressure were reflected, in the main, by MCA pressure fluctuations, it is important that MCA pressure is used when calculating ARI values.

Intracranial pressure changes after HUT are likely to be relatively small [Drummond 2000] and similar in syncopal and non-syncopal subjects alike, and changes in venous pressure will occur to an equal and proportionate degree as arterial pressure. In the absence of better, non-invasive alternatives, therefore, calculations using non-invasive plethysmography provide acceptable estimates of MCA pressure.

Our calculations also assume that MCA diameter remains constant during HUT and in the pre-syncopal period. MCA calibre does not change during simulated orthostatic stress [Serrador 2000] and changes in CO₂ concentrations to the degree demonstrated by us would not be expected to significantly affect MCA diameter [Giller 1993]. Profound hypotension during pre-syncope, however, could potentially cause MCA myogenic vasodilation and influence CBFV, but the methodology employed did not allow assessment of this possibility. As MCA diameter was not directly measured during this study, caution must, therefore, be exercised when interpreting the results. TCO₂ measurements correlate highly with arterial CO₂ levels but rely on gas diffusion and, therefore, have poor dynamic response characteristics (section 2.3.2.). In addition, doubts exist about the accuracy of end-tidal and transcutaneous CO_2 measurements during changes in cardiac output [Steurer 1997]. The TCO₂ measurements reported in this study, however, were taken over stable one-minute periods and are consistent with the findings of others [Cencetti 1997, Anthonisen 1965, Bjurstedt 1962, Yoshizaki 1998], suggesting that they are a fair reflection of arterial CO₂ levels.

5.7. Conclusions

1. Dynamic CA is preserved initially after HUT in normal subjects and patients with recurrent VVS with identical cardiovascular, cerebrovascular and carbon dioxide changes demonstrated in both groups.

2. A significant inverse relationship exists between CO_2 changes and dynamic ARI changes after HUT in normal subjects consistent with the hypothesis that hypocapnia facilitates dynamic CA. This inverse relationship suggests a potential mechanism whereby dynamic CA may be preserved during orthostatic stress.

3. Dynamic CA deteriorates during pre-syncope and remains impaired during the early post-syncopal period in both patients and syncopal control subjects. The reasons for impairment of dynamic CA during and after pre-syncope are unclear, however, and require further investigation.

Figure 5.1.

Data record of a non-syncopal control subject around the point of HUT at time 0.



Figure 5.2. Data record of a patient around the point of syncope at time 0.



Figure 5.2.

Figure 5.3.

Mean MCAMP (solid lines) and mean CBFV (dotted lines) changes for the patient group (red lines) and control group (black lines) for the 1-minute period before and 3-minute period after HUT at time 0.



Figure 5.4.

Mean TCO2 (solid lines) and ETCO2 (dotted lines) changes for the patient group (red lines) and control group (black lines) for the 1-minute period before and 3-minute period after HUT at time 0.



Parameter	Patients	Non-Syncopal Controls	Syncopal Controls	p value
n	17	17	8	N/A
Age	48 ± 22	49 ± 20	55 ± 22	0.63
(years)	(15-82)	(19-90)	(21-87)	
Systolic BP	128 ± 21	127 ± 17	129 ± 19	0.96
(mmHg)	(98-155)	(105-152)	(102-158)	
Diastolic BP	75 ± 11	78 ± 10	81 ± 11	0.42
(mmHg)	(60-89)	(63-90)	(62-91)	
Heart rate	63 ± 10	68 ± 8	66 ± 7	0.23
(beats per minute)	(46-78)	(52-82)	(56-81)	
Mean CBFV	57 ± 16	61 ± 16	60 ± 17	0.37
(cms ⁻¹)	(34-83)	(41-83)	(45-79)	
Transcutaneous CO ₂	41.1 ± 4.7	41.8 ± 3.9	41.5 ± 4.8	0.80
(mmHg)	(29.1-47.0)	(34.5-47.9)	(36.6-46.1)	
Height	1.71 ± 0.09	1.69 ± 0.08	1.69 ± 0.07	0.83
(m)	(1.50 – 1.90)	(1.54 – 1.82)	(1.57-1.80)	
Body mass index (kg/m ⁻²)	22.0 ± 2.8 (16.9-27.8)	23.8 ± 3.7 (16.8-30.3)	23.9 ± 2.7 (19.8-28.4)	0.12

Table 5.1. Demographic and baseline characteristics of the patient group, non-
syncopal control group and syncopal control groups. Values are displayed as mean \pm
standard deviation with ranges in parentheses. No differences were demonstrated
between the 3 groups for any baseline parameter.

	Time*	Α	B	С	D	E	F	G
Mean	Controls	59±15	50±12	51±11	49±11	54±15	55±15	57±15
		Diff	-9	-9	-10	-5	-4.8	-3
Moon	Controls	95% CI	-5, -14	-5, -13	-5, -16	-12, 1	-10, 1	-8, 2
		p value	< 0.0001	< 0.0001	0.0008	0.11	0.063	0.26
(cm/s)		56±16	44±12	47±11	41±13	29±9	49±15	51±15
((011/3))	Datients	Diff	-12	-9	-15	-26	-7	-5
	rationts	95% CI	-7, -16	-4, -13	-10, -21	-20, -33	-2, -12	-10, 1
		p value	< 0.0001	0.0002	< 0.0001	< 0.0001	0.0064	0.069

Table 5.2.Mean CBFV of the patient and non-syncopal control groups before,
during and after HUT. Differences cited are from baseline. Values are displayed as
mean \pm SD.

	Time*	Α	В	С	D	E	F	G
		89±16	72±14	73±16	69±14	68±12	90±14	91±15
	Controls	Diff	-17	-17	-20	-22	0.2	2
MCAMP	Controls	95% CI	-12, -23	-12, -22	-13, -27	-16, -27	-5, 5	-4, 7
		p value	< 0.0001	< 0.0001	< 0.0001	< 0.0001	0.94	0.58
(mmHg)		85±15	63±20	65±20	60±17	15±10	86±15	91±18
	Patients	Diff	-22	-20	-25	-69	1	-7
		95% CI	-16, -27	-15, -25	-17, -32	-64, -75	-4, 6	1, 12
		p value	< 0.0001	< 0.0001	< 0.0001	< 0.0001	0.58	0.027
	Controls	89±16	91±14	92±16	88±14	87±12	90±14	91±15
		Diff	2	3	-1	-2	1	2
		95% CI	-4, 7	-2, 8	-8, 6	-7, 4	-5, 5	-4, 7
MAP		p value	0.57	0.38	0.91	0.69	0.94	0.58
(mmHg)		85±15	82±20	84±20	79±17	34±10	86±15	91±18
	Datients	Diff	-3	-1	-6	-51	1	-7
	Fatients	95% CI	-8, 3	-6, 4	-13, 2	-46, -57	-4, 6	1, 12
MCAMP (mmHg) MAP (mmHg)		p value	0.32	0.89	0.096	< 0.0001	0.58	0.027

Table 5.3.MCAMP and MAP of the patient and non-syncopal control groupsbefore, during and after HUT.Differences cited are from baseline.Values aredisplayed as mean \pm SD.

Α	=	1 minute before HUT (baseline)
В	=	1 minute after HUT
С	=	3 minutes after HUT
D	=	3 minutes prior to being returning supine
E	=	1 minute prior to being returned supine
F	=	1 minute after being returned supine
G	=	3 minutes after being returned supine

*

	Time*	Α	В	С	D	E	F	G
Dulas	Controls	898±134	755±93	753±102	701±104	721±117	882±98	929±151
		Diff	143	145	198	177	16	-31
		95% CI	80, 206	102, 189	151, 244	130, 224 -35, 67		-90, 29
interval		p value	<0.0001	< 0.0001	<0.0001	< 0.0001	0.515	0.292
(ms)	Datianta	1007±192	866±196	788±197	729±190	975±756	1022±206	1060±273
		Diff	141	219	278	32	-16	-54
		95% CI	66, 217	136, 301	193, 363	-295, 359	-52, 20	-135, 26
		p value	0.001	< 0.0001	< 0.0001	0.838	0.363	0.188

Table 5.4.Pulse interval changes of the patient and non-syncopal control groups
before, during and after HUT. Differences cited are from baseline. Values are
displayed as mean \pm SD.

	Time*	Α	В	С	D	E	F	G
Trops	Controls	42.0±6.2	41.0±6.3	39.5±6.6	40.0±5.8	39.9±5.7	40.7±5.7	40.9±5.6
I rans-	Patients	41.3±6.3	40.3±6.6	38.7±6.2	38.6±6.2	37.5±6.2	37.8±6.0	38.5±5.5
CO	Patients	Diff	-1.0	-2.5	-2.3	-2.9	-2.4	-2.0
(mmHg)	&	95%CI	-0.6, -1.4	-1.9, -3.2	-1.2, -3.5	-1.7, -4.2	-1.2, -3.6	-0.9, -3.1
(IIIIIIIg)	Controls	p value	< 0.0001	< 0.0001	0.0002	< 0.0001	0.0002	0.001
	Controls	43.2±5.3	39.2±7.2	39.6±6.1				
End-tidal CO ₂	Patients	39.4±6.7	36.7±5.6	36.1±7.6				
	Patients	Diff	-3.4	-3.5				
(mmHg)	&	95% CI	-1.9, 4.9	-1.5, -5.5				
	Controls	p value	< 0.0001	0.0014				

Table 5.5.Transcutaneous and end-tidal CO_2 changes of the patient and non-
syncopal control groups before, during and after HUT. Differences cited are from
baseline. Values are displayed as mean \pm SD.

А	=	1 minute before HUT (baseline)
В	=	1 minute after HUT
С	=	3 minutes after HUT
D	=	3 minutes prior to being returning supine
E	=	1 minute prior to being returned supine
F	=	1 minute after being returned supine
G	=	3 minutes after being returned supine

*

		Patients (1		Non-Syncopal Controls (n=17)				Controls v Patients		
*Time	ARI	Difference	р	Correlation	ARI	Difference	р	Correlation	Difference	р
Period		from A	value	Coefficient		from A	value	Coefficient		value
		(95% CI)				(95% CI)				
A	4.4 ± 2.7	-	-	0.69 ± 0.16	4.2 ± 2.7	-	-	0.63 ± 0.16	-0.3 (-1.9, 1.4)	0.75
В	5.5 ± 2.6	1.1 (-0.5, 2.6)	0.17	0.62 ± 0.21	4.9 ± 2.9	0.7 (-0.8, 2.2)	0.36	0.62 ± 0.19	-0.6 (-2.3, 1.0)	0.46
C	6.2 ± 1.4	1.7 (0.2, 3.2)	0.025	0.73 ± 0.16	5.4 ± 2.6	1.2 (-0.3, 2.7)	0.12	0.74 ± 0.19	-0.8 (-2.4, 0.9)	0.35
D	5.4 ± 2.5	0.9 (-0.6, 2.5,)	0.22	0.68 ± 0.16	5.5 ± 2.3	1.3 (-0.2, 2.8)	0.096	0.68 ± 0.13	0.1 (-1.6, 1.7)	0.92
E	3.6 ± 3.0	0.9 (-0.6, 2.4)	0.24	0.69 ± 0.18	5.4 ± 2.0	1.2 (-0.3, 2.7)	0.12	0.69 ± 0.14	1.9 (0.2, 3.5)	0.027
F	2.3 ± 1.8	-2.1 (-0.6, -3.6)	0.0066	0.74 ± 0.19	4.9 ± 2.3	0.8 (-0.8, 1.8)	0.33	0.62 ± 0.21	2.6 (1.0, 4.2)	0.0021
G	5.3 ± 2.6	0.9 (-0.7, 2.4)	0.26	0.64 ± 0.22	5.4 ± 1.8	-1.2 (-0.3, 2.7)	0.12	0.66 ± 0.12	0.1 (1.6, 1.7)	0.94

Table 5.6.Comparison of dynamic ARI values between different time periods for both patient and non-syncopal control subjects.
Correlation coefficients are given for how closely measured CBFV fitted the model predicted CBFV for each time period.
Values are given as mean ± standard deviation.

- A = last minute before HUT (baseline)
 - B = first minute after HUT

*

F

- C = third minute after HUT
- D = third last minute of tilt prior to being returning supine
- E = last minute of tilt prior to being returned supine
 - = first minute after being returned supine
- G = third minute after being returned supine

CHAPTER SIX

Carbon Dioxide, Critical Closing Pressure & Cerebral

Haemodynamics During Pre-Syncope

6.1. Summary

In the previous chapter, indices of dynamic CA were shown to deteriorate during presyncope in normal subjects and patients with recurrent VVS. The reason for such deterioration is unclear but may be due to MCA mean BP falling outside the reported autoregulatory range of 60 - 150 mmHg [Paulson 1990]. This study, therefore, looked more closely at the changes in a number of cerebrovascular parameters during pre-syncope, and their potential underlying mechanisms, to help explain the deterioration in dynamic CA.

65 normal subjects with no previous history of syncope and 16 patients with recurrent VVS were subjected to 70° HUT for 30 minutes or until syncope was imminent. Bilateral MCA CBFV was measured using TCD ultrasound along with simultaneous measures of MCA BP, heart rate, ETCO₂ and TCO₂. All 16 patients and 14 of the control subjects developed VVS during HUT with similar pre-syncopal changes in both groups. During pre-syncope, mean CBFV declined predominantly due to a decrease in diastolic rather than systolic CBFV. CO₂ levels and indices of CVR decreased during pre-syncope while CrCP increased to levels approaching MCA diastolic BP before decreasing precipitously on syncope.

The rise in CrCP, possibly due to progressive hypocapnia, provides an explanation for the selective impairment of diastolic CBFV during pre-syncope. Rises in CrCP, therefore, may counteract the autoregulatory falls in CVR leading to impaired CA during pre-syncope. The fact that similar pre-syncopal cerebrovascular changes occur in normal subjects and patients with recurrent VVS suggests that impaired CA is unlikely to be the primary problem in patients.

6.2. Background

Irrespective of the pathophysiology involved in VVS (see section 1.1.3.), it is thought that loss of consciousness during syncope is caused by cerebral hypoperfusion [Van Lieshout 1991, Sung 2000, Grubb 1991a, 1998]. In chapter 5, dynamic CA was shown to deteriorate during pre-syncope induced by HUT in normal subjects and patients with recurrent VVS, suggesting that impaired dynamic CA during presyncope may contribute to the pathophysiology of VVS by failing to preserve CBF. The reasons for deteriorating dynamic CA during pre-syncope are, however, unclear. A number of studies have suggested that paradoxical cerebral vasoconstriction may occur prior to syncope [Sung 2000, Grubb 1991a, 1998, Levine 1994] but these studies have, in the main, used Gosling's pulsatility index (GPI) as a marker for rises in cerebrovascular resistance (CVR) [Sung 2000, Grubb 1991a, 1998, Levine 1994]. As arterial BP is not taken into account in its calculation, GPI may be poorly applicable to the cerebral circulation [Czosnyka 1996, Schondorf 1997, Aaslid 1992] and, more recently, others have used Poiseuille's Law and classical CVR (CCVR) to suggest that CVR actually falls prior to VVS [Schondorf 1997]. Calculations of CCVR assume, however, that blood flow continues in cerebral vessels at arterial BP levels approaching 0mmHg, an assumption that is almost certainly untrue [Dewey 1974, Panerai 1995,1999b]. It is more likely that CBF ceases at arterial pressures above 0mmHg due to vessel wall collapse, and this pressure below which flow in a blood vessel ceases is termed critical closing pressure (CrCP) [Dewey 1974, Panerai 1995,1999b]. CrCP has an inverse relationship with carbon dioxide (CO₂) concentrations [Panerai 1999b, Garnham 1999], can change markedly over a matter of seconds [Carey 2000a] and may have a profound effect on cerebral blood flow [Dewey 1974, Panerai 1995, 1999b, Carey 2000a, Czosnyka 1999, Panerai 1996,

Evans 1988]. Resistance-area product (RAP) is an index of CVR which, unlike CCVR, allows for CrCP by not making the assumption that cerebral blood flow velocity (CBFV) intercepts the arterial BP axis at 0mmHg [Panerai 1999b, Carey 2000a, Czosnyka 1999, Panerai 1996, Evans 1988]. Using these more accurate modeling techniques, the hypothesis that changes in CrCP,

RAP and CO₂ may help to explain cerebral haemodynamics prior to VVS was investigated.

6.3. Objectives

- 1. To assess the pre-syncopal cerebrovascular responses of normal subjects and patients with recurrent VVS.
- 2. To assess pre-syncopal ETCO₂ and TCO₂ changes in both groups.
- 3. To assess pre-syncopal RAP and CCVR changes in both groups.
- 4. To assess pre-syncopal CrCP changes in both groups.

6.4. Methods

6.4.1. Subjects

16 patients with a history of recurrent VVS diagnosed according to the criteria outlined in section 4.4.1. were recruited from a specialist out-patient syncope clinic. 65 control subjects were also recruited from a volunteer register in the department and from departmental staff. Control subjects were healthy as assessed by history and physical examination, full neurological assessment and standard autonomic function tests. None of the patients or control subjects was taking any medication known to have cardiovascular or cerebrovascular effects.

6.4.2. Study Protocol

All subjects underwent the initial preparatory protocol outlined in section 2.6. In addition, end-tidal CO₂ was measured via a closely fitting face-mask and an infra-red capnograph (Capnogard, Novametrix) as described in section 2.3.1. (Figure 2.7). The vertical height in centimetres (Ht) from the point of insonation of the right MCA to the second intercostal space was also recorded for each subject. Subjects then underwent a head-up tilt protocol identical to that outlined in section 5.4.2. Once again, in order to minimise discomfort and improve compliance with the study protocol, ETCO₂ measurements were discontinued 5 minutes after tilt in all subjects. For ethical reasons, all pre-syncopal subjects were replaced supine prior to loss of consciousness and the point at which subjects were replaced supine was taken as time 0.

6.4.3. Data Analysis

Data were initially analysed as outlined in section 2.7. with estimates of SBP, DBP, MAP, systolic, diastolic and mean CBFV, pulse interval (PI), TCO₂ and ETCO₂ being made for each cardiac cycle. MCA systolic pressure (MCASP), MCA diastolic pressure (MCADP) and MCA mean pressure (MCAMP) during HUT were estimated from SBP, DBP and MAP respectively by subtracting the hydrostatic pressure (Ht * $0.735 * \sin 70^{\circ}$).

CCVR and GPI were calculated for each cycle using the formulae (section 2.1.5.4.): CCVR = MCAMP / MCBFV

GPI = (SCBFV - DCBFV) / MCBFV

CrCP and RAP were calculated for each cardiac cycle by plotting the instantaneous CBFV waveform as a function of MCA BP and performing linear regression analysis, as described in section 2.1.5.4. [Panerai 1999b, Carey 2000a, Czosnyka 1999, Panerai 1996, Evans 1988]. The slope of the regression line represents the inverse of RAP while the arterial BP axis intercept of the regression line determines the CrCP (Figure 2.5.) [Panerai 1999b, Carey 2000a, Czosnyka 1999, Panerai 1996, Evans 1988]. Time 0 was synchronised for all subjects using a mark generated by an electrical device each time the tilt table passed through 45°. All files were inspected individually and detailed averaged profiles of all parameters calculated for the fourminute period from –180s to +60s for both syncopal patients and syncopal control subjects.

6.4.4. Statistical analysis

Demographic details and baseline characteristics of the syncopal patient group, syncopal control group and non-syncopal control group were compared using oneway analysis of variance. The between group differences and changes between parameter values at -180 s and time 0 were modeled for each outcome using a mixed model for repeated measures (SAS proc Mixed). Model selection was by changes in the log likelihood and denominator degrees of freedom were calculated using Satterthwaite's method [Satterthwaite 1946]. Different covariance patterns were investigated using Akaike's Information Criterion [Akaike 1973]. Due to skewing of variables, parameter values of the non-syncopal control group at the mean time to syncope of the patient group were compared with parameter values at time 0 of the syncopal patients and syncopal controls using the Kruskal-Wallis test with between group differences assessed using Wilcoxon's signed rank tests.

6.5. Results

All 16 of the patients with recurrent VVS (syncopal patients) and 14 of the control subjects (syncopal controls) developed VVS. No significant differences were demonstrated in baseline characteristics between the 16 syncopal patients, 14 syncopal controls and 51 control subjects who did not develop VVS (non-syncopal controls) (Table 6.1.). Imminent syncope tended to occur sooner after HUT in the patient group than in the control group, but the difference was not significant (302 seconds; 95% CI –116 to 737; p=0.17). No significant differences were demonstrated during pre-syncope between the syncopal patient group and syncopal control group for any parameter examined.

Changes in pre-syncopal parameters for the syncopal patient group and syncopal control group between -180 s and time 0 are shown in Table 6.2. Mean MCASP, MCAMP and MCADP and mean PI for the two groups for the four-minute period between -180s and +60s are displayed in Figures 6.1 and 6.2 respectively. SCBFV, MCBFV and DCBFV for the same period are displayed in Figure 6.3, mean RAP and CCVR in Figure 6.4, mean GPI in Figure 6.5, mean ETCO₂ and TCO₂ in Figure 6.6 and mean CrCP and MCADP in Figure 6.7. Changes in the instantaneous pressure-velocity relationships of a representative patient for the cardiac cycles at -180s and - 20s are displayed in Figure 6.8.

SCBFV, DCBFV and MCBFV fell significantly during pre-syncope in both groups, though the fall in DCBFV was greater than the fall in SCBFV in both the syncopal patient group ($41.2 \pm 23.1\%$ v 7.3 $\pm 16.5\%$, p<0.0001) and the syncopal control group $(44.5 \pm 19.8 \% \text{ v } 6.3 \pm 12.9\%, \text{ p} < 0.0001)$ and accounted for most of the fall in MCBFV (Table 6.2. & Figure 6.3.).

Although TCO₂ data were available for all subjects, pre-syncopal ETCO₂ results are available for 7 subjects only (5 patients and 2 controls) who developed syncope within 5 minutes of HUT. Despite the small numbers, ETCO₂ demonstrated a significant decline (41.2 \pm 3.7 mmHg v 37.4 \pm 4.1 mmHg; difference 3.8 mmHg, 95%CI 0.9 to 6.6, p=0.017) during pre-syncope in the 7 subjects (Figure 6.6.). CrCP rose significantly during pre-syncope in both groups, approaching MCADP (Table 6.2 & Figure 6.7) before falling precipitously on syncope.

757 seconds after HUT (the mean time to syncope of the patient group) all parameter values of the non-syncopal control group were significantly higher (p<0.0001) than values at time 0 in the syncopal patient and syncopal control groups, except for CrCP and GPI which were significantly lower (p<0.0001) (Table 6.2). These findings were unchanged irrespective of the time after HUT chosen for the non-syncopal control group (results not presented).

6.6. Discussion

This study has demonstrated the previously unreported findings during pre-syncope of progressive hypocapnia, rising CrCP, falling RAP and similar pre-syncopal changes in patients with recurrent VVS and control subjects. The demonstration of a rise in CrCP, an important factor in the control of CBF, during pre-syncope may help to explain complex and seemingly contradictory pre-syncopal cerebral haemodynamics in the light of a new hypothesis – increases in CrCP during pre-syncope, possibly due

to progressive hypocapnia, selectively impair diastolic CBF, thereby counteracting autoregulatory decreases in CVR and causing impaired CA prior to VVS.

6.6.1. Carbon Dioxide changes during pre-syncope

Changes in ventilation and CO₂ levels prior to syncope are complex. Schondorf [1997] demonstrated no change in respiratory rate prior to syncope but CO₂ concentrations were not measured. Novak [1998] demonstrated that hypocapnia, caused by hyperventilation resulting from an increased tidal depth without an increased rate, occurs in patients with orthostatic intolerance subjected to head-up tilt. Lipsitz [1998] indicated that hyperventilation probably occurred prior to VVS by illustrating an increase in respiratory amplitude without any change in respiratory rate, but, again, CO₂ levels were not measured. Taken in conjunction with the demonstration of falling CO₂ concentrations prior to syncope, it would appear that hypocapnia due to an increase in respiratory amplitude may form an intrinsic part of the pathophysiology of VVS. This is unsurprising as an increase in respiratory depth is a well-known pressor mechanism causing venoconstriction and increased preload [Shepherd 1981]. Gilliat [1948] has also reported a spinal vasoconstrictor reflex activated by deep inspiration which may lead to increases in systemic BP.

6.6.2. Pre-syncopal changes in cerebrovascular resistance

Pathophysiological hypocapnia prior to VVS might be expected to cause cerebral vasoconstriction and lead to the reduction in CBFV found and, taking GPI as an index of CVR, this would appear to be so. Calculations of GPI do not, however, take systemic BP levels or CrCP into account and probably do not, therefore, accurately reflect changes in CVR [Czosnyka 1996, Schondorf 1997, Aaslid 1992]. Indeed,

there may be an inverse relationship [Czosnyka 1996]. In this study, however, GPI was calculated (with a rise demonstrated during pre-syncope) to illustrate that the findings are consistent with those of others who have studied VVS. CCVR in probably a more reliable index of CVR as its calculation takes BP into account and this work would seem confirm the findings of Schondorf et al [11] that CCVR decreases during pre-syncope. Calculation of CCVR, however, assumes that the CrCP of the cerebral circulation is 0mmHg and constant, and both of these assumptions are most unlikely to be true. RAP is probably the best available index of CVR as CrCP is taken into account in its calculation and it truly reflects the beat-tobeat changes in the instantaneous relationship between CBFV and arterial BP [Panerai 1999b, Carey 2000a, Czosnyka 1999, Panerai 1996, Evans 1988]. In this study, a previously unreported substantial decline in RAP was shown during pre-syncope, similar to the decline in CCVR, suggesting that distal arteriolar vasodilatation is taking place. The relative preservation of SCBFV that was demonstrated in the face of a precipitously falling SBP, would appear to confirm the presence of cerebral vasodilatation during pre-syncope.

6.6.3. Pre-syncopal changes in critical closing pressure

If cerebral arteriolar vasodilatation is occurring, why are MCBFV and DCBFV not being maintained? These results suggest that changes in CrCP during the presyncopal period as well as the low MCADP may be responsible. CrCP is seen to consistently rise during pre-syncope, possibly due to the simultaneous decline in CO₂ levels [Panerai 1999, Garnham 1999, Aaslid 1992], before abruptly falling immediately prior to syncope (Figure 6.7.). DCBFV is seen to decrease by a much greater amount than SCBFV during pre-syncope (Table 6.2. & Figure 6.3.), even though MCASP declines in much the same fashion as MCADP (Figure 6.1.). This finding has been previously reported by others [Sung 2000, Grubb 1991a, Schondorf 1997] and accounts for the rise in GPI during pre-syncope (Figure 6.5.). The reason for the sharp decline in DCBFV despite active vasodilatation can be explained by CrCP approaching MCADP during pre-syncope (Figure 6.7.). The convergence of CrCP and MCADP probably results in a substantial number of downstream vessels collapsing during diastole leading to severe curtailment of DCBFV (Figures 6.3. & 6.7.). As PI rises in late pre-syncope (Table 6.2. & Figure 6.2.), diastole comprises a relatively greater proportion of the cardiac cycle and this may compound the problem of reduced CBF during diastole. Rises in CrCP, therefore, seem to counteract the effects of cerebral arteriolar vasodilatation by reducing CBF during diastole.

6.6.4. Carbon dioxide and critical closing pressure during pre-syncope

Increasing respiratory depth prior to syncope, therefore, may help maintain systemic BP but may have deleterious effects on CBF through the action of hypocapnia on CrCP. Novak [1998] found that rebreathing CO_2 has the effect of abolishing symptoms and improving CBFV and CVR in patients with orthostatic hypotension subjected to orthostatic stress. It has been suggested that higher CO_2 concentrations may work by inducing cerebral vasodilatation, and this study suggests that this may occur through an action on CrCP. Whether employing rebreathing techniques to increase CO_2 levels would be an appropriate treatment for patients with recurrent VVS is worthy of further investigation.

6.6.5. Similar changes in patients and syncopal controls

The findings of this study suggest that changes in the pre-syncopal period are similar for patients with recurrent VVS and control subjects who have VVS induced by headup tilt. It is unlikely, therefore, that impaired control of CBF is the primary problem in patients with recurrent VVS, but this warrants further investigation.

6.6.6. Limiting factors

The major limiting factor of this work, similar to the work in chapter five, remains the inability to directly measure MCA BP. As discussed in section 5.6.5., non-invasive plethysmography correlates very well with intra-arterial pressure recordings during HUT [Petersen 1995] and intracranial pressure changes after HUT are likely to be relatively small [Drummond 2000] and similar in syncopal and non-syncopal subjects alike. Changes in venous pressure will occur to an equal and proportionate degree as arterial pressure and, in the absence of better, non-invasive alternatives, therefore, calculations using non-invasive plethysmography provide acceptable estimates of MCA pressure.

Calculations also make the assumption analogous to that in chapter five that both MCA diameter and intracranial pressure remain constant in the pre-syncopal period. As discussed in section 5.6.5., simulated orthostasis does not affect MCA calibre [Serrador 2000] and changes in CO₂ concentrations to the degree demonstrated in this study would not be expected to affect MCA diameter [Hubner 1967, Giller 1993] or intracranial pressure [Drummond 2000]. Similarly, profound hypotension during presyncope could potentially cause MCA myogenic vasodilatation and influence CBFV, but the methodology employed did not allow assessment of this possibility. While TCO₂ and ETCO₂ measurements accurately reflect arterial CO₂ measurements in the steady state, some doubts remain over accuracy during hyperventilation and changes in cardiac output [Steurer 1997, Weil 1985]. In addition, temporal resolution is poor with TCO₂ measurements, but the facts that TCO₂ levels continued to decline in pre-syncopal subjects for ~40s after time 0 (Figure 6.6), that ETCO₂ levels also declined during pre-syncope (Figure 6.6) and that others have demonstrated hyperventilation during pre-syncope [Lipsitz 1998], suggest that the conclusion that progressive hypocapnia occurs during pre-syncope is correct. Hypocapnia is known to increase CrCP [Aaslid 1992, Panerai 1999b,Garnham 1999, Czosnyka 1999], but the poor temporal resolution of the TCO₂ measurements prevented assessment of whether the rise in CrCP during pre-syncope was causally linked to the CO₂ changes and this is worthy of further study.

6.7. Conclusions

1. Similar pre-syncopal cerebrovascular and CO_2 changes occur in normal subjects and patients with recurrent VVS. Impaired CA is unlikely, therefore, to be the primary problem in patients with recurrent VVS.

2. End-tidal and transcutaneous CO_2 levels decline progressively during presyncope, but further work is needed to assess if one or both of these parameters accurately reflect pre-syncopal arterial CO_2 changes.

3. RAP, probably the best index of CVR as it is calculated taking CrCP and systemic BP into account, declines significantly during pre-syncope. CCVR, declines in a similar fashion during pre-syncope, but GPI rises mainly due to a marked fall in diastolic CBFV.

4. CrCP rises during pre-syncope, possibly due to progressive hypocapnia, but further work is needed to assess if there is, indeed, a causal relationship.

5. The rise in CrCP and low MCA diastolic pressure may account for the selective fall in diastolic CBFV during pre-syncope and lead to impairment of dynamic CA by attenuating reductions in CVR.

6. As hypercapnia is known to increase CrCP, there may be a role for rebreathing in the treatment of VVS with hypercapnia-induced rises in CrCP potentially facilitating better diastolic CBF during pre-syncope.

Figure 6.1.

Mean MCASP, MCADP and MCAMP 3 minutes prior to, and 1 minute after, being replaced supine at time 0 of 16 syncopal patients (red lines) and 14 syncopal controls (black lines).



Figure 6.2.

Mean PI changes 3 minutes prior to, and one minute after, being replaced supine at time 0 of 16 syncopal patients (red lines) and 14 syncopal controls (black lines).



Figure 6.3.

Mean SCBFV, MCBFV and DCBFV changes 3 minutes prior to, and one minute after, being replaced supine at time 0 of 16 syncopal patients (red lines) and 14 syncopal controls (black lines).



Figure 6.4.

Mean RAP (solid lines) and CCVR (dotted lines) changes 3 minutes before, and 1 minute after, being replaced supine at time 0 of 16 syncopal patients (red lines) and 14 syncopal controls (black lines).



Figure 6.5.

Mean GPI changes 3 minutes prior to, and 1 minute after, being replaced supine at time 0 of 16 syncopal patients (red lines) and 14 syncopal controls (black lines).



Figure 6.6.

Mean TCO2 changes of 16 syncopal patients (red line) and 14 syncopal controls (black line) and mean ETCO2 changes (blue dotted line) of 7 subjects (2 controls and 5 patients) 3 minutes prior to, and 1 minute after being replaced supine at time 0.



Figure 6.7.

Mean CrCP (solid lines) and MCADP (dotted lines) changes 3 minutes prior to, and 1 minute after, being replaced supine at time 0 of 16 syncopal patients (red lines) and 14 syncopal controls (black lines).



Figure 6.8.

Instantaneous pressure-velocity relationships of a representative patient for the cardiac cycles at -180s (circles) and -20s (triangles). Linear regression analysis demonstrates a CrCP of 23mmHg at -20s which is higher than the CrCP of 11mmHg at -180s despite arterial blood pressure being lower.


			Non-syncopal		
Parameter	Syncopal Patients	Syncopal Controls	Controls		
n	16	14	51		
Age	52 ± 22	53 ± 22	53 ± 22		
(years)	(17-86)	(21-87)	(20-90)		
Male : Female	7:9	6:8	24:27		
Systolic BP	132 ± 21	127 ± 19	130 ± 18		
(mmHg)	(105-179)	(94-151)	(103-166)		
Diastolic BP	76 ± 11	79 ± 12	79 ± 11		
(mmHg)	(60 – 97)	(62 – 92)	(61-98)		
Heart Rate	62 ± 10	65 ± 7	66 ± 10		
(bpm)	(46-78)	(53-76)	(49-102)		
Body mass index (kgm ⁻²)	22.4 ± 3.2	23.9 ± 2.8	24.5± 3.3		
	(16.9-27.8)	(19.8-28.4)	(16.8-30.8)		
Mean CBFV	51 ± 16	52 ± 15	55 ± 17		
(cms ⁻¹)	(21-83)	(29-78)	(22-86)		
TCO₂	40.9 ± 4.9	42.8 ± 3.8	41.9 ± 5.6		
(mmHg)	(30.1-47.0)	(33.3-46.8)	(29.6-50.8)		
ETCO ₂	42.1 ± 4.7	43.7 ± 5.2	42.2 ± 4.5		
(mmHg)	(32.2-49.3)	(32.9-50.4)	(34.0-51.4)		
Time to syncope	757 ± 539	1059 ± 615	N/A		
(s)	(61-1768)	(185-1800)			

Table 6.1.Demographic data and baseline characteristics of syncopal patients,
syncopal control and non-syncopal control subjects. No differences
were shown between the 3 groups for any parameter.

Devementer	Syncopal Controls				Syncopal Patients			Non-Syncopal Controls	
rarameter	-180 s	Time 0	Difference, (95% CI)	p value	-180 s	Time 0	Difference, (95% CI)	p value	757 s after HUT*
MCBFV (cm/s)	41 ± 12	28 ± 8	-13, (-10,-16)	< 0.0001	41 ± 13	29 ± 8	-12, (-8,-15)	< 0.0001	52 ± 13
SCBFV (cm/s)	65 ± 15	60 ± 16	-5, (-2,-9)	0.041	64 ± 17	59 ± 18	-5, (-0.9,-8.1)	0.017	80 ± 17
DCBFV (cm/s)	28 ± 11	15 ± 5	-13, (-10,-17)	< 0.0001	28 ± 11	16 ± 6	-12, (-8,-16)	< 0.0001	35 ± 10
MCAMP (mmHg)	60 ± 15	18 ± 17	-42, (-32,-53)	< 0.0001	62 ± 15	15 ± 11	-47, (-41,-53)	< 0.0001	73 ± 16
MCASP (mmHg)	91 ± 17	34 ± 24	-57, (-43,-70)	< 0.0001	90 ± 20	34 ± 14	-56, (-47,-65)	< 0.0001	112 ± 27
MCADP (mmHg)	47 ± 12	14 ± 13	-33, (-25,-42)	< 0.0001	49 ± 15	13 ± 10	-35, (-29,-42)	< 0.0001	54 ± 13
† CrCP (mmHg)	5.1 ± 13.8	11.6 ± 10.8	6.5, (1.2,11.8)	0.018	7.7 ± 17.2	14.5 ±14.7	6.8, (0.2,13.4)	0.044	2 ± 16
RAP (mmHg.s/cm)	1.4 ± 0.5	0.6 ± 0.6	-0.8, (-0.6,-1.0)	< 0.0001	1.4 ± 0.8	0.6 ± 0.4	-0.8, (-0.6,-1.1)	< 0.0001	1.5 ± 0.7
CCVR(mmHg.s/cm)	1.5 ± 0.4	0.6 ± 0.6	-0.9, (-0.8,-1.1)	<0.0001	1.7 ± 0.7	0.6 ± 0.4	-1.1, (-0.9,-1.3)	< 0.0001	1.5 ± 0.5
TCO ₂ (mmHg)	40.7 ± 5.0	39.3 ± 4.8	-1.4, (-0.4,-2.5)	0.011	38.2 ± 6.2	36.9 ± 6.5	-1.3, (-0.3,-2.2)	0.014	40.8 ± 3.7
PI (ms)	789 ± 142	1219 ±253	430, (294,620)	<0.0001	730 ± 188	1153 ±538	423, (291,742)	< 0.001	753 ± 102
GPI	0.95 ± 0.32	1.60 ± 0.52	0.65, (0.43,0.87)	< 0.0001	0.94 ± 0.27	1.50 ±0.39	0.56, (0.39,0.73)	< 0.0001	0.88 ± 0.17

Cardiovascular and cerebrovascular parameters at -180s and time 0 of 16 syncopal patients and 14 syncopal controls. *Also the average time to syncope of the syncopal patient group †Compares values at -180s with values at -20s. **Table 6.2.**

SCBFV, MCBFV, DCBFV	=	systolic, mean and diastolic cerebral blood flow velocity respectively
MCASP, MCAMP, MCADF) =	middle cerebral artery systolic, mean and diastolic pressure respectively
CrCP	=	middle cerebral artery critical closing pressure
RAP	-	resistance-area product
CCVR	=	classical cerebrovascular resistance
TCO ₂	=	transcutaneous carbon dioxide
PI	=	pulse interval
GPI	=	Gosling's pulsatility index

CHAPTER SEVEN

Conclusions of the Thesis

The work described in this thesis sought to explore cardiovascular and cerebrovascular responses in orthostatically induced syncopal and non-syncopal subjects in order to investigate the hypothesis that impaired cerebral autoregulation (CA) contributes to the pathophysiology of vasovagal syncope (VVS). More specifically, the effects of ageing and orthostatic stress on CA were assessed along with cerebrovascular and cardiovascular function in syncopal subjects at rest and during pre-syncope.

7.1. Summary of Results

Chapter 1 reviewed the literature regarding the historical aspects, epidemiology, pathophysiology, clinical characteristics, investigation, treatment and prognosis of VVS. In addition, literature on the physiological entity of CA, the inherent ability of the cerebral blood vessels to keep cerebral blood flow (CBF) constant for a wide range of systemic blood pressure (BP) levels, was reviewed in association with current understandings of the role of CA in the pathophysiology of VVS. CA occurs with a substantial degree of temporal heterogeneity and dynamic CA refers to the ability to maintain CBF in the face of BP changes occurring over a matter of seconds, reflecting the latency of the cerebral vasoregulatory system. Static CA refers to CBF adjustments in response to more prolonged BP changes and is a measure of the overall efficiency of the system. Static and dynamic CA may have different control mechanisms and dynamic CA may be more susceptible to damage in pathological states [Tiecks 1995, Dawson 2000]. It was noted that CBF has been shown to be impaired during pre-syncope [Janosik 1992, Bondar 1995, Sung 2000, Grubb 1991a, 1998, Levine 1994, Giller 1992] but that controversy over cerebrovascular resistance (CVR) clouded the issue over whether the impairment of CBF is due to abnormalities

of CA [Janosik 1992, Bondar 1995, Sung 2000, Grubb 1991a, 1998, Levine 1994, Giller 1992, Czosnyka 1996b, Schondorf 1997, Aaslid 1992]. Literature concerning the effect of orthostatic stress on CA in normal subjects and patients with recurrent VVS was discussed and felt to be scanty and disadvantaged by poor methodology. Chapter 2 reviewed the methods used in the studies of this thesis to measure systemic blood pressure (BP), CBF velocity (CBFV) and carbon dioxide (CO₂) continuously and non-invasively. Finger plethysmography using the Finapres device provides an accurate assessment of rapid changes in arterial BP compared to intra-arterial measurements, though exact BP levels may differ between methods [Imholz 1988, Rongen 1995, Petersen 1995, Omboni 1993, Jellema 1996]. Although arterial pressure waveforms change with head-up tilt, non-invasive plethysmography also correlates very well with intra-arterial pressure recordings during head-up tilt]. Because of multiple assumptions, including a constant diameter of the insonated vessel and an insonation angle of 0°, calculations of CBF from transcranial Doppler ultrasound (TCD) measurements do not correlate well with values derived using the traditional methods of CBF measurement, such as intravenous Xe clearance studies (see section 2.1.5.3.) [Bishop 1986]. The correlation between CBF and TCD measured CBF velocities is much better, however, especially during physiological manoeuvres. As CBF velocity and changes in velocity correlate well with CBF results obtained using traditional techniques, and because the temporal resolution is excellent, TCD has become the method of choice for the non-invasive assessment of rapid changes in CBF.

The temporal resolution of end tidal- CO_2 measurements is excellent, but as transcutaneous CO_2 measurements rely on gas diffusion, temporal resolution is poor (see section 2.3.2.). Transcutaneous [Trempler 1981, Mahutte 1984, Carter 1989,

Sridhar 1993, Dawson 1998] and end-tidal [Hillier 1990, Tobias 1994, Campbell 1994, Napolitano 1999, Schmitz 1995, Rozycki 1998] CO_2 measurements, however, give accurate non-invasive estimates of arterial CO_2 in the steady state (see section 2.3.).

Also in Chapter 2, thigh cuff inflation and release, Valsalva manoeuvre and lower body negative pressure (LBNP) application and release were reviewed as methods of inducing static and dynamic pressor and depressor changes and techniques of assessing the efficiency of static and dynamic CA using such induced BP changes were discussed. Drawbacks of using induced BP changes to assess CA and the potential advantages of using spontaneous BP changes at rest to assess dynamic CA were also evaluated. Finally, indices of CVR in current usage and their validity and methods of calculation were reviewed in detail.

Systemic BP levels are known to increase with age [Kannel 1978] while arterial compliance [Fleg 1986], cardiac baroreceptor sensitivity (BRS) [Gribbin 1971], CBF [Matsuda 1984] and CBFV [Krejza 1999] are lower in older subjects, but the effect of age on CA is unknown. As an age-related deterioration in CA could potentially explain the increased prevalence of syncope syndromes in older people, Chapter 3 investigated the effect of physiological ageing on CA. Twenty-seven young (<40 years) subjects were pair-matched for sex, systolic BP and body-mass index with twenty-seven older (>55 years) subjects, with young recruits being a mean of 39 years younger. Although age-related differences in cardiac BRS and CBFV were demonstrated between the two groups, no deterioration in static or dynamic CA was shown in older subjects despite the use of spontaneous and induced pressor and depressor BP changes. Induced and spontaneous BP changes were of a similar

magnitude in both groups except for phase IV of the Valsalva manoeuvre which had a greater pressor effect in the younger group. Despite this, no correlation was shown between the magnitude of BP changes and autoregulatory index (ARI) values. ARI values derived using thigh cuff release were significantly higher than those derived using LBNP release, but this was a consistent finding in both groups. The confidence intervals reported suggest that no clinically significant age-related deterioration occurs in dynamic CA but the wider confidence intervals for static ARI values mean that conclusions regarding the absence of an age-related deterioration in static CA are less robust.

The hypothesis that impaired cerebral autoregulatory function may contribute to the pathophysiology of syncope in patients with recurrent VVS arose because of reported reductions in CBFV during pre-syncope [Janosik 1992, Bondar 1995, Sung 2000, Grubb 1991a, 1998, Levine 1994, Giller 1992]. Chapter 4, therefore, addressed the possibility that abnormalities of CA during supine rest may predict the development of VVS. Static CA (assessed using thigh cuff inflation) and dynamic CA (assessed using thigh cuff release and spontaneous BP changes at rest with both Aaslid's model and frequency domain analysis) were found to be comparable in 17 patients with recurrent VVS and 17 pair-matched controls.

Although abnormalities of CA do not appear, therefore, to be present in patients with recurrent VVS during supine rest, the possibility remained that orthostatic stress could compromise CA in such patients. Indeed, conflicting evidence exists regarding the effect of orthostatic stress on CA in normal subjects with both preserved [Leftheriotis 1998, Lipsitz 2000] and impaired [Zhang 1998b] dynamic CA being reported.

Chapter 5, therefore, explored the effect of orthostatic stress induced by 70° head-up tilt (HUT) on dynamic CA in normal subjects and patients with recurrent VVS. Autoregulatory responses were identical in both groups before and immediately after HUT with patients and controls demonstrating preservation of indices of dynamic CA. ARI values subsequently deteriorated, however, during pre-syncope in patients and syncopal control subjects and remained impaired during the first minute after returning supine. In addition, a significant inverse relationship was found between dynamic ARI values and CO_2 changes after HUT in normal subjects, supporting the findings of others that hypocapnia improves autoregulatory capacity [Paulson 1990, Birch 1995, Aaslid 1989]. This relationship was not explored in patients or during pre-syncope due to the effect of critical closing pressure (CrCP) on CBFV discussed in Chapter 6.

The reasons for impairment of dynamic CA in the peri-syncopal period are unclear and Chapter 6 examined pre-syncopal changes in a number of cerebrovascular and cardiovascular parameters to help explain the findings. Impairment of CBFV during pre-syncope in patients and syncopal controls was seen to be primarily due to a reduction in diastolic CBFV in the face of a relative preservation of systolic CBFV. Two indices of CVR were demonstrated to fall markedly during pre-syncope, suggesting that active cerebral vasodilatation was taking place. The CrCP of the cerebral circulation rose significantly during pre-syncope, providing an explanation for the selective impairment of diastolic CBFV. As hypocapnia is known to increase CrCP, the progressive hypocapnia demonstrated during pre-syncope in both groups could potentially explain the concomitant rise in CrCP, but the methods used could not determine if this link was causal. Finally, no differences in cerebrovascular or cardiovascular parameters were demonstrated between patients and syncopal controls at any stage, suggesting that abnormalities of CA are not the primary pathophysiological problem in patients with recurrent VVS.

7.2. Study limitations

The author has attempted to highlight the limitations of these studies in each chapter and has mentioned some of them in the above summary.

All work involving transcranial Doppler ultrasound, including these studies, makes the assumption that the diameter of the insonated vessel remains constant throughout the duration of the study. Changes in CBF can only be reliably deduced from CBFV changes if the calibre of the insonated vessel is constant [Aaslid 1982], but reliable, non-invasive assessment of vessel diameter has proven notoriously elusive. A number of authors have used the spectral power of Doppler signals to show that MCA diameter does not change during thigh cuff application and release [Aaslid 1989] and during hypocapnia and hypercapnia [Poulin 1996], but a recent study has cast doubt on the usefulness of this method of assessing vessel diameter [Deverson 2000]. Significant change in MCA diameter is unlikely, however, during thigh cuff application and release [Newell 1994], Valsalva manoeuvre [Tiecks 1996], LBNP [Serrador 2000] or supine rest. Profound hypotension during pre-syncope, however, could potentially cause MCA myogenic vasodilatation and influence CBFV and as MCA diameter was not directly measured during these studies, caution must be exercised when interpreting the results.

One of the major limiting factors of this work is the indirect measure of MCA pressure during HUT in Chapters 5 and 6. Although arterial pressure waveforms

change with HUT, non-invasive plethysmography correlates very well with intraarterial pressure recordings during HUT [Petersen 1995]. A direct measurement of MCA pressure is impossible without very invasive procedures which would, in themselves, affect the interpretation of such work. Intracranial pressure changes after HUT are likely to be relatively small [Drummond 2000] and similar in syncopal and non-syncopal subjects alike, and changes in venous pressure will occur to an equal and proportionate degree as arterial pressure. In the absence of better, non-invasive alternatives, therefore, calculations using non-invasive plethysmography provide reasonable estimates of MCA pressure during HUT.

Transcutaneous CO₂ measurements correlate highly with arterial CO₂ levels but rely on gas diffusion and, therefore, have poor dynamic response characteristics (section 2.3.2.). In addition, doubts exist about the accuracy of transcutaneous and end-tidal CO₂ measurements during changes in cardiac output [Steurer 1997]. In the steady state, as in Chapters 3 and 4, these limitations do not present any problems, but difficulties arise due to the temporal delay and cardiac output changes induced by HUT and pre-syncope in Chapters 5 and 6. The transcutaneous CO₂ measurements reported in Chapter 5, however, were taken over stable one-minute periods and are consistent with the findings of others [Cencetti 1997, Anthonisen 1965, Bjurstedt 1962, Yoshizaki 1998], suggesting that they are a fair reflection of arterial CO₂ levels. With reference to transcutaneous and end-tidal CO₂ measurements in Chapter 6, the facts that transcutaneous CO₂ levels continued to decline in syncopal subjects after returning supine, that end-tidal CO₂ levels also declined during pre-syncope and that others have demonstrated hyperventilation during pre-syncope [Lipsitz 1998], suggest that the conclusion that progressive hypocapnia occurs during pre-syncope is correct. Nevertheless, a repeat of these studies with intra-arterial measures of CO_2 may be useful, though such invasive manoeuvres may present difficulties with the interpretation of HUT results [Kenny 2000, McIntosh 1994].

In Chapters 3 and 4, static and dynamic CA were assessed using spontaneous and induced BP changes that were probably not of a large enough magnitude to exceed the upper and lower BP limits within which CA is active. BP changes were, however, within "normal" physiological limits that one might encounter clinically, but the possibility of a narrowing or a shift of the BP plateau in older normal subjects and patients with recurrent VVS was not addressed. In addition, only two static BP stimuli were employed in Chapter 3 and only one in Chapter 4 resulting in relatively wide confidence intervals and a consequent dilution of conclusions regarding static CA. Further work is, therefore, required to precisely delineate the cerebral autoregulatory curve in older subjects and patients with recurrent VVS.

7.3. Prospects for further studies

The findings of this thesis have important implications for future research into physiological ageing and the pathophysiology of syncope.

Results presented in Chapters 3, 4 and 5 are supportive of the view that CA is normal in older subjects and patients with recurrent VVS. Further work is needed, however, to confirm the findings in relation to static CA and, in addition, static and dynamic CA need to be assessed at extremes of arterial BP to more clearly define the autoregulatory curve of both groups. Because of the findings with regard to dynamic CA in Chapter 3, investigators assessing dynamic CA with similar methods in the future may feel tempted to dispense with controlling for age as a confounding factor, but this cannot be recommended until the autoregulatory curve of older subjects is more clearly defined.

The discovery of preserved dynamic CA during orthostatic stress until pre-syncope intervenes in normal subjects and patients with recurrent VVS is an important finding. Only one method of assessing dynamic CA was used, however, and the findings require confirmation using other methods of quantifying dynamic CA. In addition, no attempt was made to assess the integrity of static CA during orthostatic stress and this is worthy of future investigation.

The findings in Chapter 6 of progressive hypocapnia during pre-syncope in association with a rise in CrCP and selective impairment of diastolic CBFV raises the possibility that rebreathing may be of benefit in patients with recurrent VVS. Increasing CO₂ levels through rebreathing has the potential to decrease CrCP and to allow better CBF during diastole with the possible result of reducing symptoms and increasing the time to syncope. Rebreathing would be an easy, inexpensive technique with little risk and is worthy of further study in either physiological or clinical trials. The efficacy of current treatments for VVS can be difficult to assess in view of the relapsing and remitting nature of the condition, the fact that any tolerable intervention is unlikely to eliminate all events and symptoms and the variable reproducibility of the HUT test (section 1.1.6.). Much further work is needed to quantify the clinical and physiological responses to pharmacological and non-pharmacological therapies in current usage for patients with recurrent VVS. More information about the effects of such interventions on CA and other cerebrovascular and cardiovascular parameters, such as baroreceptor sensitivity, may help the development of newer and more consistently beneficial treatments.

As intra-arterial CO₂ measurements are invasive, painful, not without risk [Downs 1973, Bedford 1977], have the potential to affect HUT results [Kenny 2000, McIntosh 1994] and are still difficult on a beat to beat basis, further work is needed to advance non-invasive techniques of assessing CO₂ levels, such as end-tidal and transcutaneous methods. As the effects of CO₂ on cerebral haemodynamics are profound (section 1.2.3.1.), CO₂ measurement is essential in studies designed to explore CBF and an improvement in non-invasive techniques would greatly enhance understanding of the relationship between CO₂ and CBF. More particularly, such an improvement may help answer the question raised in Chapter 6 as to whether there is a causal relationship between progressive hypocapnia and rising CrCP during pre-syncope.

7.4. Conclusion

This work suggests that the high prevalence of VVS in older age groups cannot be explained by 'physiologically' deteriorating CA as no age-related declines in static or dynamic CA were detected. The studies have also identified that abnormalities of dynamic CA exist during pre-syncope induced in normal subjects and patients with recurrent VVS, possibly due to hypocapnia-induced rises in cerebrovascular CrCP, but suggest that CA is otherwise normal in patients with recurrent VVS. As no differences were demonstrated in pre-syncopal cerebrovascular changes between normal subjects and patients at any stage, it is possible that impaired dynamic CA contributes to the pathophysiology of VVS but is unlikely to be the primary problem in patients with recurrent VVS. Further studies are needed, however, to confirm these

findings and to explore the interesting hypothesis that rebreathing may be of benefit in patients with recurrent VVS.

APPENDIX

Publications arising from this thesis

Carey BJ, Eames PJ, Blake MJ, Panerai RB, Potter JF. Dynamic cerebral autoregulation is unaffected by aging. Stroke. 2000; 31:2895-2900.

Carey BJ, Manktelow BN, Panerai RB, Potter JF. Cerebral autoregulatory responses to head-up tilt in normal subjects and patients with recurrent vasovagal syncope.

Circulation. 2001;104:898-902.

Carey BJ, Eames PJ, Panerai RB, Potter JF. Carbon dioxide, critical closing pressure and cerebral haemodynamics prior to vasovagal syncope in humans.

Clinical Science. 2001;101:351-358.

REFERENCES

Aaslid R.

Cerebral hemodynamics. In: Newell DW, Aaslid R, editors. Transcranial Doppler. NewYork:Raven Press;1992. p. 49-55.

Aaslid R, Lindegaard K-F, Sorteberg W, Nornes H. Cerebral autoregulation dynamics in humans. Stroke. 1989;20:45-52.

Aaslid R. Visually evoked dynamic blood flow response of the human cerebral circulation. Stroke. 1987;18:771-775.

Aaslid R, Markwalder T-M, Nornes H. Noninvasive transcranial Doppler ultrasound recording of flow in basal cerebral arteries. Journal of Neurosurgery. 1982;57:769-774.

Abboud FM. Neurocardiogenic syncope. New England Journal of Medicine. 1993;328:1117-1120.

Abboud FM, Aylward PE, Floras JS, Gupta BN. Sensitization of aortic and cardiac baroreceptors by arginine vasopressin in mammals. Journal of Physiology. 1986;377:251-65.

Abi-Samra F, Maloney J, Fouad FM, Castle L. The usefulness of head up tilt table testing and hemodynamic investigations in the workup of syncope of unknown origin. Pacing & Clinical Electrophysiology. 1988;10:406-410.

Aerts AJ, Dendale P, Daniels C, Meyvisch P, Kaufman L, Strobel G, Block P. Intravenous nitrates for pharmacological stimulation during head-up tilt testing in patients with suspected vasovagal syncope and healthy controls. Pacing & Clinical Electrophysiology. 1999;22:1593-1598.

Akaike H.

Information theory and an extension of the maximum likelihood principle. In: Retrov BN, Csaki F, eds. Second International Symposium on Information Theory. Budapest: Akademiai Kiado; 1973:267-281.

Akgoren N, Dalgaard P, Lauritzen M.

Cerebral blood flow increase evoked by electrical stimulation of rat cerebellar cortex: relation to excitatory synaptic activity and nitric oxide synthesis. Brain Research. 1996;710:204-214.

Almquist A, Goldenberg IF, Milstein S, Chen M-Y, Chen X-C, Hansen R, Gornick CC, Benditt DG.

Provocation of bradycardia and hypotension by isoproterenol and upright posture in patients with unexplained syncope.

New England Journal of Medicine. 1989;320:346-351.

Alonso-Galicia M, Hudetz AG, Shen H, Harder DR, Roman RJ. Contribution of 20-HETE to vasodilator actions of nitric oxide in the cerebral microcirculation. Stroke. 1999;30:2727-2734.

Ameriso SF, Mohler JG, Suarez M, Fisher M. Morning reduction of cerebral vasomotor reactivity. Neurology. 1994;44:1907-1909.

Ameriso SF, Paganini-Hill A, Meiselman HJ, Fisher M. Correlates of middle cerebral artery blood velocity in the elderly. Stroke. 1990; 21:1579-1583.

Ammirati F, Colivicchi F, Toscano S, Pandozi C, Laudadio MT, De Seta F, Santini M.

DDD pacing with rate drop function response versus DDI with rate hysteresis for cardioinhibitory vasovagal syncope. Pacing & Clinical Electrophysiology. 1998;21:2178-2181.

Anderson CR, Edwards SL, Furness JB; Bredt DS, Snyder SH.

The distribution of nitric oxide synthetase-containing autonomic preganglionic terminals in the rat. Brain Research. 1993;614:78-85.

Anthonisen NR, Bartlett JR, Tenney SM. Postural effect on ventilatory control. Journal of Applied Physiology. 1965;20:191-196.

Arnolds BJ, von Reutern G-M. Transcranial Dopplersonography. Examination technique and normal reference values. Ultrasound in Medicine and Biology. 1986;12:115-123.

Attinger EO, Anné A, McDonald DA. Use of Fourier series for the analysis of biological systems. Biophysical Journal. 1966;6:291-304.

Balaji S, Oslizlok PC, Allen MC, McKay CA, Gillette PC. Neurocardiogenic syncope in children with a normal heart. Journal of the American College of Cardiology. 1994;23:779-785.

Balldin UI, Krock LP, Hopper NL, Squires WG. Cerebral artery blood flow velocity changes following rapid release of lower body negative pressure. Aviation, Space and Environmental Medicine. 1996;67:19-22.

Barcroft H, Brod J, Heijl Z, Hirsjarvi EA, Kitchen AH.

The mechanisms of the vasodilatation in the forearm muscle during stress (mental arithmetic). Clinical Science. 1960;19:577-586.

Barcroft H, Edholm OG. On the vasodilation in human skeletal muscle during post-haemorrhagic fainting. Journal of Physiology. 1945;104:161-175.

Barcroft H, Edholm OG, McMichael J et al. Posthaemorrhagic fainting: study by cardiac output and forearm flow. Lancet. 1944;1:489-490.

Baumbach GL, Heistad DD. Heterogeneity of brain blood flow and permeability during acute hypertension. American Journal of Physiology. 1985;249:H629-H637.

Baumgartneer RW, Mathis J, Sturzenegger M, Mattle HP. A validation study on the intraobserver reproducibility of transcranial color-coded duplex sonography velocity measurements. Ultrasound in Medicine & Biology. 1994;20:233-237.

Bay-Hansen J, Ravn T, Knudsen GM. Application of interhemispheric index for transcranial Doppler sonography velocity measurements and evaluation of recording time. Stroke. 1997;28:1009-1014.

Bayliss WM. On the local reactions of the arterial wall to changes in internal pressure. Journal of Physiology (London). 1902;28:220.

Bedford R. Radial artery function following percutaneous cannulation with 18- and 20-gauge catheters. Anaesthesiology. !977;47:37-39.

Bedford R, Wollman H. Complications of percutaneous radial artery cannulation: an objective prospective study in man. Anaesthesiology. 1973;38:228-236.

Bendat JS, Piersol AG. Random data analysis and measurement procedures. John Wiley & Sons, Second edition, New York. 1986.

Benditt DG, Fahy GJ, Jurie KG, Sakaguchi S, Fabian W, Samniah N. Pharmacotherapy of neurally mediated syncope. Circulation. 1999;100:1242-1248.

Benditt DG, Samniah N, Sakaguchi S, Fahy G, Wilbert L.

Midodrine is effective in patients with refractory neurally-mediated syncope. Circulation. 1998;98:17(supplement I):706. Abstract.

Benditt DG. Neurally mediated syncopal syndromes: pathophysiological concepts and clinical evaluation. Review. Pacing & Clinical Electrophysiology. 1997;20:572-84.

Benditt DG, Ferguson DW, Grubb BP, Kapoor WN, Kugler J, Lerman BB, Maloney JD, Raviele A, Ross B, Sutton R, Wolk MJ, Wood DL. Tilt table testing for assessing syncope. ACC expert consensus document. Journal of the American College of Cardiology. 1996;28:263-275.

Bie P, Secha NH, Astrup A, Warberg J. Cardiovascular and endocrine responses to head-up tilt and vasopressin infusion in humans. American Journal of Physiology. 1986;251:R735-R741.

Birch AA, Dirnhuber MJ, Hartley-Davies R, Iannotti F, Neil-Dwyer G. Assessment of autoregulation by means of periodic changes in blood pressure. Stroke. 1995;26:834-837.

Bishop CCR, Powell S, Rutt D, Browse NL. Transcranial Doppler measurement of middle cerebral artery blood flow velocity: a validation study. Stroke. 1986;17:913-915.

Bjurstedt H, Hesser CM, Liljestrand G, Matell G. Effects of posture on alveolar-arterial CO_2 and O_2 differences and on alveolar dead space in man. Acta Physiologica Scandinavia. 1962;54:65-82.

Blake MJ, Panerai RB, Potter JF. Systemic blood pressure levels do not influence dynamic cerebral autoregulatory response to depressor and pressor stimuli in older people. Journal of Human Hypertension. 2000;13:882-883. Abstract.

Bloomfield D, Maurer M, Bigger JT Jr. Effects of age on outcome of tilt-table testing. American Journal of Cardiology. 1999;83:1055-1058.

Bondar RL, Kassam MS, Stein F, Dunphy PT, Fortney S, Riedesel ML. Simultaneous cerebrovascular and cardiovascular responses during pre-syncope. Stroke. 1995;26:1794-1800.

Bondar RL, Kassam MS, Stein F, Dunphy PT, Riedesel ML. Transcranial Doppler and arterial blood pressure response to lower body negative pressure.

Journal of Clinical Pharmacology. 1994;34:584-589.

Bootsma M, Swenne CA, Bruschke AVG. Similar orthostatic defense in active, healthy young adult and late middle-aged men. American Journal of Cardiology. 1995;76:922-927.

Borst C, Van Brederode JFM, Wieling W, Van Montfrans GA, Dunning AJ. Mechanisms of initial blood pressure response to postural change. Clinical Science. 1984;67:321-327.

Bos WJ, Imholz BP, van Goudoever J, Wesseling KH, van Montfrans GA. The reliability of noninvasive continuous finger blood pressure measurement in patients with both hypertension and vascular disease. American Journal of Hypertension. 1992;5:529-535.

Boulos M, Barron S, Nicolski E, Markiewicz W. Power spectral analysis of heart rate variability during upright tilt test: a comparison of patients with syncope and normal subjects. Review. Cardiology. 1996;87:28-32.

Bouma GJ, Muizelaar JP. Relationship between cardiac output and cerebral blood flow in patients with intact and impaired autoregulation. Journal of Neurosurgery. 1990;73:368-373.

Boyajian RA, Otis SM. Acute effects of smoking on human cerebral blood flow: a transcranial Doppler ultrasonography study. Journal of Neuroimaging. 2000;10:204-8.

Bredt DS, Hwang PM, Snyder SH. Localization of nitric oxide synthetase indicating a neural role for nitric oxide. Nature. 1990;347:768-770.

Brembilla-Perrot B, Marcon F, Worms AM, Gasparini J, Grentzinger A, Retournay G, Danchin N. Effects of age on the response to tilt test in patients with syncope. Archives des Maladies du Coeur et des Vaisseaux. 1996;89:431-434.

Brian JE Jr, Faraci FM, Heistad DD. Recent insights into the regulation of cerebral circulation. Clinical & Experimental Pharmacology & Physiology. 1996;23:449-457.

Bridgden W, Howarth S, Sharpey-Schafer EP. Postural changes in the peripheral blood-flow of normal subjects with observations on vasovagal fainting reactions as a result of tilting, the lordotic posture, pregnancy and spinal anaesthesia. Clinical Science. 1950;9:79-91.

Brignole M, Gagglioli G, Menozzi C, Del Rosso A, Costa S, Bartoletti A, Bottoni N, Lolli G.

Clinical features of adenosine sensitive syncope and tilt induced vasovagal syncope. Heart. 2000; 83:24-28.

Brignole M, Menozzi C, Gianfranchi L, Lolli G, Bottoni N, Oddone D. A controlled trial of acute and long-term medical therapy in tilt-induced neurally mediated syncope.

American Journal of Cardiology. 1992;70:339-342.

Bryan RM Jr, Steenberg ML, Eichler MY, Johnson TD, Swafford MW, Suresh MS. Permissive role of NO in alpha 2-adrenoceptor-mediated dilatations in rat cerebral arteries.

American Journal of Physiology. 1995;269:H1171-H1174.

Burton AC. On the physical equilibrium of the small blood vessels. American Journal of Physiology. 1951;164:319-329.

Campbell FA, McLeod MW, Bissonnette B, Swartz JS. End-tidal carbon dioxide measurement in infants and children during and after general anaesthesia. Canadian Journal of Anaesthesia. 1994;41:107-110.

Carey BJ, Eames PJ, Panerai RB, Potter JF. A case of arrhythmia-induced transient cerebral hyperaemia. Cerebrovascular diseases. 2000;10:330-333.

Carlioz R, Graux P, Haye J, Letourneau G, Guyomar Y, Hubert E, Bodart JC, Lequeuche B, Burlaton JP. Prospective evaluation of high dose and low dose isoproterenol upright tilt protocol for unexplained syncope in young adults. American Heart Journal. 1997;133:346-352.

Carter R. The measurement of transcutaneous oxygen and carbon dioxide tension during exercise testing. Breath. 1989;38.

Cencetti S, Bandinelli G, Lagi A. Effect of pCO₂ changes induced by head-upright tilt on transcranial Doppler recordings. Stroke. 1997;28:1195-1197.

Chen XC, Chen MY, Remole S, Kobayashi Y, Dunnigan A, Milstein S, Benditt DG. Reproducibility of head-up tilt-table testing for eliciting susceptibility to neurally mediated syncope in patients without structural heart disease. American Journal of Cardiology. 1992;69:755-760.

Chern CM, Kuo TBJ, Sheng WY, Wong WJ, Luk YO, Hsu LC, Hu HH. Spectral analysis of arterial blood pressure and cerebral blood flow velocity during supine rest and orthostasis. Journal of Cerebral Blood Flow and Metabolism. 1999;19:1136-1141.

Cholet N, Bonvento G, Seylaz J. Effect of neuronal NO synthetase inhibition on the cerebral vasodilatory response to somatosensory stimulation. Brain Research. 1996;708:197-200.

Chosy JJ, Grahm DT. Catecholamines in vasovagal fainting. Journal of Psychosomatic Research. 1965;9:189-194.

Chowdhary S, Townend JN. Role of nitric oxide in the regulation of cardiovascular autonomic control. Clinical Science. 1999;97:5-17.

Cohen RA. The role of nitric oxide and other endothelium-derived vasoactive substances in vascular disease. Progress in Cardiovascular Diseases. 1995;38:105-128.

Coleridge HM, Coleridge JC. Cardiovascular afferents involved in regulation of peripheral vessels. Review. Annual Review of Physiology. 1980;42:413-27.

Connolly SJ, Sheldon R, Roberts RS, Gent M. The North American vasovagal pacemaker study (VPS). A randomised trial of permanent cardiac pacing for the prevention of vasovagal syncope. Journal of the American College of Cardiology. 1999;33:16-20.

Cooke WH, Hoag JB, Crossman AA, Kuusela TA, Tahvanainen KUO, Eckberg DL. Human responses to upright tilt: a window on central autonomic integration. Journal of Physiology. 1999;517.2:617-628.

Czosnyka M, Smielewski P, Piechnik S, Al-Rawi PG, Kirkpatrick PJ, Matta BF, Pickard JD. Critical closing pressure in cerebrovascular circulation. Journal of Neurology, Neurosurgery & Psychiatry. 1999;66:606-11.

Czosnyka M (a), Richards HK, Whitehouse HE, Pickard JD. Relationship between transcranial Doppler – determined pulsatility index and cerebrovascular resistance: an experimental study. Journal of Neurosurgery. 1996;84:79-84.

Czosnyka M (b), Smielewski P, Kirkpatrick P, Pickard JD. Monitoring of cerebral autoregulation in head-injured patients. Stroke 1996;27:1829-1834.

Czosnyka M, Pickard J, Whitehouse H, Piechnik S.

The hyperaemic response to a transient reduction in cerebral perfusion pressure. A modelling study. Acta neurochirurgica. 1992;115:90-97.

Daffertshafer M, Hemmererici M. Cardiovascular regulation and vasoneuronal coupling. Journal of Clinical Ultrasound. 1995;23:125-128.

Dahl A, Lindegaard K-F, Russell D, Nyberg-Hansen R, Rootwelt K, Sorteberg W, Nornes H. A comparison of transcranial Doppler and cerebral blood flow studies to assess cerebral vasoreactivity. Stroke. 1992;23:15-19.

Davis SM, Ackerman RH, Correia J, Alpert NM, Chang J, Buonanno F, Kelley RE, Rosner B, Taveras JM. Cerebral blood flow and cerebrovascular CO₂ reactivity in stroke-age normal controls. Neurology. 1983;33:391-399.

Dawson SL, Blake MJ, Panerai RB, Potter JF. Dynamic but not static cerebral autoregulation is impaired in acute ischaemic stroke. Cerebrovascular Diseases. 2000;10:126-132.

Dawson SL (a), Panerai RB, Potter JF. Critical closing pressure explains cerebral haemodynamics during the Valsalva manoeuvre. Journal of Applied Physiology. 1999;86:675-680.

Dawson SL (b), Robinson TG, Youde JH, Martin A, James MA, Weston PJ, Panerai RB, Potter JF. Older subjects show no age-related decrease in cardiac baroreceptor sensitivity. Age & Ageing. 1999;28:347-353.

Dawson SL, Cave C, Pavord I, Potter J. Transcutaneous monitoring of blood gases: is it comparable with arterialised earlobe? Respiratory Medicine. 1998;92:584-588.

Day SC, Cook EF, Funkenstein H, Goldman L. Evaluation and outcome of emergency room patients with transient loss of consciousness. American Journal of Medicine. 1982;73:15-23.

de Buitleir M, Grogan EW Jr, Picone MF, Casteen JA. Immediate reproducibility of the tilt-table test in adults with unexplained syncope. American Journal of Cardiology. 1993;71:304-307. Deal BJ, Strieper M, Scagliotti D, Hulse E, Auld D, Campbell R, Strasburger JF, Benson DW Jr. The medical therapy of cardioinhibitory syncope in pediatric patients. Pacing & Clinical Electrophysiology. 1997;20:1759-1761.

Del Rosso A, Bartoli P, Bartoletti A, Brandinelli-Geri A, Bonechi F, Maioli M, Mazza F, Michelucci A, Russo L, Salvetti E, Sansoni M, Zipoli A, Fierro A, Ieri A. Shortened head-up tilt testing potentiated with sublingual nitroglycerin in patients with unexplained syncope. American Heart Journal. 1999;137:575-576.

Démolis P, Carville C, Giudicelli J-F. Effects of an angiotensin-converting enzyme inhibitor, lisinopril, on cerebral blood flow autoregulation in healthy volunteers. Journal of Cardiovascular Pharmacology. 1993;22:373-380.

Deverson S, Evans DH. Using Doppler signal power to detect changes in vessel size: a feasibility study using a wall-less flow phantom. Ultrasound in Medicine & Biology. 2000;26:593-602.

Dewey RC, Pieper HP, Hunt WE. Experimental cerebral haemodynamics. Vasomotor tone, critical closing pressure, and vascular bed resistance. Journal of Neurosurgery. 1974;41:597-606.

Di Girolamo E (a), Di Iorio C, Leonzio L, Sabatini P, Barsotti A. Usefulness of a tilt training program for the prevention of refractory neurocardiogenic syncope in adolescents. Circulation. 1999;100:1798-1801.

Di Girolamo E (b), Di Iorio C, Sabatini P, Leonzio L, Barbone C, Barsotti A. Effects of paroxetine hydrochloride, a selective serotonin reuptake inhibitor, on refractory vasovagal syncope: a randomised, double-blind, placebo-controlled study. Journal of the American College of Cardiology. 1999;33:1227-1230.

Donald DE, Shepherd JT. Reflexes from the heart and lungs: physiological curiosities or important regulatory mechanisms. Review. Cardiovascular Research.1978;12:446-69.

Downs JB, Rackstein AD, Klein EF Jr. Hazards of radial artery catheterisation. Anaesthesiology. 1973;38:283-286.

Driscoll DJ, Oacobsen SJ, Porter CJ, Wollan PC. Syncope in children and adolescents. Journal of the American College of Cardiology. 1997;29:1039-1045. Drummond JC, Patel PM. Neurosurgical Anesthesia. p1895-p1933 in Anesthesia. Miller RD (ed). 2000. Churchill Livingstone.

Dun NJ, Dun SL, Forstermann U. Nitric oxide synthetase immunoreactivityin rat pontine medullary neurons. Neuroscience. 1994;59:429-445.

Eckberg DL. Sympathovagal balance. A critical appraisal. Circulation. 1997;96:3224-3232.

Eckberg DL, Harkins SW, Fritsch JM, Musgrave GE, Gardner DF. Baroreflex control of plasma norepinephrine and heart period in healthy subjects and diabetic patients. Journal of Clinical Investigation. 1986;78:366-74.

Ector H, Reybrouck T, Heidbuchel H, Gewillig M, Van de Werf F. Tilt training: a new treatment for recurrent neurocardiogenic syncope and severe orthostatic intolerance. Pacing & Clinical Electrophysiology. 1998;21:193-196.

Eklöf B, Ingvar DH, Kagström E, Olin T. Persistence of cerebral blood flow autoregulation following chronic bilateral cervical sympathectomy in the monkey. Acta Physiologia Scandinavica. 1971;82:172-176.

Elam RF, Bergman F, Feurstein G. The use of antiserotonergic agents for treatment of acute hemorrhagic shock in cats. European Journal of Pharmacology. 1985;107:275-278.

El-Bedawi KM, Hainsworth R. Combined head-up tilt and lower body suction: a test of orthostatic tolerance. Clinical Autonomic Research. 1994;4:41-47.

Eliasson S, Folkow B, Lindgren P, Uvnas B. Activation of sympathetic vasodilator nerves to the skeletal muscles in the cat by hypothalamic stimulation. Acta Physiologica Scandinavica. 1951;23:333-351.

El-Sayed H, Hainsworth R. Salt supplement increases plasma volume and orthostatic tolerance in patients with unexplained syncope. Heart. 1996;75:134-140.

El-Sayed H, Hainsworth R. Relationship between plasma volume, carotid baroreceptor sensitivity and orthostatic tolerance. Clinical Science. 1995;88:463-70. Epstein MF, Cohen AR, Feldman HA, Raemer DB. Estimation of $PaCO_2$ by two noninvasive methods in the critically ill newborn infant. Journal of Pediatrics. 1985;106:282-286.

Evans DH, Levene MI, Shortland DB, Archer LNJ. Resistance index, blood flow velocity, and resistance area product in the cerebral arteries of very low birth weight infants during the first week of life. Ultrasound in Medicine and Biology. 1988;14:103-110.

Evans DH.

Some aspects of the relationship between instantaneous volumetric blood flow and continuous wave Doppler ultrasound recordings. Ultrasound in Medicine and Biology. 1982;8:605-609.

Evans RG, Ludbrook J, Potocnik SJ. Intracisternal naloxone and cardiac nerve blockade prevent vasodilatation during simulated haemorrhage in awake rabbits. Journal of Physiology. 1989;409:1-14.

Ewing DJ, Borsey DQ, Bellavere F, Clarke BF. Cardiac autonomic neuropathy in diabetes: comparison of measures of R-R interval variation. Diabetologia. 1981;21:18-24.

Faraci FM, Sobey CG. Role of soluble guanylate cyclase in dilator responses of the cerebral microcirculation. Brain Research. 1999;821:368-373.

Faraci FM. Role of endothelium derived relaxing factor in cerebral circulation: large arteries vs. microcirculation. American Physiological Society. 1991; H1038-H1042.

Faraci FM, Heistad DD. Regulation of large cerebral arteries and cerebral microvascular pressure. Circulation Research. 1990;66:8-17.

Faraci FM, Mayhan WG, Heistad DD. Segmental vascular responses to acute hypertension in cerebrum and brain stem. American Journal of Physiology. 1987;252:H738-H742.

Fish FA, Strasburger JF, Benson DW Jr. Reproducibility of a symptomatic response to upright tilt in young patients with unexplained syncope. American Journal of Cardiology. 1992;70:605-609.

Fitzpatrick AP, Lee RJ, Epstein LM, Lesh MD, Eisenberg S, Sheinman MM. Effect of patient characteristics on the yield of prolonged baseline head-up tilt testing and the additional yield of drug provocation.

Heart. 1996;76:406-411.

Fitzpatrick AP, Banner N, Cheng A, Yacoub M, Sutton R. Vasovagal reactions may occur after orthoptic heart transplantation. Journal of the American College of Cardiology. 1993;21:1132-1137.

Fitzpatrick AP, Williams T, Ahmed R, Lightman S, Bloom SR, Sutton R. Echocardiographic and endocrine changes during vasovagal syncope induced by prolonged head-up tilt. European Journal of Cardiac Pacing & Electrophysiology. 1992;2:121-128.

Fitzpatrick AP, Theodorakis G, Vardas P, Sutton R. Methodology of head upright tilt table testing in patients with unexplained syncope. Journal of the American College of Cardiology. 1991;17:125-130.

Fitzpatrick AP, Travill CM, Vardas PE, Hubbard WN, Wood A, Ingram A, Sutton R. Recurrent symptoms after ventricular pacing in unexplained syncope. Pacing & Clinical Electrophysiology. 1990;13:619-624.

Fitzpatrick A, Sutton R.

Tilting toward a diagnosis in unexplained recurrent syncope. Lancet. 1989;1:658-660.

Fleg J.

Alterations in cardiovascular structure and function with advancing age. American Journal of Cardiology. 1986;57:33-44.

Flevari P, Theodorakis GN, Zarvaliss E, et al.

Head-up tilt test after clomipramine challenge in neurally mediated syncope patients. Pacing & Clinical Electrophysiology. 1998;24(4 part II):793, A13. Abstract.

Fog M.

Cerebral circulation II. Reaction of pial arteries to increase in blood pressure. AMA Archives of Neurology and Psychiatry. 1939;41:260.

Fog M.

Reaction of the pial arteries to a fall in blood pressure. Archives of Neurology and Psychiatry. 1937;37:351.

Foglia-Manzillo G, Giada F, Beretta S, Corrado G, Santarone M, Raviele A. Reproducibility of head-up tilt testing potentiated with sublingual nitroglycerin in patients with unexplained syncope. American Journal of Cardiology. 1999;84:284-288.

Fredman CS, Biermann KM, Patel V, Uppstrom EL, Auer AI. Transcranial Doppler ultrasonography during head-upright tilt-table testing. Annals of Internal Medicine. 1995;123:848-849.

Frey MA, Hoffler GW.

Association of sex and age with responses to lower body negative pressure. Journal of Applies Physiology. 1988;65:1752-1756.

Fujishima M, Ibayashi S, Fujii K, Mori S. Cerebral blood flow and brain function in hypertension. Hypertension Research. 1995;18:111-117.

Fujishima M, Sadoshima S, Ogata J, Yoshida F, Shiokawa O, Ibayashi S, Omae T. Autoregulation of cerebral blood flow in young and aged spontaneously hypertensive rats (SHR). Gerontology. 1984;30:30-36.

Fuledsi B, Limburg M, Bereczki D, Michels RPJ, Neuwirth G, Legemate D, Valikovics A. Csiba L. Impairment of cerebrovascular reactivity in long-term type I diabetes. Diabetes. 1997;46:1840-1845.

Furlan R, Piazza S, Dell'Orto S, Barbic F, Bianchi A, Mainardi L, Cerutti S, Pagani M. Malliani A. Cardiac autonomic patterns preceding occasional vasovagal reactions in healthy humans. Circulation. 1998;98:1756-1761.

Gardiner SM, Compton AM, Bennett T, Palmer RMJ, Moncada S. Control of regional blood flow by endothelium-derived nitric oxide. Hypertension. 1990;15:486-492.

Garnham J, Panerai RB, Naylor AR, Evans DH. Cerebrovascular response to dynamic changes in pCO₂. Cerebrovascular Diseases. 1999;9:146-151.

Gatzoulis KA, Mamarelis IE, Apostolopoulos T, Dilaveris P, Gialafos J, Toutouzas P. Polymorphic ventricular tachycardia induced during tilt table testing in a patient with syncope and probable dysfunction of the sinus node. Pacing & Clinical Electrophysiology. 1995;18:1075-1079.

Gill RW. Measurement of blood flow by ultrasound: accuracy and sources of error. Ultrasound in Medicine & Biology. 1985;11:625-641.

Giller CA, Bowman G, Dyer H, Mootz L, Krippner W. Cerebral arterial diameters during changes in blood pressure and carbon dioxide during craniotony. Neurosurgery. 1993;32:737-742.

Giller CA, Levine BD, Meyer Y, Buckley JC, Lane LD, Borchers DJ. The cerebal haemodynamics of normotensive hypovolaemia during lower-body negative pressure. Journal of Neurosurgery. 1992;76:961-966.

Gilliat RW. Vasoconstriction in the finger after deep inspiration. Journal of Physiology. 1948;107:76-88.

Globus M, Keren A, Elda M, Granot C, Tzivoni D, Lavy S, Stern S. The effect of chronic propranolol therapy on regional cerebral blood flow in hypertensive patients. Stroke. 1983;14:964-966.

Gomez C, Tanasik DL, Lewis LM.Transcranial Doppler in the evaluation of global cerebral ischaemia: syncope and cardiac arrest.In: Bibikian V, Wechsler L, editors. Transcranial Doppler ultrasonography. St. Louis (MO): Mosby, Inc. 1999. pp 141-149.

Gonzalez-Heydrich J, Peroutka SJ. Serotonin receptor and reuptake sites: pharmacologic significance. Journal of Clinical Psychiatry. 1990;5(supplement 4):12. Abstract.

Gosling RG, King DH. Arterial assessment by Doppler-shift ultrasound. Proceedings of the Royal Society of Medicine. 1974;67:447-9.

Gowers WR. A lecture on vagal and vaso-vagal attacks. Lancet. 1907;1:1551-1554.

Graham LA, Gray JC, Kenny RA. Comparison of provocative tests for unexplained syncope: isoprenaline and glyceryl trinitrate for diagnosing vasovagal syncope. European Heart Journal. 2001;22:497-503.

Gribbin B, Pickering TG, Sleight P, Peto R. Effect of age and high blood pressure on baroreflex sensitivity in man. Circulation Research. 1971;29:424-431.

Grubb BP, Karas B, Kosinski D, Boehm K. Preliminary observations on the use of midodrine hydrochloride in the treatment of refractory neurocardiogenic syncope. Journal of Interventional Cardiac Electrphysiology. 1999;3:139-143.

Grubb BP, Samoil D, Kosinski D, Wolfe D, Brewster P, Elliott L, Hahn H. Cerebral syncope: loss of consciousness associated with cerebral vasoconstriction in the absence of systemic hypotension. Pacing & Clinical Electrophysiology. 1998;21:652-658.

Grubb BP (a), Samoil D. Neurocardiogenic syncope. In: Syncope in the older patient. RA Kenny, editor. Chapman & Hall, London. 1996. pp 91-106. Grubb BP (b), Kosinski D.

Current trends in etiology, diagnosis and management of neurocardiogenic syncope. Current Opinion in Cardiology. 1996;11:32-41.

Grubb BP, Samoil D, Kosinski D, Kip K, Brewster P. Use of sertraline hydrochloride in the treatment of refractory neurocardiogenic syncope in children and adolescents. Journal of the American College of Cardiology. 1994;24:490-494.

Grubb BP (a), Wolfe D, Samoil D, Temesy-Armos P, Hahn H, Elliott L. Usefulness of fluoxetine hydrochloride for prevention of resistant upright tilt induced syncope.

Pacing & Clinical Electrophysiology. 1993;16:458-464.

Grubb BP (b), Samoil D, Temesy-Armos P, et al. Episodic periods of neurally mediated hypotension and bradycardia mimicking transient ischaemic attacks in the elderly. Cardiology in the Elderly. 1993;1:221-226.

Grubb BP (a), Wolfe D, Temesy-Armos P, Hahn H, Elliott L. Reproducibility of head up tilt table test results in patients with syncope. Pacing & Clinical Electrophysiology. 1992;15:1477-1481.

Grubb BP (b), Wolfe D, Samoil D, Madu E, Temesy-Armos P, Hahn H, Elliott L. Recurrent unexplained syncope in the elderly: the use of head upright tilt table testing in evaluation and management. Journal of the American Geriatrics Society. 1992; 40:1123-1128.

Grubb BP (c), Rubin AM, Wolfe D, Temesy-Armos P, Hahn H, Elliott L. Head upright tilt table testing: a useful tool in the evaluation and management of recurrent vertigo of unknown origin associated with syncope or near syncope. Otolaryngology – Head & Neck Surgery. 1992;107:570-575.

Grubb BP (a), Gerard G, Roush K, Temesy-Armos P, Montford P, Elliott L, Hahn H, Brewster P. Cerebral vasoconstriction during head-upright tilt-induced vasovagal syncope: a paradoxic and unexpected response. Circulation. 1991;84:1157-1164.

Grubb BP (b), Temesy-Armos P, Hahn H, Elliott L. Utility of upright tilt-table testing in the evaluation and management of syncope of unknown origin. American Journal of Medicine. 1991;90:6-10.

Gur D, Wolfson SK, Yonas H, Good WF, Shabson L, Latchaw RE, Miller DM, Cook EE.

Progress in cerebrovascular disease: local cerebral blood flow by Xenon enhanced CT.

Stroke. 1982;13:750-758.

Hackel A, Linzer M, Anderson N, Williams R. Cardiovascular and catecholamine responses to head-up tilt in the diagnosis of recurrent unexplained syncope in elderly patients. Journal of the American Geriatrics Society. 1991;39:663-669.

Hainsworth R, el-Bedawi KM. Orthostatic tolerance in patients with unexplained syncope. Clinical Autonomic Research. 1994;4:239-244.

Hainsworth R. Syncope and fainting. p776. In: Autonomic failure. Bannister R & Mathias CJ editors. Oxford University Press. 1992.

Hainsworth R. Reflexes from the heart. Physiological Review. 1991;71:617-658.

Halsey J. Effect of emitted power on waveform intensity in transcranial Doppler. Stroke. 1990;21:1573-1578.

Hamilton W, Woodbury R, Harper H. Physiological relationships between intrathoracic, intraspinal and arterial pressures. Journal of the American Medical Association. 1936;107:853-856.

Hand IL, Shepard EK, Krauss AN, Auld PA. Discrepancies between transcutaneous and end-tidal carbon dioxide monitoring in the critically ill neonate with respiratory distress syndrome. Critical Care Medicine. 1989;17:556-559.

Harder DR, Lombard JM. Voltage-dependent mechanisms of receptor stimulation in cerebral arterial muscle. In: Bevan JA, Godfraind T, Maxwell RA, Stoclet JC, Wrocel M, eds. Vascular neuroeffector mechanisms. Amsterdam: Elsevier. 1985;181-186.

Hare JM, Givertz MM, Creager MA, Colucci WS. Increased sensitivity to nitric oxide synthetase inhibition in patients wit heart failure: potentiation of beta-adrenergic inotropic responsiveness. Circulation. 1998; 97: 161-166.

Hargreaves AD, Muir AL. Lack of variation in venous tone potentiates vasovagal syncope. British Heart Journal. 1992;67:486-490.

Harper AM. Autonomic control of cerebral blood flow. Cerebral Vascular Diseases. 1975;27-47.

Harper AM.

235

Autoregulation of cerebral blood flow: influence of the arterial blood pressure on the blood flow through the cortex.

Journal of Neurology, Neurosurgery and Psychiatry. 1966;29:398-403.

Harper AM. The interrelationship between pCO_2 and blood pressure in the regulation of blood flow through the cerebral cortex. Acta Neurologia Scandinavica. 1965;41:94-103.

Harper S, Bohlen G, Rubin M.

Arterial and microvascular contributions to cerebral cortical autoregulation in rats. American Journal of Physiology. 1984;246:H17-H24.

Haruhiko II, Shigeki I, Mitsuru M, Tetsuro U, Naoki M, Hiroshi K. Abnormal baroreflex sensitivity in patients with neurally mediated syncope. Circulation. 1996;94(Supplement I):I-621.

Hayano J, Sakakibara Y, Yamada A, Yamada M, Mukai S, Fujinama T, Yokoyama K, Wantanabe Y, Takata K. Accuracy of assessment of cardiac vagal tone by heart rate variability in normal subjects.

American Journal of Cardiology. 1991; 67:199-204.

Heistad DD (a), Marcus ML, Gross PM. Effects of sympathetic nerves on cerebral vessels in dog, cat and monkey. American Journal of Physiology. 1978;235:H544-H552.

Heistad DD (b), Marcus ML. Evidence that neural mechanisms do not have important effects on cerebral blood flow. Circulation Research. 1978;42:295-301.

Heistad DD, Marcus ML. Total and regional cerebral blood flow during stimulation of carotid baroreceptors. Stroke. 1976;7:239-243.

Hellebrandt FA, Franseen EB. Physiological study of the ventricle stance in man. Physiology Reviews. 1943;23:220-225.

Hernandez-Perez MJ, Raiche ME, Stone HL. The role of the peripheral sympathetic nervous system in cerebral blood flow autoregulation. Stroke. 1975;6:284-292.

Hildebrandt W, Schutze H, Stegemann J. On the reliability of the Penaz cuff during systemic and local fingertip vasodilatation at rest and in exercise. European Journal of Applied Physiology & Occupational Physiology. 1991;62:175-179.

Hillier SC, Badgwell JM, McLeod ME, Creighton RE, Lerman J. Accuracy of end-tidal pCO₂ measurements using a sidestream capnometer in infants and children ventilated with the Sechrist infant ventilator. Canadian Journal of Anaesthesia. 1990;37:318-321.

Hirsch AT, Levenson DJ, Cutler SS. Regional vascular responses to prolonged lower body negative pressure in normal subjects. American Journal of Physiology. 1989;257:H219-H225.

Hoffman WE, Albrecht RF, Miletich DJ. The influence of aging and hypertension in cerebral autoregulation. Brain Research. 1981;214:196-199.

Hongo K, Nakagomi T, Kassell NF, Sasaki T, Lehman M, Vollmer DG, Tsukahara T, Ogawa H, Torner J. Effects of aging and hypertension on endothelium-dependent vascular relaxation in rat carotid artery. Stroke. 1988;19:892-897.

Hubner P, Handa J. Effect of contrast material, hypercapnia, hyperventilation, hypertonic glucose and papaverine on the diameter of cerebral arteries. Investigative Radiology. 1967.2:17-32.

Hussain RM, McIntosh SJ, Lawson J, Kenny RA. Fludrocortisone in the treatment of hypotensive disorders in the elderly. Heart. 1996;76:507-509.

Iadecola C (a), Pelligrino DA, Moskowitz MA, Lassen NA. Nitric oxide synthetase inhibition and cerebrovascular regulation. Journal of Cerebral Blood Flow and Metabolism. 1994;14:175-192.

Iadecola C (b), Zhang F. Nitric-oxide-dependent and –independent components of cerebrovasodilation elicited by hypercapnia. American Journal of Physiology. 1994;266:R546-R552.

Imholz BPM, Van Montfrans GA, Settels JJ, Van der Hoeven GMA, Karemaker JM, Wieling W. Continuous non-invasive blood pressure monitoring: reliability of Finapres device during the Valsalva manoeuvre. Cardiovascular Research. 1988;22:390-397.

Janosik D, Gomez C, Njemanze P, et al.

Abnormalities in cerebral blood flow autoregulation during tilt induced neurocardiogenic syncope. Pacing & Clinical Electrophysiology. 1992;15:592.

Jellema WT, Imholz BP, van Goudoever J, Wesseling KH, van Lieshout JJ. Finger arterial versus intrabrachial pressure and continuous cardiac output during head-up tilt testing in healthy subjects. Clinical Science. 1996;91:193-200.

Johnson PC. Review of previous studies and current theories and autoregulation. Circulation Research. 1964;XIV & XV:1-3 - 1-9.

Jones RD, Kornberg JP, Roulson CJ, Visram AR, Irwin MG. The Finapres 2300e finger cuff. The influence of cuff application on the accuracy of blood pressure measurement. Anaesthesia. 1993;48:611-615.

Kannel W. Evaluation of cardiovascular risk in the elderly: the Framingham Study. Bulletin of the New York Academy of Medicine. 1978;54:573-591.

Karnik R, Valentin A, Winkler WB, Khaffaf N, Donath P, Slany J. Sex-related differences in acetazolamide-induced cerebral vasomotor reactivity. Stroke. 1996;27:56-58.

Kapoor WN, Smith MA, Miller NL. Upright tilt testing in evaluating syncope: a comprehensive literature review. American Journal of Medicine. 1994;97:78-88.

Kapoor WN, Brant N. Evaluation of syncope by upright tilt testing with isoproterenol. A non-specific test. Annals of Internal Medicine. 1992;116:358-363.

Kapoor WN. Diagnostic evaluation of syncope. American Journal of Medicine. 1991;90:91-106.

Kapoor WN. Evaluation and outcome of patients with syncope. Medicine. 1990;69:160-175.

Kapoor WN, Karpf M, Wieand S, Peterson JR, Levey GS. A prospective evaluation and follow-up of patients with syncope. New England Journal of Medicine. 1983;309:197-204.

Karnik R, Valentin A, Winkler WB, Khaffaf N, Donath P, Slany J. Sex-related differences in acetazolamide-induced cerebral vasomotor reactivity. Stroke. 1996;27:56-8.
Kastrup A, Dichgans J, Niemeier M, Schabet M. Changes of cerebrovascular CO_2 reactivity during normal aging. Stroke. 1998;29:1311-1314.

Kaufmann H, Berman J, Oribe E, Oliver JA. Possible increases in EDRF/NO in neurally mediated syncope. Clinical Autonomic Research. 1993;3:77. Abstract.

Kaufmann H, Oribe E, Oliver JA. Plasma endothelin during upright tilt: relevance for orthostatic hypotension? Lancet. 1991;338:1542-1545.

Kenny RA, O'Shea D, Parry SW. The Newcastle protocols for head-up tilt table testing in the diagnosis of vasovagal syncope, carotid sinus hypersensitivity, and related disorders. Heart. 2000;83:564-569.

Kenny RA. Introduction. In: Syncope in the older patient. RA Kenny, editor. Chapman & Hall, London. 1996. pp 1-14.

Kenny RA, Ingram A, Bayliss J, Sutton R. Head-up tilt: a useful tool for investigating unexplained syncope. Lancet. 1986;2:1352-1354.

Kety SS.

Human cerebral blood flow and oxygen consumption as related to aging. Journal of Chronic Diseases. 1956;8:478-486.

Kety SS, Schmidt CF. The nitrous oxide method for the quantitative determination of cerebral blood flow in

man: theory, procedure and normal values. Journal of Clinical Investigation. 1948;476-483.

Kety SS, Hafenschiel JH, Jeffers WA, Leopold IH, Shenkin HA. The blood flow, vascular resistance and oxygen consumption of the brain in essential hypertension. Cerebral Circulation in Essential Hypertension. 1947;pp 511-514.

Klimaschewski L, Kummer W, Mayer B, Couraud JY, Preissler U, Philippin B, Heym C.

Nitric oxide synthetase in cardiac nerve fibres and neurons of rat and guinea pig heart. Circulation Research. 1992;71:1533-1537.

Klingenheben T, Credner S, Hohnloser SH.

Prospective evaluation of a two-step therapeutic strategy in neurocardiogenic syncope: midodrine as second line treatment in patients refractory to beta-blockers. Pacing & Clinical Electrophysiology. 1999;22:276-281.

Kobari M (a), Fukuuchi Y, Tomita M, Tanahashi N, Takeda H. Role of nitric oxide in regulation of cerebral microvascular tone and autoregulation of cerebral blood flow in cats. Brain Research. 1994;667:255-262.

Kobari M (b), Fukuuchi Y, Tomita M, Tanahashi N, Konno S, Takeda H. Constriction/dilatation of the cerebral microvessels by intravascular endothelin-1 in cats.

Journal of Cerebral Blood Flow & Metabolism. 1994;14:64-69.

Koch AR.

Some mathematical forms of autoregulatory models. Circulation Research. 1964;XIV & XV:1-269 - 1-277.

Kochiadakis GE, Orfanakis A, Chryssostomakis SI, Manios EG, Kounali DK, Vardas PE.

Autonomic nervous system activity during tilt testing in syncopal patients, estimated by power spectral analysis of heart rate variability. Pacing & Clinical Electrophysiology. 1997;20:1332-41.

Koenig HM, Pelligrino DA, Wang Q, Albrecht RF. Role of nitric oxide and endothelium in rat pial vessel dilation response to isoflurane. Anesthesia & Analgesia. 1994;79:886-891.

Kontos HA. Nitric oxide and nitrosothiols in cerebrovascular and neuronal regulation. Stroke. 1993;24(Supplement):1155-158.

Kontos HA. Validity of cerebral arterial blood flow calculations from velocity measurements. Stroke. 1989;20:1-3.

Kontos HA, Wei EP. Oxygen-dependent mechanisms in cerebral autoregulation. Annals of Biomedical Engineering. 1985;13:329-334.

Kontos HA (a), Wei EP, Navari RM, Levasseur JE, Rosenblum WI, Patterson JL. Responses of cerebral arteries and arterioles to acute hypotension and hypertension. American Journal of Physiology. 1978;4:371-383.

Kontos HA (b), Wei EP, Raper AJ, Rosenblum WI, Navari RM, Patterson JL. Role of tissue hypoxia in local regulation of cerebral microcirculation. American Journal of Physiology. 1978;234:H582-H591.

Kosinski DJ, Grubb BP, Temesy-Armos PN. The use of serotonin re-uptake inhibitors in the treatment of neurally mediated cardiovascular disorders. Journal of Serotonin Research. 1994;1:85-90.

240

Kou WH, Randall DK, Dorset DN, Koch KS. Immediate reproducibility of tilt-table test results in elderly patients referred for evaluation of syncope or pre-syncope. American Journal of Cardiology. 1997;80:1492-1494.

Kouakam C, Lacroix D, Zghal N, Logier R, Klug D, Le Franc P, Jarwe M, Kacet S. Inadequate sympathovagal balance in response to orthostatism in patients with unexplained syncope and a positive head up tilt test. Heart. 1999;82:312-318.

Krejza J, Mariak Z, Walecki J, Szydlik P, Lewko J, Ustymowicz A. Transcranial color Doppler sonography of basal cerebral arteries in 182 healthy subjects: age and sex variability and normal reference values for blood flow parameters.

American Journal of Roentgenology. 1999;172:213-218.

Kuchinsky W, Wahl M. Local chemical and neurogenic regulation of cerebral vascular resistance. Physiological Reviews. 1978;58:656-689.

Kurbaan AS, Franzén A-C, Bowker TJ, Williams TR, Kaddoura S, Petersen MEV, Sutton R.

Usefulness of tilt test-induced patterns of heart rate and blood pressure using a twostage protocol with glyceryl trinitrate provocation in patients with syncope of unknown origin.

American Journal of Cardiology. 1999;84:665-670.

Lal SKL, HendersonRJ, Cejinar M, Hart MG, Hunyor SN. Physiological influences on continuous finger and simultaneous intra-arterial blood pressure.

Hypertension. 1995;26:307-314.

Lambert GW, Vaz M, Rajkumar C, Cox HS, Turner AG, Jennings GL, Esler MD. Cerebral metabolism and its relationship with sympathetic nervous activity in essential hypertension: evaluation of the Dickinson hypothesis. Journal of Hypertension. 1996;14:951-959.

Landmark K, Forsman M, Lindberg K, Ryman T, Martmann-Moe K, Haaverstd S, Wiel S.

Nitrendipine and mefruside in elderly hypertensives: effect on blood pressure, cardiac output, cerebral blood flow and metabolic parameters. Journal of Human Hypertension. 1995;9:281-285.

Lagerhof H, Eliash H, Werkol L.

Orthostatic changes of the pulmonary and peripheral circulation in man. Scandinavian Journal of Clinical & Laboratory Investigation. 1951;3:85.

Larsen FS, Olsen KS, Hansen BA, Paulson OB, Knudsen GM.

241

Transcranial Doppler is valid for determination of the lower limit of cerebral blood flow autoregulation. Stroke. 1994;25:1985-1988.

Larson J, Linnet N, Vesterager P. Transcutaneous devices for the measurement of pO_2 and pCO_2 . State of the art, especially emphasising a pCO_2 sensor based on a solid state glass pH sensor. Annales de Biologie Clinique. 1993;51:899-902.

Lassen NA, Roland PE, Larsen B, Melamed E, Soh K. Mapping of human cerebral functions: a study of the regional cerebral blood flow pattern during rest, its reproducibility and the activations seen during basic sensory and motor functions.

Acta Neurologica Scandinavica. Supplementum. 1977;64:262-263,274-275.

Lassen NA. Control of cerebral circulation in health and disease. Circulation Research. 1974;34:749-758.

Lassen NA. Cerebral blood flow and oxygen consumption in man. Physiological reviews. 1959;39:183-238.

Ledingham IM, Hanning CD.

Monitoring of ventilation. In: The Society of Critical Care Medicine. Textbook of Critical Care. Shoemaker WC, Ayres S, Grenvik A, et al. Editors. Philadelphia, WB Saunders, pp201-215.

Lee T-M, Su S-F, Chen M-F, Liau C-S, Lee Y-T. Usefulness of transdermal scopolamine for vasovagal syncope. The American Journal of Cardiology. 1996;78:480-482.

Leftheriotis G, Preckel MP, Fizanne L, Victor J, Dupuis JM, Saumet JL. Effect of head-upright tilt on the dynamic of cerebral autoregulation. Clinical Physiology. 1998;18:41-47.

Leftheriotis G, Geraud JM, Preckel MP, Saumet JL. Cerebral blood flow and resistances during hypotensive haemorrhage in the rabbit: transcranial Doppler and laser-Doppler flowmetry. Clinical Physiology. 1995;15:537-545.

Lepori M, Sartori C, Trueb L, Owlya R, Nicod P, Scherrer U. Haemodynamic and sympathetic effects of inhibition of nitric oxide synthetase by systemic infusion of N(G)-monomethyl-L-arginine into humans are dose dependent. Journal of Hypertension. 1998;16:519-523.

Levine BD, Giller CA, Lane LD, Buckley JC, Blomqvist CG. Cerebral versus systemic hemodynamics during graded orthostatic stress in humans. Circulation. 1994;90:298-306. Levy MN. Cardiac sympathetic-parasympathetic interactions. Federation Proceedings. 1984;43:2598-2602.

Lewis T. A lecture on vasovagal syncope and the carotid sinus mechanism with comments on Gowers's and Nothnagel's syndrome. British Medical Journal. 1932;1:873-876.

Lightfoot JT, Tsintgiras KM. Quantification of tolerance to lower body negative pressure in a healthy population. Medicine and Science in Sports and Exercise. 1995;27:697-706.

Lindegaard KF, Lundar T, Wiberg J, Sjoberg D, Aaslid U, Nornes H. Variations in middle cerebral artery blood flow investigated with non-invasive transcranial blood velocity measurements. Stroke. 1987;18:1025-1030.

Linzer M, Yang EH, Estes NAM, Wang P, Vorperian VR, Kapoor WN. Clinical guidline: Diagnosing syncope. Parts I & II. Annals of Internal Medicine. 1997;126&127:989-996&76-86.

Linzer M, Pontinen M, Gold DT, Divine GW, Felder A, Brooks WB. Impairment of physical and psychosocial function in recurrent syncope. Journal of Clinical Epidemiology. 1991;44:1037-1043.

Lippman N, Stein KM, Lerman BB. Failure to decrease parasympathetic tone during upright tilt predicts a positive tilttable test. American Journal of Cardiology. 1995;75:591-5.

Lipsitz LA, Hayano J, Sakata S, Okada A, Morin RJ. Complex demodulation of cardiorespiratory dynamics preceding vasovagal syncope. Circulation. 1998;98:977-983.

Lipsitz LA, Marka ER, Koestner J, Jonnson PV, Wei JY. Reduced susceptibility to syncope during postural tilt in old age. Is beta-blockade effective? Archives of Internal Medicine. 1989;149:2079-2083.

Lipsitz LA, Pluchino FC, Wei JV, Minaker JL, Rowe JW. Cardiovascular and norepinephrine responses after meal consumption in the elderly (older than 75 years) persons with postprandial hypotension and syncope. American Journal of Cardiology. 1986;58:810-815.

Liu JE, Hahn RT, Stein KM, Markowitz SM, Okin PM, Devereux RB, Lerman BB. Left ventricular geometry and function preceding neurally mediated syncope. Circulation. 2000;101:777-783. Loscalzo J. Nitric oxide and vascular disease. New England Journal of Medicine. 1995;333:251-253.

Lurie KG, Dutton J, Mangat R, Newman D, Eisenberg S, Scheinman m. Evaluation of edrophonium as a provocative agent for vasovagal syncope during head-up tilt table testing. American Journal of Cardiology. 1993;72:1286-1290.

Lurie KG, Dutton J, Mangat R, Scheinman MM. Pindolol is effective in patients with vasovagal syncope. Pacing in Clinical Electrophysiology. 1992;15:592. Abstract.

MacKenzie ET, McGeorge AP, Graham DI, Fitch W, Edvinsson L, Harper AM. Breakthrough of cerebral autoregulation and the sympathetic nervous system. Acta Neurologica Scandinavica. Supplementum. 1977;64:48-49.

Maeda H, Matasumoto M, Handa N, Hougaku H, Ogawa S, Tuskomoto Y, Kamada T.

Cerebral hemodynamics in hypertensive patients compared with normotensive volunteers: a transcranial Doppler study. Journal of Hypertension. 1994;12:191-197.

Mahanonda N, Bhuripanyo K, Kangkagate C, Wansanit K, Kulchot B, Nademanee K, Chaithiraphan S. Randomised double-blind placebo-controlled trial of oral atenolol in patients with unexplained syncope and positive upright tilt table results. American Heart Journal. 1995;130:1250-1253.

Mahony PJ, Panerai RB, Deverson ST, Hayes PD, Evans DH. Assessment of the thigh cuff technique for measurement of dynamic cerebral autoregulation. Stroke. 2000;31:476-480.

Mahutte CK, Michiels TM, Hassell KT, Trueblood DM. Evaluation of a single transcutaneous pO_2 -pCO₂ sensor in adult patients. Critical Care Medicine. 1984;12:1063-1066.

Mallett BL, Veall N. The measurement of regional cerebral clearance rates in man using xenon-133 inhalation and extracranial recording. Clinical Science. 1965;29:179-191.

Mangato DT, Hickey RF. Ischaemic injury following uncomplicated radial artery catheterisation. Anaesthesia and Analgesia. 1979;58:55-57.

Manolis AS, Linzer M, Salem D, Estes NA 3rd. Syncope: current diagnostic evaluation and management. Review. Annals of Internal Medicine. 1990;112:850-863.

Mansourati J, Blanc JJ. Tilt test procedure: angle, duration and positivity criteria. In: Blanc JJ, Benditt D, Sutton R, eds. Neurally mediated syncope: pathophysiology, investigations and treatment. The Bakken Research Center Series Vol 10. Armonk, New York: Futura. 1996:77-78.

Marangoni E, Zucchi A, Lissoni F, Oddone A, Ferraris P, Galloni G, Zappa MC, Orlandi M. Tilt test results in young and elderly patients with syncope of unknown origin. Aging (Milano). 1996;8:409-416.

Mark AL. The Bezold-Jarish reflex revisited: clinical implications of inhibitory reflexes originating in the heart. Journal of the American College of Cardiology. 1983;1:90-92.

Markus HS, Vallance P, Brown MM. Differential effect of three cyclooxygenase inhibitors on human cerebral blood flow velocity and carbon dioxide reactivity. Stroke. 1994;25:1760-1764.

Markwalder T-M, Grolimund P, Seiler RW, Roth F, Aaslid R. Dependency of blood flow velocity in the middle cerebral artery on end-tidal carbon dioxide partial pressure - a transcranial Doppler ultrasound study. Journal of Cerebral Blood Flow and Metabolism. 1984;4:368-372.

Martin PJ, Evans DH, Naylor AR. Transcranial color-coded sonography of the basal cerebral circulation. Reference data from 115 volunteers. Stroke. 1994;25:390-396.

Mason DT, Zelis R, Amsterdam EA. Actions of the nitrites on the peripheral circulation and myocardial oxygen consumption: significance in the relief of angina pectoris. Chest. 1971;59:296-305.

Mathew NT, Meyer JS, Hartmann A. Effect of alpha and beta adrenergic blocking agents on regional cerebral blood flow and CO₂ responsiveness in patients with cerebrovascular diseases. Stroke. 1973;4:372. Abstract.

Mathias CJ, Deguchi K, Schatz I. Observations on recurrent syncope and presyncope in 641 patients. Lancet. 2001;357:348-353.

Mathias CJ, Deguchi K, Bleasdale-Barr K, Kimber JR. Frequency of family history in vasovagal syncope. Lancet. 1998;352:33-34. Letter.

Matsuda H, Maeda T, Yamada Luo Xi Gui M, Hisada K. Age matched normal values and topographic maps for regional cerebral blood flow measurements by 133-Xe inhalation. Stroke. 1984;15:336-42.

Matteis M, Troisi E, Monaldo BC, Caltagirone C, Silvestrini M. Age and sex differences in cerebral hemodynamics. A transcranial Doppler study. Stroke. 1998;29:963-967.

Mattle HP, Nirkko AC, Baumgartner RW, Sturzenegger M. Transient cerebral circulatory arrest coincides with fainting in cough syncope. Neurology. 1995;45:498-501.

Matzen S, Emmeluth C, Milliken MC, Secher NH. Plasma endothelin-1 during central hypovolaemia in man. Clinical Physiology. 1992;12:653-658.

McAuley D, Silke B, Farrell S. Reliability of blood pressure determination with the Finapres with altered physiological states or pharmacodynamic conditions. Clinical Autonomic Research. 1997;7:179-184.

McGrady A, Bernal G. Relaxation based treatment of stress induced syncope. Behav. Ther. Experiment. Psychiatr. 1986;17:23-27.

McIntosh SJ, Lawson J, Kenny RA. Intravenous cannulation alters the specificity of head-up tilt testing for vasovagal syncope in elderly patients. Age & Ageing. 1994;23:317-319.

McIntosh S, daCosta D, Kenny RA. Benefits of an integrated approach to the investigation of dizziness, falls and syncope in elderly patients referred to a syncope clinic. Age & Ageing. 1993;22:53-58.

McPherson RW, Kirsch JR, Traystman RJ. N sup-omega-nitro-L-arginine methyl ester prevents cerebral hyperemia by inhaled anesthetics in dogs. Anesthesia & Analgesia. 1993;77:891-897.

Mehagnoul-Schipper DJ, Vloet LCM, Colier WNJM, Hoefnagels WHL, Jansen RW. Cerebral oxygenation declines in healthy elderly subjects in response to assuming the upright posture. Stroke. 2000;31:1615-1621.

Meyer JS, Rogers RL, Mortel KF, Judd BW.

Hyperlipidemia is a risk factor for decreased cerebral perfusion and stroke. Archives of Neurology. 1987;44:418-422.

Minson CT, Wladkowski SL, Pawelczyk JA, Kenney WL. Age, splanchnic vasoconstriction, and heat stress during tilting. American Journal of Physiology. 1999;276:R203-R212.

Mizumaki K, Fujiki A, Tani M, Shimono M. Left ventricular dimensions and autonomic balance during head-up tilt differ between patients with isoproterenol-dependent and isoproterenol-independent neurally mediated syncope. Journal of the American College of Cardiology. 1995;26:164-173.

Moncada S, Radomski MW, Palmer RM. Endothelium-derived relaxing factor. Identification as nitric oxide and role in the control of vascular tone and platelet function. Biochemical Pharmacology. 1988;37:2495-2501.

Montano N, Ruscone TG, Porta A, Lombardi F, Pagani M, Malliani A. Power spectrum analysis of heart rate variability to assess the changes in sympathovagal balance during graded orthostatic tilt. Circulation. 1994;90:1826-31.

Morgan DA, Thoren P, Wilczynski EA, Victor RG, Mark AL. Serotonergic mechanisms mediate renal sympathoinhibition during severe hemorrhage in rats. American Journal of Physiology. 1988;255:H496-H502.

Morillo CA, Eckberg DL, Ellenbogen KA, Beightol LA, Hoag JB, Tahvanainen KUO, Kuusela TA, Diedrich AM. Vagal and sympathetic mechanisms in patients with orthostatic vasovagal syncope. Circulation. 1997;96:2509-2513.

Morillo CA (a), Klein GJ, Gersh BJ.

Can serial tilt testing be used to evaluate therapy in neurally mediated syncope? American Journal of Cardiology. 1996;77:521-523.

Morillo CA, Ellenbogen KA, Wood MA, Beightol LA, Eckberg DL. Impaired cardio-vagal and muscle sympathetic baroreflex outflow in patients with head-up tilt induced neurally mediated syncope. Circulation. 1996;94(supplement I):I-544. Abstract.

Morillo CA, Klein GJ, Jones DL, Yee R. Time and frequency domain analyses of heart rate variability during orthostatic stress in patients with neurally mediated syncope. American Journal of Cardiology. 1994;74:1258-62.

Morillo C, Leitch JW, Yee R, Klein GJ.

247

A placebo-controlled trial of intravenous and oral disopyramide for prevention of neurally mediated syncope induced by head-up tilt. Journal of the American College of Cardiology. 1993;22:1843-1848.

Morita H, Nishida Y, Motochigawa H, Uemura N, Hosomi H, Vatner SF. Opiate receptor-mediated decrease in renal nerve activity during hypotensive hemorrhage in conscious rabbits. Circulation Research. 1988;63:165-172.

Morita Y, Hardebo JE, Bouskela E.

Influence of cerebrovascular parasympathetic nerves on resting cerebral blood flow, spontaneous vasomotion, autoregulation, hypercapnic vasodilation and sympathetic vasoconstriction.

Journal of the Autonomic Nervous System. 1994;49:s9-s14.

Mosqueda-Garcia R, Furlan R, Fernandez-Violante R, Desai T, Snell M, Jarai Z, Ananthram V, Robertson RM, Robertson D.

Sympathetic and baroreceptor reflex function in neurally mediated syncope evoked by tilt.

Journal of Clinical Investigation. 1997;99:2736-2744.

Mtinangi BL, Hainsworth R. Effects of moderate exercise training on plasma volume, baroreceptor sensitivity and orthostatic tolerance in healthy subjects. Experimental Physiology. 1999;84:121-130.

Mtinangi BL, Hainsworth R. Increased orthostatic tolerance following moderate exercise training in patients with unexplained syncope. Heart. 1998;80:596-600.

Murphy DGM, DeCarli C, McIntosh AR, Daly E, Mentis MJ, Pietrini P, Szczepanik J, Schapiro MB, Grady CL.

Sex differences in human brain morphometry and metabolism: an in vivo quantitative magnetic resonance imaging and positron emission tomography study on the effect of aging.

Archives of General Psychiatry. 1996;53:585-594.

Muscholl E. Peripheral muscarinic control of norepinephrine release in the cardiovascular system. American Journal of Physiology. 1980;239:H713-20.

Nag S, Kilty DW. Cerebrovascular changes in chronic hypertension. Stroke. 1997;28:1028-1034.

Napolitano LM. Capnography in critical care: accurate assessment of ARDS therapy? Critical Care Medicine. 1999;27:862-863. Naritomi H, Meyer JS, Sakai F, Yamaguchi F, Shaw TG. Effects of advancing age on regional cerebral blood flow studies in normal subjects and subjects with risk factors for atherothrombotic stroke. Archives of Neurology. 1979;36:410-416. Naschitz JE, Gaitini L, Mazov I, Eridzhanyan L, Keren D, Sabo E, Yeshurun D, Hardoff D. Jaffe M. The capnogaphy-tilt test for the diagnosis of hyperventilation syncope. Quarterly Journal of Medicine. 1997;90:139-145. Natale A, Sra J, Akhtar M, Kusmirek L, Tomassoni G, Leonelli F, Newby K, Beheiry S. Pacifico A. Use of sublingual nitroglycerin during head-up tilt-table testing in patients >60 years of age. American Journal of Cardiology. 1998;82:1210-1213. Natale A, Geiger MJ, Maglio C, Newby KH, Dhala A, Akhtar M, Sra J. Recurrence of neurocardiogenic syncope without pharmacologic interventions. American Journal of Cardiology. 1996;77:1001-1003. Nelson SD, Stanley M, Love CJ, Coyne KS, Schaal SF. The autonomic and hemodynamic effects of oral theophylline in patients with vasodepressor syncope. Archives of Internal Medicine. 1991;151:2425-2429. Nelson D, Fazekas JF. Cerebral blood flow in polycythemia vera. Archives of Internal Medicine. 1956;98:328-331. Newell DW (a), Aaslid R, Douville C, Byrd S, Schoonover K. Comparison of autoregulation and CO₂ reactivity in occlusive disease. Stroke. 1994;25:748. Newell DW (b), Aaslid R, Lam A, Mayberg TS, Winn R. Comparison of flow and velocity during dynamic autoregulation testing in humans. Stroke. 1994;25:793-797. Newell DW, Aaslid R. Transcranial Doppler: clinical and experimental uses. Cerebrovascular and Brain Metabolism Reviews. 1992;4:122-143. Njemanze P. Cerebral circulation dysfunction and hemodynamic abnormalities in syncope during head up tilt test. Canadian Journal of Cardiology. 1993;9:238-242. Nobili F, Rodriguez G, Marenco S, De Carli F, Gambaro M, Castello C, Pontremoli R, Rosadini G.

Regional cerebral blood flow in chronic hypertension. A correlative study.

Stroke. 1993;24:1148-1153.

Novak V (a), Spies J, Novak P, McPhee BR, Rummans TA, Low PA. Hypocapnia and cerebral hypoperfusion in orthostatic intolerance. Stroke. 1998;29:1876-1881.

Novak V (b), Novak P, Spies JM, Low PA. Autoregulation of cerebral blood flow in orthostatic hypotension. Stroke. 1998;29:104-111.

Novak V, Honos G, Schondorf R. Is the heart 'empty' at syncope? Journal of the Autonomic Nervous System. 1996;60:83-92.

Obrist WD, Thompson HK Jr, Wang HS, Wilkinson WE. Regional cerebral blood flow estimated by ¹³³Xenon inhalation. Stroke. 1975;6:245-256.

Okamoto H (a), Hudetz AG, Roman RJ, Bosnjak ZJ, Kampine JP. Neuronal NOS-derived NO plays permissive role in cerebral blood flow response to hypercapnia. American Journal of Physiology. 1997;272:H559-H566.

Okamoto H (b), Meng W, Ma J, Ayata C, Roman RJ, Bosnjak ZJ, Kampine JP, Huang PL, Moskowitz MA, Hudetz AG. Isoflurane-induced cerebral hyperemia in neuronal nitric oxide synthetase gene deficient mice. Anesthesiology. 1997;86:875-884.

Omboni S, Parati G, Frattola A, Mutti E, Di Rienzo M, Castiglioni P, Mancia G. Spectral and sequence analysis of finger blood pressure variability. Comparison with analysis of intra-arterial recordings. Hypertension. 1993;22:26-33.

Overgaard J, Mosdal X, Tweed WA.

Cerebral circulation after head injury. Part 3: Does reduced regional cerebral blood flow determine recovery of brain function after blunt head injury? Journal of Neurosurgery. 1981;55:63-74.

Owlya R, Vollenweider L, Trueb L, Sartori C, Lepori M, Nicod P, Scherrer U. Cardiovascular and sympathetic effects of nitric oxide inhibition at rest and during static exercise in humans. Circulation. 1997;96:3897-3903.

Padayachee TS, Kirkham FJ, Lewis RR, Gillard J, Hutchinson MCE, Gosling RG. Transcranial measurement of blood velocities in the basal cerebral arteries using pulsed Doppler ultrasound: a method of assessing the circle of Willis. Ultrasound in Medicine & Biology. 1986;12:5-14.

250

Pagani M, Lombardi F, Guzzetti S, Rimoldi O, Furlan R, Pizzinelli P, Sandrone G, Malfatto G, Dell'Orto S, Piccaluga E. Power spectral analysis of heart rate and arterial pressure variabilities as a marker of sympatho-vagal interaction in man and conscious dog. Circulation Research. 1986;59:178-93.

Palmero HA, Caeiro TF, Iosa DJ, Bas J. Baroreceptor reflex sensitivity index derived from phase IV of the Valsalva manoeuvre. Hypertension. 1981;3(SII):134-137.

Pandita-Gunawardena ND, Clarke SEM.

Amlodipine lowers blood pressure without affecting cerebral blood flow as measured by single photon emission computed tomography in elderly hypertensive subjects. Age & Ageing. 1999;28:451-457.

Panerai RB, Dawson SL, Eames PJ, Potter JF. Cerebral blood flow velocity response to induced and spontaneous sudden changes in arterial blood pressure. American Journal of Physiology. 2001;280:H2162-H2174.

Panerai RB (a), Dawson SL, Potter JF. Linear and nonlinear analysis of human dynamic cerebral autoregulation. American Journal of Physiology. 1999;277:H1089-1099.

Panerai RB (b), Deverson ST, Mahony P, Hayes P, Evans DH. Effect of CO_2 on dynamic cerebral autoregulation measurement. Physiological Measurement. 1999;20:265-275.

Panerai RB (a), Rennie JM, Kelsall AWR, Evans DH. Frequency-domain analysis of cerebral autoregulation from spontaneous fluctuations in arterial blood pressure. Medical & Biological Engineering & Computing. 1998;36:315-322.

Panerai RB (b), White RP, Markus HS, Evans DH. Grading of cerebral dynamic autoregulation from spontaneous fluctuations in arterial blood pressure. Stroke. 1998;29:2341-2346.

Panerai RB, Kelsall AWR, Rennie JM, Evans DH. Analysis of cerebral blood flow autoregulation in neonates. IEEE Transactions on Biomedical Engineering. 1996;43:779-788.

Panerai RB, Kelsall AW, Rennie JM, Evans DH. Estimation of critical closing pressure in the cerebral circulation of newborns. Neuropaediatrics. 1995;26:168-173.

Panerai RB, Coughtrey H, Rennie JM, Evans DH. A model of the instantaneous pressure-velocity relationships of the neonatal cerebral circulation. Physiological Measurement. 1993;14:411-418.

Parati G, Casadei R, Groppelli A, Di Rienzo M, Mancia G. Comparison of finger and intra-arterial blood pressure monitoring at rest and during laboratory testing. Hypertension. 1989;13:647-655.

Patrick JT, Adamo JM, Fritz JV, Dandonna P. Phase-contrast magnetic resonance angiography for the determination of cerebrovascular reserve. Journal of Neuroimaging. 1996;6:137-143.

Paulson OB, Strandgaard S, Edvinson L. Cerebral Autoregulation. Cerebrovascular and Brain Metabolism reviews. 1990;2:161-192.

Paulson OB, Olesen J, Christensen MS. Restoration of autoregulation of cerebral blood flow by hypocapnia. Neurology. 1972;22:286-293.

Pelletier CL. Shepherd JT. Circulatory reflexes from mechanoreceptors in the cardio-aortic area. Review. Circulation Research. 1973;33:131-8.

Penaz J.

Photoelectric measurement of blood pressure, volume and flow in the finger. Digest of the International Conference on Medicine and Biological Engineering. 1973:104.

Perez-Paredes M, Pico AF, Sanchez VJG, Exposito OE, Gonzalvez OM, Gonzalez CE, Inigo GL, Espinosa G, Florenciano SP, Ruiperez AJA. Role of adenosine triphosphate (ATP) in head-up tilt-induced syncope. Revista Espanola de Cardiologia. 1998;54:129-135.

Perez-Paredes M, Pico AF, Ruiperez AJA, Martinez SJ, Florenciano SR, Ruiz MF, Sanchez VJG, Exposito OE, Ruiz-Ros JA, Campos PJV. Use of transdermal scopolamine in the prevention of neurocardiogenic syncope induced by the tilt test. Revista Espanola de Cardiologia. 1995;48:480-485.

Perna GP, Ficola U, Salvatori MP, Stanislao M, Vigna C, Villella A, Russo A, Fanelli R, Paleani VPG, Loperfido F. Increase in plasma beta-endorphins in vasodepressor syncope. American Journal of Cardiology. 1990;65:929-930.

Petersen MEV, Williams TR, Sutton R. A comparison of non-invasive continuous finger blood pressure measurement (Finapres) with intra-arterial pressure during prolonged head-up tilt. European Heart Journal. 1995;16:1647-1654. Petersen MEV, Chamberlain-Webber R, Fitzpatrick AP, Ingram A, Williams T, Sutton R. Permanent pacing for cardioinhibitory malignant vasovagal syndome. British Heart Journal. 1994;71:274-281.

Planiol T, Pourcelot L, Itti R. Radioisotopes, ultrasonics and thermography in the diagnosis of cerebral circulatory disorders.

Revue d Electroencephalographie et de Neurophysiologie Clinique. 1974;4:221-236.

Potter EK.

Presynaptic inhibition of cardiac vagal postganglionic nerves by neuropeptide-Y. Neuroscience Letters. 1987;83:101-106.

Potter EK.

Prolonged non-adrenergic inhibition of cardiac vagal action by sympathetic stimulation: neuromodulation by neuropeptide-Y? Neuroscience Letters. 1985;54:117-121.

Poulin MJ, Robbins PA. Indexes of flow and cross-sectional area of the middle cerebral artery using Doppler ultrasound during hypoxia and hypercapnia in humans. Stroke. 1996;27:2244-2250.

Prado R, Watson BD, Kuluz J, Dietrich WD. Endothelium-derived nitric oxide synthetase inhibition. Stroke. 1992;23:1118-1124.

Prinz-Zaiss M, Yeap AN, Moguilevski V, Trigg L, McGrath BP. Power spectral analysis of heart rate variability during graded head-up tilting in patients with vasodepressor syncope. Clinical & Experimental Pharmacology & Physiology. 1995;22:472-4.

Pruvot E, Vesin JM, Schlaepfer J, Fromer M, Kappenberger L. Autonomic imbalance assessed by heart rate variability analysis in vasovagal syncope. Pacing & Clinical Electrophysiology. 1994;17:2201-6.

Raven PB, Pawelczyk JA. Chronic endurance exercise training: a condition of inadequate blood pressure regulation and reduced tolerance to LBNP. Review. Medicine & Science in Sports & Exercise. 1993;25:713-721.

Raven PB, Tohm-Young D, Blomqvist CG. Physical fitness and cardiovascular response to lower body negative pressure. American Physiological Society. 1984;56:138-144.

Raviele A, Brignole M, Sutton R, Alboni P, Giani P, Menozzi C, Moya A. Effect of etilefrine in preventing syncopal recurrence in patients with vasovagal syncope: a double-blind, randomized, placebo-controlled trial.

Circulation. 1999;99:1452-1457.

Raviele A, Menozzi C, Brignole M, Gasparini G, Alboni P, Musso G, Lolli G, Oddone D, Dinelli M, Mureddu R. Value of head-up tilt testing potentiated with sublingual nitroglycerin to assess the origin of unexplained syncope. American Journal of Cardiology. 1995;76:267-272.

Raviele A, Gasparini G, Di Pede F, Menozzi C, Brignole M, Dinelli M, Alboni P, Piccolo E. Nitroglycerin infusion during upright tilt: a new test for the diagnosis of vasovagal syncope. American Heart Journal. 1994;127:103-11.

Raviele A, Gasparini G, Di Pede F, Delise B, Bonso A, Piccolo E. Usefulness of head-up tilt table test in evaluating syncope of unknown origin and negative electrophysiologic study. American Journal of Cardiology. 1989;65:1322-1327.

Rea R.

Neurally mediated hypotension and bradycardia: which nerves? How mediated? Journal of the American College of Cardiology. 1989;14:1633-1634.

Rein H. Vasomotorische Regulationen. Ergebn Physiologie. 1931;32:28.

Reynolds KJ, Panerai RB, Kelsall AWR, Rennie JM, Evans DH. Spectral pattern of neonatal cerebral blood flow velocity: comparison with spectra from blood pressure and heart rate. Pediatric Research. 1997;41:276-284

Roatta S, Micieli G, Bosone D, Losano G, Bini R, Cavallini A, Passatore M. Effect of generalised sympathetic activation by cold pressor test on cerebral haemodynamics in healthy humans. Journal of the Autonomic Nervous System. 1998;71:159-166.

Robinson TG, James M, Youde J, Panerai R, Potter J. Cardiac baroreceptor sensitivity is impaired after acute stroke. Stroke. 1997;28:1671-1676.

Roddie IC. Human responses to emotional stress. Irish Journal of Medical Science. 1977;146:395-417.

Rodrigo J, Riveros-Moreno V, Bentura ML, Uttenthal LO, Higgs EA, Fernandez AP, Polak JM, Moncada S, Martinez-Murillo R. Subcellular localization of nitric oxide synthetase in the cerebral ventricular system, subfornical organ, area postrema, and blood vessels of the rat brain. Journal of Comparative Neurology. 1997;378:522-534.

Rogers RL, Meyer JS, Shaw TG, Mortel KF, Thornby J. The effects of chronic cigarette smoking on cerebrovascular responsiveness to 5 per cent and 100 per cent O_2 inhalation. Journal of the American Geriatrics Society. 1984;32:415-420.

Rogers RL, Meyer JS, Shaw TG, Mortel KF. Reductions in regional cerebral blood flow associated with chronic consumption of alcohol. Journal of the American Geriatrics Society. 1983;31:540-543.

Rongen GA, Bos WJW, Lenders JWM, van Montfrans GA, van Lier HJJ, van Goudoever J, Wessleing KH, Thien T. Comparison of intrabrachial and finger blood pressure in healthy elderly volunteers. American Journal of Hypertension. 1995;8:237-248.

Rosenblum WI. Neurogenic control of cerebral circulation. Stroke. 1971;2:429-439.

Rowell LB, Blackmon JR. Hypotension induced by central hypovolaemia and hypoxaemia. Clinical Physiology. 1989;9:269-277.

Rozycki HJ, Sysyn GD, Marshall MK, Malloy R, Wiswell TE. Mainstream end-tidal carbon dioxide monitoring in the neonatal intensive care unit. Pediatrics. 1998;101:648-653.

Ruiz GA, Scaglione J, Gonzalez-Zuelgaray J. Reproducibility of head-up tilt test in patients with syncope. Clinical Cardiology. 1996;19:215-220.

Sadoshima S, Yoshida F, Ibayashi S, Shiokawa O, Fujishima M. Upper limit of cerebral autoregulation during development of hypertension in spontaneously hypertensive rats: effect of sympathetic denervation. Stroke. 1985;16:477-481.

Sakuma I, Togashi H, Yoshioka M, Saito H, Yanagida M, Tamura M, Kobayashi T, Yasuda H, Gross SS, Levi R. NG-methyl-L-arginine, an inhibitor of L-arginine-derived nitric oxide synthesis,

stimulates renal sympathetic nerve activity in vivo. A role for nitric oxide in the central regulation of sympathetic tone?

Circulation Research. 1992; 70:607-611.

Samoil D, Grubb BP. Vasovagal (neurally mediated) syncope: pathophysiology, diagnosis, and therapeutic approach. European Journal of Cardiac Pacing & Electrophysiology. 1992;2:234-241. Sander-Jensen K, Secher NH, Astrup A, Christensen NJ, Giese J, Schwartz TW, Warberg J, Bie P. Hypotension induced by passive head-up tilt: endocrine and circulatory mechanisms. American Journal of Physiology. 1986;251:R742-748.

Sander-Jensen K, Garne S, Schwartz TW. Pancreatic polypeptide release during emotionally induced vasovagal syncope. Letter. Lancet. 1985;2:1132.

Satterthwaite FW. An approximate distribution of estimates of variance components. Biometrics Bulletin. 1946;2:110-114.

Savin E, Bailliart O, Checoury A, Bonnin P, Grossin C, Martineaud J-P. Influence of posture on middle cerebral artery mean flow velocity in humans. European Journal of Applied Physiology. 1995;71:161-165.

Schobel HP, Oren RM, Mark AL, Ferguson DW. Naloxone potentiates cardiopulmonary baroreflex sympathetic control in normal humans. Circulation Research. 1992;70:172-183.

Schondorf R, Benoit J, Wein T. Cerebrovascular and cardiovascular measurements during neurally mediated syncope induced by head-up tilt. Stroke. 1997;28:1564-1568.

Scott WA, Pongiglione G, Bromberg BI, Schaffer MS, Deal BJ, Fish FA, Dick M. Randomized comparison of atenolol and fludrocortisone acetate in the treatment of pediatric neurally mediated syncope. American Journal of Cardiology. 1995;76:400-402.

Selkurt EE. Relationship of renal blood flow to effective arterial pressure in the intact kidney of the dog. American Journal of Physiology. 1946;147:537.

Semplicini A, Simonella C, Meneghetti G, Chierichetti F, Serena L, Claroni F, Fazari G, Ferlin G, Pessina AC. Effects of fosinopril and hydrochlorthiazide on cerebral perfusion in uncomplicated essential hypertension. Journal of Hypertension. 1993;11:s372-s373.

Serrador JM, Picot PA, Rutt BK, Shoemaker JK, Bondar RL. MRI measures of middle cerebral artery diameter in conscious humans during simulated orthostasis. Stroke. 2000;31:1672-1678. Serrador JM, Bondar RL, Hughson RL. Ventilatory response to passive head up tilt. Advances in Experimental Medicine & Biology. 1998;450:133-139.

Shalev Y, Gal R, Tchou PJ, Anderson AJ, Avitall B, Akhtar M, Jazayeri MR. Echocardiographic demonstration of decreased left ventricular dimensions and vigorous myocardial contraction during syncope induced by head-up tilt. Journal of the American College of Cardiology. 1991;18:746-51.

Sharpey-Schafer EP. Emergencies in general practice: syncope. British Medical Journal. 1956;1:506-509.

Shaw TG, Mortel KF, Meyer JS, Rogers RL, Hardenberg J, Cutaia MM. Cerebral blood flow changes in benign aging and cerebrovascular disease. Neurology. 1984;34:855-862.

Sheldon R, Splawinski J, Killam S. Reproducibility of isoproterenol tilt-table tests in patients with syncope. American Journal of Cardiology. 1992;69:1300-1305.

Shen W-K, Hammill SC, Munger TM, Stanton MS, Packer DL, Osborn MJ, Wood DL, Bailey KR, Low PA, Gersh BJ. Adenosine: potential modulator for vasovagal syncope. Journal of the American College of Cardiology. 1996;28:146-154.

Shepherd JT. The lungs as receptor sites for cardiovascular regulation. Circulation. 1981;63:1-10. Review.

Shipley RE, Study RS.

Changes in renal blood flow, excretion of insulin, glomerular filtration rate, tissue pressure and urine flow with acute alteration of renal artery blood pressure. American Journal of Physiology. 1951;167:676-688.

Silverstein MD, Singer DE, Mulley AG, Thibault GE, Barnett GO. Patients with syncope admitted to medical intensive care units. Journal of the American Medical Association. 1982;248:1185-1189.

Simon RP. Syncope. In: Cecil Textbook of Medicine. Goldman L, Bennett JC editors. Saunders. 2000;p2028.

Sivan Y, Eldadah MK, Cheah TE, Newth CJ. Estimation of arterial carbon dioxide by end-tidal and transcutaneous pCO₂ measurements in ventilated children. Pediatric Pulmonology. 1992;12:153-157.

Skinhøj E.

The sympathetic nervous system and the regulation of cerebral blood flow in man. Stroke. 1972;3:711-716.

Skinhøj E. The upper limit of autoregulation and the sympathetic nervous system. Stroke. 1973;4:372. Abstract.

Slivka A. Effect of antihypertensive therapy on focal stroke in spontaneously hypertensive rats. Stroke. 1991;22:884-888.

Smielewski P, Czosnyka M, Kirkpatrick P, Pickard JD. Evaluation of the transient hyperaemic response test in head-injured patients. Journal of Neurosurgery. 1997;86:773-778.

Smielewski P, Czosnyka M, Kirkpatrick P, McEroy H, Rutkowska H, Pickard JD. Assessment of Cerebral Autoregulation Using Carotid Artery Compression. Stroke. 1996;27:2197-2203.

Smielewski P, Czosnyka M, Iyer V, Piechnik S, Whitehouse H, Pickard J. Computerised transient hyperaemic response test – a method for the assessment of cerebral autoregulation. Ultrasound in Medicine & Biology. 1995;21:599-611.

Smith JJ, Porth CM, Erickson M. Hemodynamic response to the upright posture. Journal of Clinical Pharmacology. 1994;34:375-386.

Smith ML, Carlson MD, Thames MD. Naloxone does not prevent vasovagal syncope during simulated orthostasis in humans. Journal of the Autonomic Nervous System. 1993;45:1-9.

Smith SA, Stallard TJ, Salih MM, Littler WA. Can sinoaortic baroreceptor heart reflex sensitivity be determined from phase IV of the Valsalva manoeuvre? Cardiovascular Research. 1987;21:422-427.

Sneddon JF, Scalia G, Ward DE, McKenna WJ, Camm AJ, Frenneaux MP. Exercise induced vasodepressor syncope. British Heart Journal. 1994;71:554-557.

Sneddon JF, Counihan PJ, Bashir Y, Haywood GA, Ward DE, Camm AJ. Assessment of autonomic function in patients with neurally mediated syncope: augmented cardiopulmonary baroreceptor responses to graded orthostatic stress. Journal of the American College of Cardiology. 1993;21:1193-8.

Squire IB.

Actions of angiotensin II on cerebral blood flow autoregulation in health and disease.

Journal of Hypertension. 1994;12:1203-1208.

Sra J, Maglio C, Biehl M, Dhala A, Blanck Z, Deshpande S, Jazayeri MR, Akhtar M. Efficacy of midodrine hydrochloride in neurocardiogenic syncope refractory to standard therapy. Journal of Cardiovascular Electrophysiology. 1997;8:42-46.

Sra JS, Murthy V, Natale A, Jazayeri MR, Dhala A, Deshpande S, Sheth M, Akhtar M. Circulatory and catecholamine changes during head-up tilt testing in neurocardiogenic (vasovagal) syncope. American Journal of Cardiology. 1994;73:33-37.

Sra JS, Jazayeri MR, Avitall B, Dhala A, Deshpande S, Blanck Z, Akhtar M. Comparison of cardiac pacing with drug therapy in the treatment of neurocardiogenic (vasovagal) syncope with bradycardia or asystole. New England Journal of Medicine. 1993;328:1085-1090.

Sra JS, Murthy VS, Jazayeri MR, Shen Y-H, Troup P, Avitall B, Akhtar M. Use of intravenous esmolol to predict efficacy of oral beta-adrenergic blocker therapy in patients with neurocardiogenic syncope. Journal of the American College of Cardiology. 1992;19:402-408.

Sridhar MK, Carter R, Moran F, Banham SW. Use of a combined oxygen and carbon dioxide transcutaneous electrode in the estimation of gas exchange during exercise. Thorax. 1993;48:643-647.

Steiger HJ, Aaslid R, Stooss R, Seiler RW. Transcranial Doppler monitoring in head injury: relations between type of injury, flow velocities, vasoreactivity and outcome. Neurosurgery. 1994;34:79-85.

Steurer J, Hoffmann U, Dur P, Russi E, Vetterr W. Changes in arterial and transcutaneous oxygen and carbon dioxide tensions during and after voluntary hyperventilation. Respiration. 1997;64:200-205.

Stevens PM, Lamb LE. Effects of lower body negative pressure on the cardiovascular system. American Journal of Cardiology. 1965;16:506-515.

Stoll M, Seidel A, Schimrigk K, Hamann GF. Orthostasis as a test for cerebral autoregulation in normal persons and patients with carotid artery disease. Journal of Neuroimaging. 1999;9:113-117.

Strandgaard S, Paulson OB. Cerebral blood flow and its pathophysiology in hypertension. American Journal of Hypertension. 1989;2:486-492.

Strandgaard S. Autoregulation of cerebral blood flow in hypertensive patients. The modifying influence of prolonged antihypertensive treatment on the tolerance of acute, druginduced hypotension. Circulation. 1976;53:720-727.

Strandgaard S, Olesen J, Skinhøj E, Lassen NA. Autoregulation of brain circulation in severe arterial hypertension. British Medical Journal. 1973;1:507-510.

Strasberg B, Rechavia E, Sagie A, Kusniec J, Mager A, Sclarovsky S, Agmon J. Usefulness of head-up tilt table test in evaluating patients with syncope of unknown origin.

American Heart Journal. 1989;118:923-927.

Strebel S, Lam AM, Matta B, Mayberg TS, Aaslid R, Newell DW. Dynamic and static cerebral autoregulation during isoflurane, desflurane and propofol anaesthesia. Anaesthesiology. 1995;83:66-76.

Streeten D, Anderson G, Richardson R, Thomas D. Abnormal orthostatic changes in blood pressure and heart rate in subjects with intact sympathetic nervous function: evidence for excessive venous pooling. Journal of Laboratory & Clinical Medicine. 1988;111:326-335.

Sugimori H, Ibayashi S, Fujii K, Sadoshima S, Kuwabara Y, Fujishima M. Can transcranial Doppler really detect reduced cerebral perfusion states? Stroke. 1995;26:2053-2060.

Sung RYT, Du ZD, Yu CW, Yam MC, Fok TF. Cerebral blood flow during vasovagal syncope induced by active standing or head up tilt.

Archives of Disease in Childhood. 2000;82:154-158.

Sutton R.

Syncope and palpitation. p2173-2175. In:Oxford Textbook of Medicine. Wetherall DJ, Ledingham JGG, Warrell DA editors. Oxford University Press. 1996.

Sutton R, Petersen M, Brignole M, Raviele A, Menozzi C, Giani P. Proposed classification for tilt induced vasovagal syncope. European Journal of Cardiac Pacing & Electrophysiology. 1992;3:180-183.

Talman WT, Dragon DN, Ohta H. Baroreflexes influence autoregulation of cerebral blood flow during hypertension. American Journal of Physiology. 1994;267:H1183-H1189.

Tanaka H, Thulesius O.

Effect of temperature on finger artery pressure evaluated by volume clamp technique. Clinical Physiology. 1993;13:535-545.

Tanaka K, Chiba T. Nitric oxide synthetase containing neurons in the carotid body and sinus of the guinea pig. Microscopy Research & Technique. 1994;29:90-93.

Tanaka K, Fukuuchi Y, Gomi S, Mihara B, Shirai T, Nogawa S, Mozaki H, Nagata E. Inhibition of nitric oxide synthesis impairs autoregulation of local cerebral blood flow in the rat.

Neuroreport. 1993;4:267-270.

Ten Harkel, ADJ, Van Lieshout JJ, Van Lieshout EJ, Wieling W. The assessment of cardiovascular reflex tests: influence of posture and period of preceding rest. Journal of Applied Physiology. 1990;68:147-153.

Thames MD. Effect of d- and l-propranolol on the discharge of cardiac vagal C fibres. American Journal of Physiology. 1980;238:H465-H470.

Thilenius O, Ryd K, Husayni J.

Variations in expression and treatment of transient neurocardiogenic instability. American Journal of Cardiology. 1992;69:1192-1195.

Theodorakis GN, Kremastinos DT, Avrambos GT, Stefanakis GS, Karavolias GK, Toutouzas PK. Heart rate variability in patients with vasovagal syndrome. Pacing & Clinical Electrophysiology. 1992;15:2221-5.

Thomas DJ. Whole blood viscosity and cerebral blood flow. Stroke. 1982;13:285-286.

Thomson HL, Wright K, Frenneaux M. Baroreflex sensitivity in patients with vasovagal syncope. Circulation. 1997;95:395-400.

Thomson HL, Atherton JJ, Khafagi FA, Frenneaux MP. Failure of reflex venoconstriction during exercise in patients with vasovagal syncope. Circulation. 1996;93:953-959.

Thomson HL, Lele SS, Atherton JJ, Wright KN, Stafford W, Frenneaux MP. Abnormal forearm vascular responses during dynamic leg exercise in patients with vasovagal syncope. Circulation. 1995;92:2204-9.

Thorin E, Nguyen T, Bouthillier A.

Control of vascular tone by endogenous endothelin-1 in human pial arteries. Stroke. 1998;29:175-180.

Thorin-Trescases N, Bartolotta T, Hyman N, Penar PL, Walters CL, Bevan RD, Bevan JA. Diameter dependence of myogenic tone of human pial arteries. Stroke. 1997;28:2486-2492.

Thulin T, Fager B, Grabowski M, Ryding E, Elmqvist?, Johansson BB. Cerebral blood flow in patients with severe hypertension and acute and chronic effects of felodipine. Journal of Hypertension. 1993;11:83-88.

Tiecks FP, Douville CBA, Byrd S, Lam AM, Newell DW. Evaluation of impaired cerebral autoregulation by the Valsalva maneuver. Stroke. 1996;27:1177-1182.

Tiecks FP, Lam AP, Aaslid R, Newell DW. Comparison of static and dynamic cerebral autoregulatory measurements. Stroke. 1995;26:1014-1019.

Tobias JD, Meyer DJ. Noninvasive monitoring of carbon dioxide during respiratory failure in toddlers and infants: end-tidal versus transcutaneous carbon dioxide. Anesthesia & Analgesia. 1997;85:55-58.

Tobias JD, Flanagan JF, Wheeler TJ, Garret JS, Burney C. Noninvasive monitoring of end-tidal CO_2 via nasal cannulas in spontaneously breathing children during the perioperative period. Critical Care Medicine. 1994;22:1805-1808.

Toda N, Okamura T. Nitroxidergic nerve: regulation of vascular tone and blood flow in the brain. Journal of Hypertension. 1996;14:423-434.

Tominaga S, Strandgaard S, Uemura K, Ito K, Kutsuzawa T. Cerebrovascular CO_2 reactivity in normotensive and hypertensive man. Stroke. 1976;7:507-510.

Totaro R, Marini C, Cannarsa C, Prencipe M. Reproducibility of transcranial Dopplersonography: a validation study. Ultrasound in Medicine & Biology. 1992;18:173-177.

Traub YM, Shapiro AP, Dujovny M, Nelson D. Cerebral blood flow changes with diuretic therapy in elderly subjects with systolic hypertension. Clinical & Experimental Hypertension. 1982;A4:1193-1201.

Trempler KK, Shoemaker WC.

Transcutaneous oxygen monitoring of critically ill adults, with and without low flow shock.

Critical Care Medicine. 1981;9:706-709.

Tschudi MR, Barton M, Bersinger NA, Moreau P, Cosentino F, Noll G, Malinski T, Luscher TF.

Effect of age on kinetics of nitric oxide release in rat aorta and pulmonary artery. Journal of Clinical Investigation. 1996;98:899-905.

Ueno T, Yoshimoto S, Mayanagi Y, Sekiguchi C, Yajima K. Effect of lower body negative pressure on cerebral circulation. Aviation, Space and Environmental Medicine. 1993;1006-1010.

van Lieshout JJ, Wieling W, Karemaker JM. Neural circulatory control in vasovagal syncope. Review. Pacing & Clinical Electrophysiology. 1997;20:753-63.

van Lieshout JJ, Wieling W, Karemaker JM, Eckberg DI. The vasovagal response. Clinical Science. 1991;81:575-586.

Venkata SR, Meese R, Kaplan NM, Devous MD, Bonte FJ. Antihypertensive therapy in the elderly. Effects on blood pressure and cerebral blood flow. American Journal of Medicine. 1987;82:53-56.

Verrill PJ, Aellig WH. Vasovagal faint in the supine position. British Medical Journal. 1970;4:348.

Victor J.

Tilt test: environment, material, patient preparation. In: Blanc JJ, Benditt D, Sutton R, eds. Neurally mediated syncope: pathophysiology, investigations and treatment. The Bakken Research Center Series Vol 10. Armonk, New York: Futura. 1996:77-78.

Victor RG, Thoren P, Morgan DA, Mark AL. Differential control of adrenal and renal sympathetic nerve activity during hemorrhage hypotension in rats. Circulation Research. 1989;64:686-694.

Vingerhoets AJJM. Biochemical changes in two subjects succumbing to syncope. Psychosomatic Medicine. 1984;46:95-103.

Vybiral T, Bryg RJ, Maddens ME, Boden WE. Effect of passive tilt on sympathetic and parasympathetic components of heart rate variability in normal subjects. American Journal of Cardiology. 1989;63:1117-20. Wacholder K. Haben die rhythmischen Spontankontraktionen de Gefasse einen nachweisbaren einfluss auf den blutstrom? Pflueger Arch. Ges. Physiol. 1921;190:222.

Wagner EM, Traystman RJ. Cerebrovascular transmural pressure and autoregulation. Annals of Biomedical Engineering. 1985;13:311-320.

Waldemar G, Ibsen H, Strandgaard S, Andersen A, Rasmussen S, Paulson OB. The effect of fosinopril sodium on cerebral blood flow in moderate essential hypertension. American Journal of Hypertension. 1990;3:464-470.

Waldemar G, Vorstrup S, Andersen A, Pedersen H, Paulson OB. Angiotensin-converting enzyme inhibition and regional cerebral blood flow in acute stroke. Journal of Cardiovascular Pharmacology. 1989;14:722-729.

Wallin BG, Sundolf. Sympathetic outflow to muscles during vasovagal syncope. Journal of the Autonomic Nervous System. 1982;6:287-291.

Ward CR, Gray JC, Gilroy JJ, Kenny RA. Midodrine: a role in the management of neurocardiogenic syncope. Heart. 1998;79:45-49.

Warlow CP, Dennis MS, van Gijn J, Hankey GJ, Sandercock PAG, Bamford JM, Wardlaw J. Cerebral Metabolism. In: Stroke – A practical guide to management. Blackwell Science. 1996. pp. 386-388.

Waxman MB, Asta JA, Cameron DA.

Myocardial and vascular contraction: Vasodepressor reaction induced by inferior vena cava occlusion and isoproterenol in the rat: role of beta₁- and beta₂-adrenergic receptors.

Circulation. 1994;89:2401-2411.

Waxman MB, Yao L, Cameron DA, Wald RW, Roseman J. Isoproterenol induction of vasodepressor-type reaction in vasodepressor-prone persons. American Journal of Cardiology. 1989;63:58-65.

Wayne HH. Syncope: physiologic considerations and an analysis of the clinical characteristics in 510 patients. American Journal of Medicine. 1961;30:418-438.

Weil MH, Bisera J, Trevino RP, Rackow EC.

264

Cardiac output and end-tidal carbon dioxide. Critical Care Medicine. 1985;13:907-909.

Weisbrod RM, Griswold MC, Yaghoubi M, Komalavilas P, Lincoln TM, Cohen RA. Evidence that additional mechanisms to cyclic GMP mediate the decrease in intracellular calcium and relaxation of rabbit aortic smooth muscle to nitric oxide. British Journal of Pharmacology. 1998;125:1695-1707.

White R, Deane C, Vallance P, Markus HS. Nitric oxide synthase inhibition in humans reduces cerebral blood flow but not the hyperemic response to hypercapnia. Stroke. 1998;29:467-472.

White R, Markus HS. Impaired dynamic cerebral autoregulation in carotid artery stenosis. Stroke. 1997;28:1340-1344.

Wieling W, Wesseling KH. Reflex adjustments to postural change. p43-49. In: Cardiovascular reflex control in health and disease. Hainsworth R & Mark AL editors. Saunders. 1993.

Wolthuis RA, Bergman SA, Nicogossian AE. Physiological effects of locally applied reduced pressure in man. Review. Physiological Reviews. 1974;54:566-595.

Wolthuis RA, Hoffler GW, Johnson RL. Lower body negative pressure as an assay technique for orthostatic tolerance: I. The individual response to a constant level (-40 mm. Hg) of LBNP. Aerospace Medicine. 1970;41:29-35.

Wynne HA, Schofield S. Drug-induced postural hypotension. In: Syncope in the Older Patient. 1996. Chapman & Hall. Kenny RA (Ed.).

Yamaguchi F, Meyer JS, Sakai F, Yamamoto M. Normal human aging and cerebral vasoconstrictive responses to hypocapnia. Journal of the Neurological Sciences. 1979;44:87-94.

Yamamoto M, Meyer JS, Sakai F, Yamaguchi F. Aging and cerebral vasodilator responses to hypercarbia. Archives of Neurology. 1980;37:489-496.

Yamanouchi Y, Jaalouk S, Shehadeh AA, Jaeger F, Goren H, Fouad-Tarazi FM. Changes in left ventricular volume during head-up tilt in patients with vasovagal syncope: an echocardiographic study. American Heart Journal. 1996;131:73-80.

Yamori Y, Horie R.

Developmental course of hypertension and regional cerebral blood flow in strokeprone spontaneously hypertensive rats. Stroke. 1977;8:456-461.

Yonehara T, Ando Y, Kimura K, Uchino M, Ando M. Detection of reverse flow by duplex ultrasonograpphy in orthostatic hypotension. Stroke. 1994;25:2407-2411.

Yoshida K, Okamura T, Toda N. Histological and functional studies on the nitroxidergic nerve innervating monkey cerebral, mesenteric and temporal arteries. Japanese Journal of Pharmacology. 1994;65:351-359.

Yoshizaki H, Yoshida A, Hayashi F, Fukuda Y. Effect of posture change on control of ventilation. Japanese Journal of Physiology. 1998;48:267-273.

Zaidi A, Benitez D, Gaydecki PA, Vohra A, Fitzpatrick AP. Haemodynamic effects of increasing angle of head up tilt. Heart. 2000;83:181-184.

Zaidi A, Clough P, Scheepers B, Fitzpatrick A. Treatment resistant epilepsy or convulsive syncope. British Medical Journal. 1998;317:869-870.

Zanzinger J, Seller H. Species differences in the distribution of nitric oxide synthetase in brain stem resions that regulate sympathetic activity. Brain Research. 1997; 764:265-268.

Zeng C, Zhu Z, Hu W, Liu G, Zhu S, Zhou Y, Shi W. Value of sublingual isosorbide dinitrate before isoproterenol tilt test for diagnosis of neurally mediated syncope. American Journal of Cardiology. 1999;83:1059-1063.

Zhang R (a), Zuckerman JH, Giller CA, Levine BD. Transfer function analysis of dynamic cerebral autoregulation in humans. American Journal of Physiology. 1998;274:H233-H241.

Zhang R (b), Zuckerman JH, Levine BD. Deterioration of cerebral autoregulation during orthostatic stress: insights from the frequency domain. Journal of Applied Physiology. 1998;85:1113-1122.